# University of Tennessee, KnoxvilleMasthead LogoTrace: Tennessee Research and Creative Exchange

#### **Doctoral Dissertations**

**Graduate School** 

12-2018

## Knowledge Sharing Matters: A Mixed Methods Study on Knowledge Practices Among University Presses

Pui-Yi Judy Li University of Tennessee, pli9@utk.edu

#### **Recommended** Citation

Li, Pui-Yi Judy, "Knowledge Sharing Matters: A Mixed Methods Study on Knowledge Practices Among University Presses." PhD diss., University of Tennessee, 2018. https://trace.tennessee.edu/utk\_graddiss/5293

This Dissertation is brought to you for free and open access by the Graduate School at Trace: Tennessee Research and Creative Exchange. It has been accepted for inclusion in Doctoral Dissertations by an authorized administrator of Trace: Tennessee Research and Creative Exchange. For more information, please contact trace@utk.edu.

To the Graduate Council:

I am submitting herewith a dissertation written by Pui-Yi Judy Li entitled "Knowledge Sharing Matters: A Mixed Methods Study on Knowledge Practices Among University Presses." I have examined the final electronic copy of this dissertation for form and content and recommend that it be accepted in partial fulfillment of the requirements for the degree of Doctor of Philosophy, with a major in Communication and Information.

Suzie Allard, Major Professor

We have read this dissertation and recommend its acceptance:

Devendra Potnis, Anne Smith, Carol Tenopir

Accepted for the Council: <u>Carolyn R. Hodges</u>

Vice Provost and Dean of the Graduate School

(Original signatures are on file with official student records.)

**Knowledge Sharing Matters:** 

A Mixed Methods Study on Knowledge Practices Among University Presses

A Dissertation Presented for the Doctor of Philosophy Degree The University of Tennessee, Knoxville

Pui-Yi Judy Li

December 2018

### Copyright©2018 by Pui-Yi Judy Li

All rights reserved.

#### **DEDICATION**

This dissertation honors my beloved parents for all their love, the sharing and transfer of their explicit and tacit knowledge, and their hard work in a small business to provide six great children with a better life. They were my role models for rising above difficult circumstances in life, always being positive and having the courage to move forward. I would also like to dedicate this research project to my grandmother, who saved and adopted a World War II orphan toddler (my father), even when she barely made ends meet. Without her, I would not be here.

#### ACKNOWLEDGEMENTS

The four and a half years in this program have been a challenging time for me, having to balance the heavy workload of a tenure-track faculty position and a full-time doctoral study program. The first semester in Fall 2014 was the most stressful time, with my enduring family issues in Hong Kong that began with my youngest sister's July passing and my mother's final-stage cancer diagnosis, to her October passing. I worried daily that my return home to Hong Kong in October might be too late to say goodbye to her.

Besides maintaining a self-disciplined spirit to get up at 5am daily to work on my course work/research project, I must say that I could not have gotten through my program without the following people:

- Dr. Suzie Allard, Chancellor's Professor and Dissertation Chair, for her patience and encouragement, and for helping me to think and solve unexpected research issues;
- Dr. Carol Tenopir, Chancellor's Professor, whose passion for research is inspiring, for being my Academic Advisor and helping me to understand research aspects;
- Dr. Devendra Potnis, Associate Professor, for helping me to understand the research challenges;
- Dr. Anne Smith, King and Judy Rogers Professor in Business, and Head of the Management Department at Haslam Business College, one of the best qualitative research teachers I have ever had, for her patience and encouragement, and for helping me to tackle qualitative research issues;

My colleagues Ken Wise, Allison Sharp, John Hanson and Rachel Ladd for their helping hands;

My fellow doctoral program classmates Frank Aviles and Abhijit Mazumdar for their encouragement;

All of the university press volunteers who participated in the survey/interviews and Brenna McLaughlin (Research & Communications Director at the Association of University Presses), for their helping hands and volunteering spirit;

My long-time friend Kerry Jacoby, for his helping hand and encouragement in my personal life;

And, finally, my employer the University of Tennessee and the Hodges Library, for giving me the tuition assistantship and a leave of absence in Fall 2018 to finish my dissertation.

#### ABSTRACT

Knowledge sharing in an organization helps improving its organizational performance (Walczak, 2008). Knowledge flows from various sources in various ways (Ahmed & Ahsan, 2014). Without effective knowledge sharing among employees, organizations cannot remain competitive in the ever-changing environments they face. The capturing of knowledge and human capital management are strategically important in a climate of retiring baby boomer workers and millennial workers' behavior of working for a short term with each employer (Liebowitz, 2008).

This study adapted Brenda Dervin's Sense-Making Metaphor and Weick's Organizational Sensemaking Theory to identify and explore the current knowledge-sharing practices of university presses. Since 2016, the not-for-profit university publishers have experienced alarming declines in their sales data and are struggling for survival. Acquiring an understanding of their current knowledge practices is a productive first step in their strategic planning process. It is time for university publishers to reinvent themselves strategically. This research used a sequential mixed methods study to collect quantitative and qualitative data. Quantitative data were gathered using a five-issue survey of university press employees. Qualitative data were gleaned from interviews of university press directors to understand top management's perceptions of their organizations' knowledge sharing practices. A major finding of this study is that the information behavior of the members of the university press organizations is strongly tied to social norms. Implications for knowledge management studies in Information Science contexts are discussed, and recommendations are made for future research.

#### PREFACE

Cognition, the tacit knowledge, is acquired through our natural instincts, beginning at birth and extending across our entire lives. Human beings are social animals by nature. It is only human beings, separate although still a part of the animal kingdom, who have a nature that distinguishes them from other species, by sharing explicit knowledge in all sorts of forms.

This dissertation in Information Science aims to explore how human beings share their explicit and implicit knowledge. In terms of the frequency of our constant interactions with it, knowledge is like oxygen that we inhale every second. In fact, a famous author's saying summarizes it nicely:

"Knowledge is of two kinds. We know a subject ourselves, or we know where we can find information on it" – Samuel Johnson

#### **TABLE OF CONTENTS**

Chapter One: Introduction	1
Background of the Study	1
Statement of the Problem	2
Purpose of the Study	3
Significance of the Study	3
Scope of the Study	5
Organization of the Study	5
Chapter Two: Review of the Literature	7
Theoretical Orientation related to Research Questions	7
Communication/Information Sciences and Organizational Theories	8
Dervin's Sense-making and Weick's Organizational Sensemaking	8
What Exactly is "Knowledge" About?	13
The Data-Information-Knowledge-Wisdom (DIKW) Hierarchy	14
Different Forms and Types of Knowledge	15
Knowledge Sharing, Knowledge Transfer and Knowledge Barriers	17
The Importance of Knowledge Management in Strategic Management	21
Issues of Information and Communication in Business Organizations	24
Knowledge Management in Business Enterprises	25
Scholarly Publishing	27
University Presses as Knowledge-Intensive Firms	29
Crisis at University Presses	29
Research Questions	30
Key Terms and Description of Variables	31
Chapter Three Methodology	33
An Overview of Research Methodology for this Project	33
Rationale for Selecting a Mixed Methods Approach for this Research Study	34
Research Questions and the Focus addressed in this study	36
Research Design – Mixed Methods Sequential Design	38
Philosophical Assumptions and Research Design for the Quantitative Portion	40
Data Collection and Procedures for the OUAN study – Surveying	41
Survey Development	41
Stage One – Review Other Surveys and Design Survey Instrument	41
Stage Two – Determining Sample Size using Statistical Prior Power Analysis	42
Survey Instrument	$\frac{12}{\Delta \Delta}$
Respondents	46
Compliance	40
Philosophical Assumptions and Research Design for the Qualitative Portion	18
Data Collection Procedure for the OUAL study - Interviewing	40 50
Interview Development/Instrument	50
Interview Guide/Protocol Development	51
The Use of the Critical Incident Technique (CIT) in Interviewing	51
Determining Sample Size	57 57
Interviewing Dreasedure	54 55
interviewing Procedure	22

Chapter Four: Data Analysis ar	d Results	58
Quantitative Research Resul	ts	58
The Rate of Return		58
Data Cleaning Process		59
Developing a Data Codeb	ook	62
Data Analysis using SPSS	5	63
Results		64
Profiles of the Respor	idents	65
Research Question 1:	Factors Influencing Individuals' Knowledge-Seeking and	
	Knowledge-Sharing in University Presses	70
Research Question 2:	Ways That University Presses Apply and Retain	
	Their Operational Knowledge in Their Organizations	81
Research Question 3:	Types of Knowledge Barriers that University Presses	
	Encountered	87
Qualitative Research Results	3	92
About the Interviews		92
Interviewees' Privacy and	I Identity Protection	92
Coding and Analysis usin	g QDA Miner	93
Developing a Theme Cod	ebook	95
Results		99
Interviewees' Profiles	·	99
Consistent Themes O	oserved	100
Identified Outliers		106
Research Question 1:	Factors Influencing Individuals' Knowledge-Seeking and	
	Knowledge-Sharing in University Presses	107
Research Question 2:	Ways That University Presses Apply and Retain	
	Their Operational Knowledge in Their Organizations	108
Research Question 3:	Types of Knowledge Barriers that University Presses	
	Encountered	109
Coded Segments in E	merging Themes by Organization Size	111
Philosophical Assumptions	of Mixed Method Approach for this Study	118
Data Integration and Inter	pretation: OUAN + OUAL	119
Validity and Reliability		124
OUAN		124
OUAL		126
Reflexivity Statement of	the Researcher	127
Chapter Five: Discussion and C	Vonclusions	128
Discussion		128
Lessons Learned		130
Limitations of the Study		133
Areas for Future Research		136
Recommendations		137
List of References		139
Appendices		161
Appendix A - Consent Form	s (Quantitative and Qualitative Portions)	162
Appendix B - Interview Gui	de (Qualitative Research)	166
The second secon		100

Appendix C - Recruiting Email (For Qualitative Portions)	169
Appendix D - Survey (Quantitative Research)	170
Appendix E – Variables (Quantitative Research)	179
Appendix F - Statistical Result Charts (Quantitative Research)	180
Appendix G - SPSS Codebook (Quantitative Research)	186
Appendix H - QDA Miner Codebook (Qualitative Research)	191
Appendix I - List of Diagrams	192
Diagram 1 - Sanders et al.'s Research Onion	192
Diagram 2 - Dervin's Sense-Making Theory	193
Diagram 3 - Ackoff's DIKW model	193
Diagram 4 - Clark's 2004 and Liew's 2013 models	194
Diagram 5 - Nonaka and Takeuchi's SECI model	194
Appendix J - Summary of Interviews	195
Vita	198

#### **LIST OF FIGURES**

Figure	1: An Overview of the Phases for this Research	6
Figure	2: Matching Research Questions with Theories	
Figure	3: An Overview of this Mixed Methods Sequential Design	
Figure	4: Flowchart of the Procedures in Implementing this Mixed Methods	
	Sequential Design	
Figure	5: An Example of Color Coded Segment	94
Figure	6: The Development of Theme Code Level	97
Figure	7: An Example of Content Analysis Count #1	
Figure	8: An Example of Content Analysis Count #2	
Figure	9: Reliability - Scale of Culture	
Figure	10: Reliability - Scale of Trust	

#### LIST OF TABLES

Table	1 a: Demographics (N =106)	68
Table	1_b: Demographics (N =106)	. 69
Table	2: (Q7) How did you acquire the skills and expertise that you use in your job?	.71
Table	3: (Q8) How often do you need to ask for information from your co-workers to	
	perform your daily job?	71
Table	4: (Q12) In my organization, I share my knowledge with co-workers because	. 72
Table	5: (Q13) My organization's management encourages informal activities and	
	opportunities to communicate/share experiences and work knowledge	73
Table	6: (Q14) There are formal activities such as training sessions, forums and	
	meetings in the organization to share knowledge and experience	.74
Table	7: (Q15) In my organization	75
Table	8: (Q16) There is a willingness in my organization that	76
Table	9: (Q17) Do you agree that staff, in your organization, do not have social	
	activities after work hours?	77
Table	10: (Q18) Do you agree that staff, in your organization, are eager to share work	
	experience in the hallways or in casual gatherings?	77
Table	11: (Q20) How motivated are you to share your knowledge and experience?	78
Table	12: (Q21) Do you agree or disagree with the following statements? (N=91)	80
Table	13_a: (Q9) Where would you look for knowledge that you need to do your work?	. 81
Table	13_b: (Q9) Write-In Answers	82
Table	14: (Q11) I depend on documented procedures to do my work when I have	
	questions in fulling my job tasks	83
Table	15: (Q22) In your organization, documenting of work knowledge is a required	
	part of your work practices	84
Table	16: (Q23) Mechanisms and tools to share work knowledge are widely available	
	in my organization	84
Table	17_a: (Q24) Do you use any of the following tools or methods to share or store	
	knowledge, experiences or best practices within your company?	85
Table	17_b: (Q24) Write-In Answers	86
Table	18_a: (Q10) What are the prohibitions in instances that you were unable to share or	
	receive knowledge and/or best practices from your team members in your job?	.87
Table	18_b: (Q10) Write-In Answers	88
Table	19: (Q19) From your view, what are the challenges of sharing knowledge with	
	people from other sections of your company?	. 89
Table	20: (Q25) The biggest barrier for you to store information you received more	
	efficiently and effectively is	90
Table	21: (Q26) From a knowledge sharing point of view, what changes would you like	
	to see in your company? Write-In Answers	91
Table	22: A summary of Interviewees' Profiles in the Qualitative Research	99
Table	23: Emerging Themes by Each Interview	105
Table	24: Mapping of the QUAN and QUAL Datasets to the Conceptual Frameworks	120

#### **CHAPTER ONE**

#### **INTRODUCTION**

#### **Background of the Study**

The importance of having knowledge has been accepted and appreciated ever since the philosopher Francis Bacon's saying "Knowledge is Power" first became popular in 1597. In fact, knowledge is now considered to be one of the important resources (Little & Ray, 2005; Jashapara, 2004) and the centerpiece of the survival and success of many organizations (Kluge, Stein & Licht, 2001). The concepts *knowledge-intensive firms* and *knowledge workers* appeared when economies and industries became more information and knowledge driven (Hislop, 2005). Workers' "knowhow" (i.e. human capital) is recognized as the one of most valuable assets of an organization and is the essence of innovation and profitability (Giddens, 1979).

The concepts of capturing human capital became formalized through the techniques of knowledge management (KM), which emerged as an established scientific discipline in the early 1990s (Nonaka, 1991). Knowledge management is an interdisciplinary endeavor, and spans the fields of Business Administration, Information Systems, Management, and Information Sciences (Alavi & Leidner, 1999). The management of knowledge involves the process of capturing, developing, sharing, and effectively using organizational intellectual assets to enhance the organizations' competitiveness (Davenport, 1994).

The concept of Saunders' Research Onion (2009) was used to map out the framework design for this dissertation. The Research Onion concept (Diagram 1) is adaptable for almost any type of research methodology and functions well in a variety of contexts (Bryman, 2012). This framework provided the researcher with a useful road map to walk through the layers in sequential steps, enabling the construction of a research design that would reflect the research objectives. In

general, the research philosophy and process were defined through the researcher's lens and were shaped by the nature of the research questions to be investigated. This technique led to the methodological choices in selecting the appropriate research approach (deductive or inductive, single-method or mixed-methods). Then, the research strategy for how to address or answer the research questions was designed. Next, the fourth layer identified the time horizon for collecting the data. The final stage identified the appropriate data collection methods. In this way, the onion metaphor sprouted into a series of stages through which an original methodological study was designed and described.

#### **Statement of the Problem**

Scholarly literature has confirmed that a positive correlation exists between a firm's performance and that organization's capacity to convert knowledge into value (Giju et al., 2010; Grant, 1996). The maintaining of organizational knowledge as a valuable intellectual asset is a driving force for innovation and competitiveness in a knowledge-based economy (Giju et al., 2010). Knowledge asset management provides small businesses with a competitive advantage for their survival and growth (Omerzel & Antoncic, 2008).

However, few empirical studies exist in the scholarly literature that identify the factors influencing knowledge management adoption in small businesses (Finkl & Ploder, 2009). The knowledge management literature situated in small organizational environment is relatively scarce, and is mainly centered on their practices related to competing in the market, rather than on the improvement of their internal efficiency (Sparrow, 2001). Therefore, there is a growing need for the analysis of knowledge management practices within networked small business enterprises (Valkokari & Haelander, 2007).

Reaping the benefits of sound knowledge management practices is particularly important for not-for-profit small business organizations such as university presses. University academic presses play an important role as disseminators of the best scholarship from their faculty and from affiliated scholars. With the rising trends of electronic books and open access scholarship in the publishing industry, university presses are experiencing decreasing book sales and are facing the challenges of competition from commercial publishers. An investigation of how effectively the employees of university presses share organizational knowledge internally, can provide the presses with a foundation for self-assessment in choosing their survival business models.

#### **Purpose of the Study**

The purpose of this mixed methods study with a sequential design was to examine the behavioral, cultural, social and technological aspects that affect knowledge practices in small knowledge-intense firms such as university presses. The research also identifies the methods and tools they use in sharing and curating knowledge. An understanding of these aspects and their interactions with internal knowledge flows and innovativeness can help to provide a foundation for the creation of a future small business self-assessment tool design, for evaluating and improving knowledge practices.

#### Significance of the Study

People interact through communication processes to create and share information, in order to reach a mutual understanding (Roger, 1995). Organizations are comprised of structured groups of interacting people who are tasked with specific responsibilities that serve the organization's mission and goals. Even for smaller organizations, the communicative interaction that must take place between the organization's individual members, to create shared understandings, become far more complex when they are amplified on a group rather than an individual scale. These communicative interactions are the channels through which information and knowledge can be shared among the members of an organization. However, few companies, especially among smaller businesses, have the means to capture employee knowledge and store it in a way that makes it easily accessible to members of the organization who might need it to inform their own work. Organizations put themselves at risk without having a knowledge management system, particularly in this era when there is a high level of workforce turnover, as baby-boomers in the workforce are retiring. Organizations clearly need to consider their processes for capturing and storing the vital knowledge and experience baby-boomer workers have amassed.

Strategic human capital management has become even more essential as many organizations are experiencing a "knowledge bleed," from not only the retiring baby-boomer workers, but also from the millennial workers' tendency to have a short work lifespan with each of a series of different employers, making frequent job changes (Liebowitz, 2008). Without a knowledge management system in place, organizations put themselves at risk of brain drain, especially as the generation gap widens and they find themselves with an impending talent vacuum (Rigoni, B. & Adkins, A., 2015).

This research was designed to gain insight into the current knowledge management practices of a small, knowledge-intensive firm environment, particularly regarding knowledge-sharing practices and knowledge-sharing barriers. An understanding of the behavioral, cultural, social and technological aspects of managing their knowledge would provide a foundational background for the development of a small-scale business enterprise self-assessment knowledge management auditing tool.

An understanding of the social-cultural aspects of businesses is important, because the success of a knowledge management system does not rely on technology alone (EIU & Cisco

System, 2006). Culture influences behavior and may subdivided into cultural subsystems or aspects (Gastil, 1961). The social-cultural aspects of knowledge and the technology need to be balanced in order to have a successful knowledge management system (Bhatt, 2001). A knowledge management tool cannot be implemented by itself to achieve success, without considering cultural issues (Davenport & Prusak, 1998).

#### **Scope of the Study**

Research is a process of discovery (Collis & Hussey, 2009) and the objective of this research was to discover and understand the behavioral, social, cultural and technological factors that impact employees' and managers' knowledge practices in the scholarly publishing sector, particularly in knowledge-intensive firms like university presses. In order to have a more in-depth and comprehensive study of this phenomenon, a comparison of these aspects from both a quantitative perspective on the employee level and from a qualitative perspective on the management level was conducted. This mixed methods study provided the opportunity to understand their knowledge practices from both top-down and bottom-up angles.

#### **Organization of the Study**

This dissertation is organized in five chapters covering three working phases: theoretical work, field work and the evaluation and conclusion (see Figure 1 – *Overview of the Phases for this Research*). Chapter One has introduced the entire research study in a summary form, including the background for the study as well as the significance of this research in the field. Chapter Two presents the review of relevant literature from the fields of Communication/Information Sciences and Organization Science. In addition, Chapter Two explicates the theoretical foundations for the study, which are Dervin's Sense-Making Theory and Weick's Organizational Sensemaking Theory. Chapter Three explains the research methodology used to conduct this study, the rationale

for the research design choices and specific procedures for the data analysis. Chapter Four presents the findings from each portion of the quantitative and qualitative data collection and integrates the two datasets to identify any convergence or divergence. Chapter Five provides a discussion of the significance and implications of the findings, and concludes with recommendations and suggestions for future research.



Figure 1 - An Overview of the Phases for this Research

#### **CHAPTER TWO**

#### **REVIEW OF THE LITERATURE**

In this chapter, the basis for the theoretical framework for this study is addressed by a discussion of the linkage between relevant theories and the research questions used to guide this research project. It is then followed by a review of the existing literature on these theories in two main strands: Communication/Information Science and Organizational theories. The literature review further develops the foundation that supports the research questions and illuminates the new area for exploration that this research addresses.

#### **Theoretical Orientations Related to Research Questions**

A theoretical framework is the use of a theory to explain an event or to shine light on a particular phenomenon or research problem (Fox & Bayat, 2007). Having the guidance of theoretical frameworks, the researcher gains an integrated understanding of issues in order to address each specific research problem (Imenda, 2014). Theories contribute to the selection of appropriate research methods because "producing methodological fit depends on the state of relevant theory at the time the research is designed and executed" (Edmondson & McManus, 2007, p. 1158). This study uses both Communication/Information Sciences and Organization Science theories to guide the research.

Dervin's Sense-Making Theory from the disciplines of Communication and Information Sciences, and Weick's Organizational Sensemaking from Organization Studies were used for the theoretical framework. In brief, Dervin's Sense-Making Theory (1983) describes the process of how individuals engage in internal cognitive sense-making, shaped by elements of their individual experience, when attempting to make sense of observed data and information. This theory is directly relevant to RQ1, which aims to understand how individuals seek out knowledge and experience knowledge barriers, to use Dervin's metaphor from her theory.

Weick's (1995) Organizational Sensemaking was selected to guide the portion of this research that addresses knowledge diffusion in the environment of the firm, specifically university press organizations. These theories provide a framework for RQ2 to understand how knowledge is being shared at an organizational level. For RQ3's exploration of knowledge barriers encountered during employees' performance of routine business processes, Dervin's Sense-Making Theory was selected because it addresses individual experiences. Weick's Organizational Sensemaking is relevant to explore for barriers encountered at the group and organization levels. An explanation of each theory is presented in detail in the following section.

## Communication/Information Sciences and Organization Theories Dervin's Sense-Making and Weick's Organizational Sensemaking

Sense making is the ability or attempt to gain insight into or an understanding of an ambiguous situation. In other words, the process of sense making is to create situational awareness and to understand circumstances of high complexity or uncertainty in order to facilitate decision making. Sense making is applicable to the study of information seeking and information uses in the workplace (Cheuk, 2007). The two most notable academic scholars on this topic are Karl Weick and Brenda Dervin. Weick's theory is more normative and takes an organizational approach, while Dervin's theory addresses individual communication.

Organizations operate in a dynamic and uncertain world, so it is very important for them to make sense of the changes and developments occurring in their external environments. Organizations generate new knowledge which can then be used in designing new products and services, or in enhancing their existing offerings, and in improving the efficiency and effectiveness of organizational processes. Organizations search for and evaluate new information in order to make educated decisions. Organizational members evaluate what is currently happening in their internal organizational environments in order to have a meaningful interpretation of organizational activities. Sense making begins when there are changes in the organizational environment. The application of a sense-making approach helps the understanding of the organization and leads to new knowledge creation.

In fact, sense making is a philosophically-informed methodological approach for attending to and researching human "communicating" (Foreman-Wesrnet, 2003). "By going through a continuous and simultaneous interplay, knowledge is created and re-created via sense making of external changes, sense making in knowledge creation and sense making in decision making." (Yao, Othman, Abdalla & Jing, 2011).

As Yao et al.'s article (2011) indicated, during the sense-making process, organizational members identify what information is significant and deserving of their attention. They form potential explanations of phenomena, shaped by their past experiences. They exchange and discuss their opinions to reach a collective interpretation, which guides the next information conversion or knowledge creation. During knowledge creation, members share their personal knowledge through meetings and trainings as well as through other formal or informal channels. When its members have attained sufficient understanding and knowledge, the organization is expected to select or design and evaluate a proper knowledge management strategy that is closely aligned with the overall organizational vision and goals. All three sense-making-based modules are dynamic information flow processes, subject to interruptions and iterations (Yao, Othman, Abdalla, & Jing, 2011).

Dervin's Sense-Making Theory focuses on how human beings communicatively make and unmake sense of the many diverse kinds of inputs that have the potential to be "informative." Dervin's Sense Making is a conceptual tool for understanding the relationship of information and meaning, and how human beings derive meaning from information. In Information Science, sense-making methodology was used in research projects, to shift the emphasis from information sources to information users (Dalrymple, 2001). This shift in focus was accomplished by viewing the act of "information seeking and use" as "modes of communication practice" (Savolainen, 1993, p. 13).

Dervin's Sense-Making Methodology (SMM) has been used as a theory or model for studying human behaviors within the meta-categories of information needs, seeking and use (Al-Suqri, 2015). SMM has been applied to research in Library and Information Science with a focus mainly on information seeking and knowledge gaps among users of information systems, as well as by other disciplines such as knowledge management, gender studies, librarianship practices and nursing practices (Dervin, 2014).

As per the illustrations by Dervin in the 2015 publication *Information Seeking Behavior and Technology Adoption*, central to SMM (see Appendix I - Diagram 2) as a methodology is conceptualizing the human being as a body-mind-heart-spirit living in a time-space, moving from a past, in a present, and to a future anchored in material conditions. The time-space, across situational conditions, includes a person's history, experience, horizons (past, present, future), constraints, barriers, habits and skills. This human being carries a type of metaphorical baggage, including power structures, organizational systems and procedures, cultures and communities labelled as "context." These factors are usually interpreted in research as structural arrangements within which individual agency operates. The human being faces gaps (questions, confusions) in his or her knowledge about a topic of interest or a decision to be made. The bridges to connect the past and future to permit the person to move onward are listed as ideas, cognitions, and thoughts, which include: attitudes, beliefs, values, feelings, emotion, and memories. Sources of information (which can take the form of channels, media, other people or institutions) are shown as potentially providing fodder for building the bridges that overcome the gaps. Along the way, the person can potentially make judgments about what information has either served or impeded movement along this journey. The outcomes of these kinds of judgments, called *relevance* in the metaphor, may be applied by this person to any material/interpretive encounter.

SMM assumes there are two requirements to allow humans beings to make some pronouncement on this disentanglement - one is to provide a framework for inviting input about outside expectations. The second is inventing ways of communicating collectively that allow unhampered differences to be aired without excessively self-centered outcomes. The intent of SMM is to try to yield research results from which researchers can make inferences and connections within their own discourse communities. SMM does not conceptualize the individual human being as the research unit of analysis. Rather, the unit of analysis can be described in multiple ways, such as person-in-situation, or person-asking-question, or person-evaluating-outcomes. SMM assumes that habitual, repetitive patterns can be found only under certain conditions and it is the researcher's task to use a research design that will illuminate those conditions as well as the conditions that foster flexibility of response (Al-Suqri, 2015).

In general, Dervin's methods focus on locating authentic situations in which meaning is created through the discovery of "gaps" and discontinuities in a person's information experience. Sense making characterizes knowledge perspectives as an "approach to studying human sense making" that primarily emphasizes the active process of communicating that occurs in time and space, not just what content was communicated. Sense-making is "what users want from systems, what they get, and what they think about them" (Dervin, 1992).

11

With a similar focus on cognition, Karl Weick's Sensemaking Theory emerged in the Organizational Communication and Management fields. The focus of Communication and Information Science in behavior studies had shifted to user-oriented rather than system-oriented approaches. In addition, LIS had opened up to more interpretative or phenomenological assumptions about the nature of communication and the nature of human sense making.

Weick's Organizational Sensemaking (1995) developed the concept of sensemaking as a way to understand "a central activity in the construction of both the organization and the environments it confronts." He views organizations as "collections of people trying to make sense of what is happening around them" (Weick, 1995). This theory defines sensemaking as the process through which we define our identity and continuously shape experience into meaningful patterns, enabling us to move forward in action despite contextual ambiguity.

Weick's model is more widely cited in the Organization Studies literature than in Information Science research (Fisher, Erdelez & McKechnie, 2005). Organizational researchers have studied sensemaking as part of strategic learning (Thomas, Sussman & Henderson, 2001). Researchers using Weick's Sensemaking Theory have followed varied methodological paths, but qualitative, interpretative approaches can be appropriate (Fisher, Erdelez & McKechnie, 2005). Weick's (1995) Sensemaking consists of seven elements: *identity* and identification are central – who people think they are, within their context, shapes what actions they may take and how they interpret events; *retrospection* provides the opportunity for sensemaking, because the point of retrospection in time affects what people notice; people *enact* the environments they face in dialogues and narratives; sensemaking is a *social* activity in that plausible stories are preserved, retained or shared; sensemaking is *ongoing*, so individuals simultaneously shape and react to the environments they face; people *extract cues* from the context to help them decide what information is relevant and what explanations are acceptable; and people favor *plausibility over accuracy* in accounts of events and contexts.

Weick's Sensemaking Theory has been applied in understanding personal, small group, organizational, national and global communication practices. This versatile theory has been used by constructivists as the framework for asking neutral questions via interview techniques. Through human stories that review cognitive and affective motivations to contextualize information behavior, sensemaking and narrative analysis offer the potential for discovering new vistas of information behavior (Fisher, Erdelez & McKechnie, 2005).

#### What Exactly is "Knowledge" About?

Knowledge is a fundamental building block for organizations to develop core competencies in order to face challenges, manage complexities, and remain competitive. Knowledge can be present in either explicit or tacit forms (Polanyi, 1962). Additional aspects of knowledge that scholars have investigated include the learning process (Szulanski, 1996), the creation of knowledge (Nonaka & Takeuchi, 1995), and the transfer of knowledge (Zander & Kogut, 1995). There is a need to understand the epistemology of this concept, through inquiries into "what knowledge is" and what types of knowledge exist, in order to understand the knowledge sharing, transfer and retention phenomena occurring in business enterprises, as well as to identify knowledge barriers in the performance of business processes.

In the time of the ancient Greeks, Plato (427-347 BC) and his student Aristotle (384-322 BC) were the earliest philosophers to try to answer the question "what is knowledge?" Plato, from the idealistic perspective, defined knowledge as a perception, true judgment and justified true belief (Annas, 2003). Aristotle, from the empiricist perspective, defined knowledge as that which is gathered through logical and empirical methods. Plato viewed knowledge as a state of being, while

Aristotle viewed knowledge as an action (Jashapara, 2003). These two early philosophical concepts of knowledge still exert a great influence on the logic and concept of knowledge used in the present day. For instance, Nonaka and Takeuchi (1995) defined knowledge as a dynamic human process for justifying personal belief of truth, similar to Plato's concept of true judgment. Davenport and Prusak (1998) defined knowledge as framed experiences and contextual information, similar to Aristotle's empirical viewpoints. As the centuries moved on, the conceptualization of knowledge was related to agricultural work (Wiig, 1997). In the Industrial Age, knowledge was about product leadership, operational efficiency, metrics and standards (Skyrme, 2000). In the Information Age, knowledge was focused on management leadership and organizational theories (Bennett, Wise, Woods, & Harvey, 2003), and in the knowledge era, knowledge is associated with intangible assets, value creation, and innovation (Skyrme, 2000).

#### The Data-Information-Knowledge-Wisdom (DIKW) Hierarchy

As a means to gain insight into what knowledge is and what it is not, Ackoff (1989) built a taxonomy of the DIKW hierarchy model (Diagram 3). As the DIKW pyramid indicates, wisdom is based on knowledge, knowledge comes from information and information comes from data (Ackoff, 1989). Therefore, data are the most basic units that are in such a raw form that they alone cannot be used to predict events or to describe the need to obtain data, but still play a great part in business and organizational processes (Davenport & Prusak, 1998). Information is defined as organized data, to the extent that what was originally only data becomes meaningful from a receiver's perspective (Jaspara, 2004). Knowledge resides in an individual's mind and is interpreted from information by individuals (Liyanage, Elhag, Ballal & Li, 2009). Therefore, knowledge is built on data and information.

Ackoff's 1989 DIKW hierarchy model (Diagram 3) has been used in the Information Science literature as a foundation for understanding knowledge. Some modifications have been made to it; for example, Clark (2004) introduced "understanding" into the DIKW model as a cognitive and analytic process, and Liew (2013) introduced "intelligence" as another unit in the DIKW model, to emphasize the inseparable relationship between knowledge and wisdom.

#### **Different Forms and Types of Knowledge**

Scholarly literature in Management and Information Science areas have indicated that knowledge is acquired through self-learning experience (internalization) or through interactive human activities (Small & Sage, 2005). Unlike data or information, knowledge is subjective, often based on experience, and highly contextual. Although several philosophers, as with Plato's "justified true beliefs," have tried to define knowledge in the past, there remains no definitive answers to the question of "what is knowledge?" However, philosophers have identified two essential classes of knowledge, which are explicit and implicit (tacit) knowledge. Knowledge has also been described as existing in three forms: public knowledge is explicit knowledge that has been taught and shared routinely in public domains; shared expertise is proprietary knowledge exclusively held and shared among knowledge workers in their own communication settings; and personal knowledge is tacit knowledge that exists in people's minds. Personal knowledge is the least accessible form, but it is used non-consciously in work and daily life (Dalkir, 2011). New knowledge is generated through the process of individual learning.

This conceptualization of knowledge also applies to organizational learning, which is based on applying knowledge for a purpose and knowledge is also generated by the process. Organizational knowledge acquisition is the "amplification and articulation of individual knowledge at the firm level" (Malhotra, 2000). Therefore, organizational learning (OL) is defined as "the bridge between working and innovating" that links learning to action for useful improvement (Brown & Duguid, 1991). In other words, organizational learning can be defined as cognitive learning from the process by which the organization improves over time, by assessing what has worked and what did not work in the past, and this collective knowledge is then transferred to benefit future knowledge workers (Dalkir, 2011).

Knowledge can be broadly classified into two main types: explicit knowledge and tacit knowledge, and then further subdivided into several sub-types: procedural, declarative, semantic and episodic knowledge. Explicit knowledge is knowledge that can be presented in various systematically written or encoded forms. However, tacit knowledge is subjective and based on experience, and is not easily articulated, captured, encoded or communicated in any written forms (Polanyi, 1966; Nonaka & Takeuchi, 1995). Polanyi (1966) asserted that explicit knowledge is rooted in tacit knowing, and tacit knowledge brings more value to an organization than explicit knowledge does. According to Nonaka and Takeuchi (1995), tacit knowledge has two dimensions: technical and cognitive. The technical dimension is the "know-how" that individuals have acquired from experience, but is difficult to articulate. The cognitive dimension is comprised of the schema, personal belief and perceptions that are ingrained in an individual.

Nonaka and Takeuchi (1995) illustrated the close tie between explicit knowledge and tacit knowledge in their knowledge creation framework called the SECI model (Diagram 5) which stands for: Socialization, Externalization, Combination and Internalization. This model defines a process of how tacit knowledge is converted to explicit knowledge and then generates new tacit knowledge. Socialization describes the processes in which tacit knowledge generates new tacit knowledge via human interaction in experience-sharing opportunities. The next step is Externalization, a process in which tacit knowledge is converted to an explicit form and is

embodied in documents, manuals or other codification formats. In this stage, tacit knowledge is crystalized into explicit forms which allow knowledge sharing with other participants. Combination is the process of converting explicit knowledge into another form of explicit knowledge, such as putting content from a written document into a database. Internalization is the process of converting explicit knowledge into tacit knowledge, such as when people learn and remember the explicit knowledge contained in a training manual, creating their own tacit knowledge. This cycle continues to spin in generating new knowledge.

From the psychological and behavioral perspectives, there are four types of knowledge (Collins, 1993): embrained knowledge, embodied knowledge, encoded knowledge and embedded knowledge. Embrained knowledge refers to explicit theoretical knowledge, and its availability depends on the level of the individual knowledge-seeker's skills. Embodied knowledge is action-oriented, similar to know-how knowledge (Polanyi, 1962) and the knowledge of experience (Nonaka & Takeuchi, 1995). Embodied knowledge is exemplified by the tacit knowledge of "learning-by-doing." Encoded knowledge is collective-specific explicit knowledge (Polanyi, 1962) that is stored in different databases, symbols and signs. Embedded knowledge is tacit knowledge based on common shared beliefs and understandings within organizations, that promote effective communication between employees for knowledge exchange. Tacit knowledge is generally difficult to articulate and transmit, but embedded knowledge reduces ambiguity or misinterpretations.

#### Knowledge Sharing, Knowledge Transfer and Knowledge Barriers

To study the use of communities of practice in knowledge management, key terms like knowledge sharing and knowledge transfer would be used in the discussion (Paulin & Suneson, 2012). These two terms have been used interchangeably in some literature (Badaracco, 1991;

17

Hansen, 1999) while some authors (Liyanage, Elhag, Ballal & Li, 2009) used these terms together. In fact, there is a common dividing line on the level of analysis between knowledge sharing and knowledge transfer. The term *knowledge sharing* is used by authors focusing on the individual level, while *knowledge transfer* is frequently used when focusing on groups, departments, or organizations (Argote & Ingram, 2000). This insight is further supported by the *Encyclopedia of Knowledge Management* (Schwartz, 2006), which states that knowledge sharing is defined as "the exchange of knowledge between and among individuals, within and among teams, organizational units, and organizations." In this source, knowledge transfer is defined as "the focused, unidirectional communication of knowledge between individuals, groups or organizations such that the recipient of knowledge has a cognitive understanding, the ability to apply the knowledge, or applies the knowledge." In fact, Christensen (2003, p. 8) emphasized that knowledge transfer is about not only identifying (accessible) knowledge, but also acquiring it and subsequently applying this knowledge to develop new ideas or enhance existing ideas, to make a process or action faster.

Knowledge sharing is an interactive people-to-people process to exchange knowledge (Ryu, Ho & Han, 2003). Knowledge transfer is "the conveyance of knowledge from one place, person or ownership to another" (Liyanage, Elha, Ballal & Li, 2009), and so it can also be defined as "the process through which one unit is affected by the experience of another" (Argote & Ingram, 2000, p. 151). The quality and degree of knowledge sharing that can actually take place is determined by the nature of the specific knowledge in question, the motivation to share, the opportunities to share, the organizational culture and the work environment (Paulin & Suneson, 2012).

In the literature, there are two dominant views of knowledge that have been derived to differentiate between knowledge sharing and knowledge transfer: 1) knowledge as an object to

share, and 2) knowledge as a subjective contextual construction in a social context (Sveiby, 2007). Knowledge sharing is considered to be an ongoing process in an organization, along with other activities (Christensen, 2007).

According to Christensen (2007), there are five factors that impact knowledge sharing in organizations: the stickiness of knowledge (sharing tacit knowledge requires more effort than sharing explicit knowledge), the lack of an identity, a weak relationship between a receiver and a sender of knowledge, the lack of a willingness to share knowledge, and no knowledge about knowledge. For knowledge transfer, Szulanski (1996) states that there are four stages: initiation, implementation, ramp-up and integration. The initiation stage is when the needs and wishes to have knowledge exist in an organization. The implementation stage includes transferring knowledge according to the receiver's requirements and involves a need to overcome problems. In the ramp-up stage, a receiver begins to determine and utilize the transferred knowledge to solve problems. The integration stage is when a receiver met her or his needs by applying the transferred knowledge. Szulanski also identifies four sets of factors affecting the transfer of knowledge: the attributes of the knowledge transfer, the attributes of the knowledge source, the attributes of the knowledge receiver and the attributes of the context.

The knowledge management literature also indicates that trust and motivation are important components of knowledge transfer and knowledge sharing. There are three types of trust involved (Dignum & Van Eijk, 2005): a personality-based trust to identify the trustworthiness of the receiver, an interpersonal trust based on their previous experience and interactions, and an impersonal trust, such as the person's trust in the organization.

Besides the above-mentioned factors that affect knowledge sharing and knowledge transfer, several knowledge sharing and transfer barriers have been identified across the Information

Science literature. In general, several types of knowledge barriers have been identified by scholars. Technological barriers refer to how different technological approaches or systems may cause incompatibilities that make it difficult for departments to transfer and share knowledge (McLaughlin, Paton & Macbeth, 2008). Content barriers occur when employees do not recognize or understand some procedures. Barriers can exist within the organization, such as having a weak culture of sharing. For example, in some Asian organizations, people will share knowledge only among family members and close colleagues (Yao, Kam, & Chan, 2007). Barriers based on organizational culture also include when employees perceive "knowledge sharing" as a threat or violation, and they "keep secret information to avoid losing their job" (Hermann, 2011). Personal barriers include circumstances where organizational members have no motivation to share, and withhold information to make themselves more competitive and to have an advantage over others (Hermann, 2011).

Knowledge transfer and innovation diffusion can help organizations to gear up for and be competitive in the knowledge-based economy. The process of knowledge transfer is fundamentally a people-to-people interaction. Communication lies at the heart of the knowledge transfer process. This leads to the notion that both innovation diffusion and knowledge transfer are related in some ways. Knowledge transfer involves networking with people to share knowledge between members and within organizations, so it can be identified as an act of communication. The knowledge transfer process has two main components: the source or sender shares the knowledge, and the receiver acquires the knowledge (Liyanage, Elhag & Ballal, 2012). Knowledge transfer (KT) is an area of knowledge management concerned with the movement of knowledge across the boundaries created by specialized knowledge domains (Carlile & Rebentisch, 2003). Knowledge management and innovation are interrelated in the sense that how well the organization adapts to its environment and becomes innovative, may depend on how well it succeeds in its knowledge-creation activities (Choo, 1998).

Diffusion of innovation refers to the communication, spread and adoption of new ideas among social communities (Rogers, 2003). Newell et al. (2000) indicated that there are strong ties (close associations among members of firms) and weak ties that link individuals from organizations across different sectors or communities that would not normally make contact during their day-to-day business. Regardless, there are strong and weak relationships, and both are important for the diffusion of new ideas. Researchers have affirmed that knowledge transfer accelerates the diffusion of innovation "by making exclusive know-how and/or proprietary knowledge available to others" as an input to further research and development (Liyanage, Elhag & Ballal, 2012). Knowledge transfer and innovation both require an organizational culture where people both want to and are encouraged to be innovative and share their knowledge (Horibe, 2007).

#### The Importance of Knowledge Management in Strategic Management

Knowledge flows are recognized as being among the most important elements in the economy (Yao, Othman, Abdalla & Jung, 2011). In fact, knowledge has become the key economic resource and the dominant source of competitive advantage (Drucker, 2011).

The field of knowledge management does suffer from the "Three Blind Men and an Elephant" syndrome (Dalkir, 2011). In fact, knowledge management is a highly multidisciplinary field. Knowledge management draws upon a vast number of diverse fields such as Organizational Science, Cognitive Science, Information Technologies, Information and Library Science, Sociology and Education.

Knowledge management is a systematic approach using procedures for capturing, structuring, managing, and disseminating knowledge throughout an organization, in order to work

21

fast, reuse best practices and reduce costly rework from project to project (Nonaka and Takeuchi, 1995).

Knowledge management is important to making the organization more productive, more effective and more successful. The application of knowledge management enhances collaboration, improves productivity, and enables and encourages innovation (Hibbard, 1997). In other words, organizational knowledge will lead to more effective means of generating, sharing and managing knowledge in an organization, as knowledge is inseparable from knowing how to get things done in complex organizational work (Guo & Sheffield, 2007).

Organizational knowledge is an important bundle of intangible resources that can be the source of a sustainable competitive advantage (Hitt, Ireland & Hoskisson, 2014). Knowledge has the greatest ability of all resources to serve as a source of sustainable differentiation, because of immobility (McEvily & Chakravarthy, 2002) and general applicability (Miller & Shamise, 1996). Knowledge permits the firm to predict more accurately the nature of the commercial potential of changes in the environment and the appropriateness of strategic and tactical actions (Cohen & Levinthal, 1990). Without such knowledge, an organization is less capable of discovering and exploiting new opportunities (Wiklund & Shepherd, 2003).

There are three levels of organizational perspectives on knowledge management (Wiig, 1993): the business perspective, the management perspective and the hands-on perspective. The business perspective focuses on why, where and to what extent the organization must invest in exploiting knowledge for business strategies, products and services planning, alliances, acquisitions or divestment from knowledge-relation points of views. The management perspective focuses on determining, organizing, directing, facilitating and monitoring knowledge-related practices and activities required to achieve the desired business strategies, innovations and

22
objectives. The operational hands-on perspective focuses on applying the "know-how" expertise knowledge to conduct specific knowledge-related work and tasks.

Procedural knowledge refers to knowing the step-by-step actions for how to do things and arises from one's experience with similar situations (Lesgold, 1998). It is difficult to formalize, articulate and transfer procedural knowledge between organizational contexts (Nonaka & Takeuchi, 1995). Knowledge about markets and technology represent two strands of procedural knowledge that can potentially have strong performance implications, because they increase the organization's ability to discover and exploit opportunities (Wiklund & Shepherd, 2003).

Entrepreneurial Orientation (EO) is a combination of three qualities: innovativeness, proactiveness and risk-taking (Wiklund, 1999). Knowledge-based resources (applicable to the discovery and exploitation of opportunities) are positively related to firm performance, and having an EO outlook enhances this relationship (Wiklund & Shepherd, 2003).

Developing the requisite market-technology knowledge for success requires knowledge creation and exploitation (Dougherty, 1992). Nonaka & Takeuchi (1995) distinguish between tacit and articulable. Tacit knowledge (know-how) is personal, not easily formalized, not easily communicated and is rooted in a specific context (Brown & Duguid, 1998). Knowledge may be tacit, not codifiable and thus acquired only through hands-on experience. Articulate knowledge is explicit, codifiable and transmittable through a formal or systematic language (Brown & Duguid, 1991). Tacit knowledge is rich and dense but not easily shared, while articulate knowledge is thin and grainy but easily shared (Dougherty, 1992).

Nonaka and Takeuchi (1995) indicated the creation of new knowledge occurs as the two types expand and interact over time. Cohen and Levinthal (1990) emphasized that the ability to recognize the value of new ideas, assimilate them and apply them to commercial ends depends in part on the base of prior knowledge, both tacit and articulated. Cohen and Levinthal suggested that the more objects, patterns and concepts are stored, the more readily new information about these constructs can be acquired. Knowledge creation is a social rather than an individual process since the transformation of tacit into articulate knowledge requires "direct and continual dialogues between people" who are grounded in the same situation (Nonaka and Takeuchi, 1995). Nonaka and Takeuchi suggested two approaches for transforming tacit into articulate knowledge. One is to create concepts, which are condensations of tacit images into language, drawings or gestures. Another knowledge creation process is to cluster and re-cluster information and meanings as they accumulate.

Knowledge is also said to be an immersion in the ongoing flow of events in a field of endeavor (Dougherty, 1992). A diversity of knowledge is important for novel domains because it increases the prospect that "incoming insights will relate to what is already known" (Cohen and Levinthal, 1990). An understanding of organizational renewal must be based on the processes that creating and exploiting knowledge is necessary to formulate viable products (Dougherty, 1992).

### **Issues of Information and Communication in Small Business Organizations**

Industry sectors vary when it comes to determining what constitutes a small business. In general, a small business is an independently-owned and operated company that is limited in size and in revenue, depending on the industry. Regardless of the types of industries, small businesses come across similar concerns in handling their human intellectual assets and managing knowledge.

Small business enterprises differ from large companies in several ways that affect their information-seeking practices. These differences include: the lack of a substantial information management system, the frequent concentration of information-gathering responsibilities that are

borne by only one or two individuals and lower levels of resources available for informationgathering (Lang, Calantone & Gudmundson, 1997).

Small businesses often face technological disadvantages in comparison to larger organizations. Information and communication technologies (ICTs) possess the potential to contribute significantly to economic growth. Small businesses are adopting ICTs to support their competitiveness, productivity and profitability. However, the diffusion of ICTs in small and medium enterprises (SBEs) is low (Assinform, 2010). ICT diffusion in small businesses differs from that in larger organizations because of the specific characteristics of SBEs, such as having relatively limited resources, technology and capabilities, although the less-complicated structure provides small firms with more flexibility in response to changes (Girgin, Kurt & Odabasi, 2011). The inhibiting factors that suppress investments in ICTs by SBEs are: the high initial financial investment, the lack of skilled staff, and that the technology is not user-friendly without adequate training (Consoli, 2012).

#### **Knowledge Management in Business Enterprises**

The importance of knowledge management in organizations is growing because we are immersed in a knowledge-based economy. Knowledge management (KM) emerged as an established scientific discipline in the early 1990s (Nonaka, 1991). Knowledge management spans the fields of business administration, information systems, management, and information sciences (Alavi & Leidner, 1999). As mentioned above, in the Information Science literature, the DIKW (Data, Information, Knowledge, Wisdom) model (also known as the Knowledge Pyramid) represents the purported structural and functional relationships between data, information, knowledge and wisdom (Rowley, 2007). This model implies that having knowledge is more than just having power; it also generates wisdom that is essential in strategic planning for organizational success. Currently, the overload of data which is the foundation of the pyramid, makes knowledge management increasingly important to an organization's success, by facilitating decision-making capabilities, building learning routines, and stimulating innovation. A trace of the roots of knowledge management indicate that a number of management theories have contributed to the evolution of KM concepts. Knowledge management-like approaches have existed for years in large organizations within commercial sectors all over the world, but they operated under different labels such as competitive intelligence, the learning organization, human knowledge or artificial intelligence.

Managing knowledge assets provides small businesses with new tools for survival growth and maintaining a sustainable competitive advantage (Omerzel & Antoncic, 2008). Many literatures describe how various large firms successfully practice knowledge management (Evangelista et al., 2010), but few empirical studies have identified the factors influencing knowledge management adoption in small business enterprises (Finkl & Ploder, 2009). The issues that small businesses face in implementing knowledge management practices are not simply a scaled-down replica of large company experiences (Sparrow, 2001). This is because most of the knowledge management research studies have focused on large companies and were oriented for their situations and needs (Wong & Aspinwall, 2004). Without an understanding of SBE's specific conditions, these research findings cannot be directly applied to the SBE environment (Wong, 2005).

Five key peculiarities (Desouza & Awazu, 2006) are identified in the literature to differentiate knowledge management practices in small business enterprises from those in large companies. They are: 1) lack of explicit knowledge repositories, because each manager/owner acts as the knowledge repository; 2) common knowledge possessed by SBE members is deep and

broad to ease the issues of knowledge transfer, sense-making and application; 3) the close social ties between SBE members act as a deterrent against the issues; 4) SBEs have a knack for exploiting foreign sources of knowledge because of their limited resources and because they cannot spend efforts to create new knowledge; and 5) technology is not as much a part of their knowledge management approach. Their use of technology is usually more minimal than for larger organizations.

Knowledge management has been a popular subject matter in doctoral dissertations since 1998 (Grossman, 2007). Knowledge management is a strategic initiative that changes the paradigm of information systems from data processing and providing information, to harvesting and capitalizing on the knowledge derived from individual members of the organization's expertise and capturing it as documented material (Hussain, Lucas & Ali, 2004). The scheme of applying Information Sciences to business administration typically focuses on organizational objectives such as improved performance, competitive advantage and innovation, the sharing of lessons learned, integration and the continuous improvement of the organization (Gupta & Sharma, 2004). However, the application of knowledge management in small business enterprises (SBEs) is quite different. Several researchers have done organization studies of knowledge management, but the existing empirical literature provides only fragmented insights into KM in SBEs. The field of knowledge management as applied in a small business context is a highly important phenomenon that "stills calls for more research" (Durst & Edvardsson, 2012, p. 898).

#### **Scholarly Publishing**

As a subset of the publishing industry, academic publishing includes thousands of publishers that disseminate academic research and scholarship. Most academic research is published in the form of academic journals or books. Many of these academic and scholarly works, though not all, are based on some form of peer-reviewed or editorial refereed processing to qualify for publication. The process of academic publishing begins when an author's manuscript is submitted to a publisher. Then the manuscript passes through two distinct phases: peer review and production.

The peer review process is organized by the journal editor and is increasingly managed online, through one or more rounds of review of the author's modifications in accordance with the reviewers' comments. This process is repeated until the work is accepted. The production process is controlled by the publisher, in that the article is taken through copy editing, typesetting and inclusion in a specific issue of the journal, often appearing in both print and online formats.

In general, there are two distinct endpoints on the spectrum of academic publishers. At one end, there are the university presses and discipline-specific associations that publish only one or two journals. At the other end, five giants – Elsevier, SpringerNature, Wiley-Blackwell, Taylor & Francis and Sage – publish the majority of academic papers, accounting for a huge market share worth in excess of \$10 billion US. In between are some small private publishers and major multidisciplinary associations such as the American Chemical Society (ACS), and the Institute of Electrical and Electronic Engineers (IEEE), that publish most of the research in their respective fields.

Since the early 1990s, academic publishing has undergone a major transition from the print to the electronic format, including the licensing of electronic resources that are used in a digital environment. Currently, open access for journal articles via the Internet is a trend, with authors' making upfront payments of hundreds or thousands of dollars in publication fees. This trend is also considered to be a threat to many not-for-profit scholarly publishing firms.

### **University Presses as Knowledge-Intensive Firms**

A knowledge-intensive firm (KIF) is defined as an organization that offers to the market the use of fairly sophisticated knowledge of knowledge-based products (Alvesson, 2004). The publishing sector is an example of a KIF-oriented industry. In terms of the nature of the work and how it is managed and organized, some specific characteristics of a KIF are: knowledge workers using intellectual and symbolic skills in their work; a high degree of individual autonomy, along with the downplaying of the organizational hierarchy; extensive communication for coordination and problem-solving; and subjective and uncertain quality assessment (Alvesson, 1995).

Small businesses in the academic publishing sector include independent commercial publishers and university presses. According to the Association of University Presses, university presses differ from commercial publishers in their place in the academic landscape. University presses are affiliated with their parent institutions, serving the public good by generating and disseminating knowledge as not-for-profit, mission-driven scholarly publishers.

#### **Crisis at University Presses**

Academic book publishing operates mainly in three major streams: the traditional formats of textbooks, monographs and general-interest titles (Schonfeld, 2016). University Presses face challenges to their traditional business model, because of the trends of digital access, changing acquisition patterns at the library and open-access monograph models in the publishing sector. With budget cuts and digital scholarly publishing trends, the book sales of university presses have declined ("Academic Books", 2017). The 2016 university press' sales data dropped down to the level of the 2005 figures and its core business – print monograph sales – is crumbling away with no clear revenue source to replace it. Used books and rental books are also eating into textbook sales (Straumsheim, 2016).

Among the university presses, the world's two oldest and largest university publishing houses, the Universities of Cambridge and Oxford (both established in the sixteenth century) have needed to reorganize and lay off employees to survive. Several American universities such as the University of Missouri have shut down their presses (Eligon, 2012). The rest of the university presses have struggled to survive by maintaining their impressive resilience and/or merging with their academic libraries (Schonfeld, 2016). Change and innovation are inevitable in the fight for survival by university presses.

#### **Research Questions**

This study is designed to understand the university presses' directors and employees in terms of their knowledge practices, including knowledge seeking, knowledge sharing, knowledge transfer and knowledge retention, in completing their job tasks. With these considerations, this research study addresses the following questions:

- RQ1: What factors influence individuals' knowledge-seeking and knowledge-sharing in knowledge-intensive firms such as university presses?
- RQ2: In what ways do knowledge-intensive firms like university presses apply and retain their operational knowledge in their organizations?

Ancillary question:

RQ3: What types of knowledge barriers do university presses encounter in their daily processes?

	RQ	Theories		
RQ1:	What factors influence individuals' knowledge seeking	Dervin's Sense-Making		
	and knowledge sharing in knowledge-intensive firms			
	such as university presses?			
RQ2:	In what ways do knowledge-intensive firms like	Weick's Organizational		
	university presses apply and retain their operational	Sensemaking		
	knowledge in their organizations?			
RQ3:	What types of knowledge barriers do university presses	Devin's Sense-Making,		
	encounter in their daily processes?	Weick's Organizational Sensemaking		

Figure 2 – Matching Research Questions with Theories

# Key Terms and Description of Variables

The following are definitions of the main terms being used throughout this study. Further explanation of and applicability in the use of these terms are discussed in Chapter Three. The description of measuring variables used in the quantitative research of this study can be found in Appendix E.

*Mixed Methods Research*: "Mixed methods research is a research design with philosophical assumptions as well as methods of inquiry ... Its central premise is that the use of quantitative and qualitative approaches in combination provides a better understanding of research problems than either approach alone" (Creswell & Plano-Clark, 2007).

*Knowledge-Intensive Firms*: firms where most work is said to be of an intellectual nature and where well-educated, qualified employees form the major part of the workforce (Alvesson, 2004).

*Small Business*: The term "small business" refers to the not-for-profit scholarly publishers of universities that publish books for scholars and specialists. A small business is defined by the United States Small Business Administration as an organization with less than 500 employees. In this research, the employee sizes of university presses were from one to 130 employees. Therefore, the small businesses in this research are mini-businesses.

*Behavior*: the ways in which people behave, individually and collectively, when working together in organizations.

*Culture*: the sharing of assumptions, values and beliefs which govern how people behave in organizations. These shared values have a strong influence and dictate how people in the organization interact and perform their jobs.

*Social Norm*: the common standards within a social group regarding socially acceptable or appropriate behavior in particular social situations.

*Technology*: the application of information to the design, production and utilization of goods and services related to the organization's activities.

*Knowledge Sharing*: an activity through which information, skills and expertise are exchanged among people in an organization.

32

#### **CHAPTER THREE**

# **METHODOLOGY**

This chapter discusses the methodology that was used for the research. It provides an overview of the methodological process, comprised of paradigm supposition, the underlying philosophical assumptions, and the different research inquiry designs, including methods for data collection and analysis.

#### An Overview of Research Methodology for this Project

Research methodology is the principle underlying the research methods to be carried out (Creswell, 2007) and a general approach for the research that is to be undertaken (Silverman & Marvasti, 2008). Research methodology includes how the research is conducted, and how data are gathered and analyzed to achieve the research goals. It is the principle and the logical processes that apply to a scientific investigation (Fellows & Liu, 1997). Research approaches are "plans and procedures that determine the philosophical assumptions that underlie the study, the procedures of inquiry or research design, and specific research methods of data collection, analysis and interpretation" (Creswell, 2014). The selection of a research approach depends on the nature of the research problem to be addressed. There are three major types of research approaches (Creswell, 2014): quantitative research, qualitative research and a mixed methods approach.

Quantitative research methods are adopted from natural sciences research. The strength of quantitative research methods is that quantifiable data can potentially be generalized to estimate the population. However, quantitative methods are weak in creating an understanding of the context or the social environment or setting of the research (Creswell & Plano-Clark, 2007). On the other hand, qualitative methods explore and/or explain social phenomenon via observation of or interaction with the participants of the study, in their natural settings. The researcher becomes

the data collection instrument, generating rich descriptions of the participants' interpretations of understanding "why" a phenomenon has occurred. This approach possesses downsides, with the researcher's possibly introducing bias through personal interpretations. In addition, there is the difficulty of generalizing the findings to a large population, because of the limited number of participants being studied (Creswell & Plano-Clark, 2007). Therefore, a mixed methods approach can offset the weakness of each component by providing a complete picture that notes trends and generalizations, as well as yielding in-depth knowledge of participants' perspectives (Creswell & Plano-Clark, 2007).

### Rationale for Selecting a Mixed Methods Approach for this Research Study

Any set of beliefs that guides action is a paradigm (Kuhn, 1962) that characterizes its adherents' view of the world in terms of their ontology, epistemology and methodologies, in order to go about the process of exploring it (Guba, 1990). The methodological design of this research project took the research goal and research questions into consideration, mapping out a triangulation approach, with both qualitative and quantitative methods being used (see Figure 3). As knowledge management is a dual paradigm by nature (Gloet & Berrell, 2003) and this research project studied the humanistic side as well as the technical side, both quantitative data and qualitative data were required to address these issues.



Figure 3 - An Overview of this Mixed Methods Sequential Design

The research questions in this study examined four aspects (behavioral, cultural, social and technological) of knowledge practices in the daily work activities of university press employees and managers. This project sought to understand the behavioral, cultural, social and technological aspects of the knowledge practices associated with university presses in the publishing sector. The behavioral aspect related to the range of actions taken by individuals or groups in knowledge practices. The cultural aspect encompassed the ethnicity and the organizational culture behavior in knowledge transfer and practices. The social aspect concerned norms and informal social behaviors related to sharing knowledge, and the technological aspect was the innovation adoption involved in knowledge practices.

#### **Research Questions and the Focus Addressed in this Study**

The research questions for this research project were:

- RQ1: What factors influence individuals' knowledge-seeking and knowledge-sharing in knowledge-intensive firms such as university presses?
- RQ2: In what ways do knowledge-intensive firms like university presses apply and retain their operational knowledge in their organizations?
- RQ3: What types of knowledge barriers do university presses encounter in their daily processes?

To understand these issues, the following lists the data types required to address each research question.

RQ	Quantitative Data	Qualitative Data
RQ1	Survey – Self-Selection Sampling	Interview – Purposive Sampling
RQ2	Survey – Self-Selection Sampling	Interview – Purposive Sampling
RQ3	Survey – Self-Selection Sampling	Interview – Purposive Sampling

Therefore, both quantitative data and qualitative data were combined for this research, as in other mixed methods research studies (Morse, 2003). The mixed method research approach allowed the researcher to investigate the research problems with a wider scope (Creswell, 2008). This design type provided statistics and stories that may complement or contrast with each other to inform the exploration of the research questions in this social science inquiry, for a "better understanding" of the inherent complexities of human phenomena (Watkins & Gioia, 2015).

The sequential design was chosen for this research in order to have a generalized understanding of the four categories (behavioral, cultural, social and technological aspects) in knowledge sharing, via the more in-depth approach of critical incident methods. The research process began with collecting and analyzing quantitative data, followed by collecting and interpreting the qualitative data, as a follow-up to the quantitative results. The purpose of this sequence was to use the qualitative data to enhance the explanation of the quantitative results that needed further exploration. This sequence also permitted the use of quantitative results purposefully to select the best direction for the subsequent qualitative research (Watkins & Gioia, 2015).

The researcher used two phases to implement the quantitative and qualitative studies in the research process, in order to explore any similarities or differences in these four aspects (behavioral, cultural, social and technological) of knowledge practices among employees and directors. Using this approach, findings from the qualitative data could possibly corroborate or explain any quantitative findings that presented an unanticipated outcome (convergence or divergence) of the study where a mixed methods study was undertaken (Dolye, Brady & Byrne, 2016).

The researcher collected quantitative and qualitative data simultaneously, analyzed the two datasets separately, and then interpreted the findings by comparing the results. Therefore, the sequential design was appropriate for this dual paradigm nature of social science research when the researcher sought a more complete understanding. This design was intuitive, with a clear distinction maintained between the qualitative and quantitative methods (Watkins & Gioia, 2015).

A quantitative approach was used to study the knowledge practices among all employees in a bottom-up style, to gain an understanding of the population. The subsequent qualitative approach employed a top-down style for senior management personnel, to explore the research topic from different levels. The use of qualitative research is "to understand and represent the experiences and actions of people as they encounter, engage and live through situations" (Elliott, Fischer & Rennie, 1999). It was interesting to see if there were any similarities or differences in employees' and directors' perspectives, and how their knowledge seeking began when problems arise. In fact, this mixed methods approach is frequently referred to as triangulation, mixed methodology, multi-strategy research or integrated methods (Denscombe, 2007), to incorporate both qualitative and quantitative research methods to understand and provide insight into the research problems (Neuman, 2006).

# **Research Design - Mixed Methods Sequential Design**

The research design for this study featured a mixed methods methodology approach involving both qualitative and quantitative components (see Figure 4). The focus of this sequential study design was to examine behavioral, social, cultural and technological aspects in knowledgesharing and knowledge-retention practices in knowledge-intensive firms, specifically university press organizations. The quantitative design provided more generalizable data, while the qualitative research illuminated individual experiences of participants (Patton, 2002). Therefore, the qualitative methods complemented the quantitative analysis in this research project (Hunt, 1994). The rationales for choosing this approach were: 1) combining research strands offset their weaknesses to draw on the strengths of both, and 2) bringing together a more comprehensive picture of the phenomenon through both quantitative and qualitative research (Harrison & Reilly, 2011). The ultimate goal was to use the findings of this research to develop a knowledge management audit tool that would allow business enterprises to perform periodic self-evaluations of their knowledge practices. In the sequential design, both quantitative and qualitative data were collected to play mutual supporting roles for each other or to play a supporting role in a larger design (Harrison & Reilly, 2011). This chapter, therefore, presents the research procedures that were used by the researcher to perform data collection.

Step 1	Quantitative Design Strand: • State quantitative research questions and determine the quantitative approach. Collect the Quantitative Data: • Obtain permissions from AUP, IRB • Statistical sample analysis • Collect close/open-ended data with online questionnaire as an instrument.	Analyze the Quantitative Data: • Analyze the quantitative data using statistics tests.				
Step 2	Qualitative Design Strand:         • Determine the qualitative approach: interviews.         • Prepare interview guide         Collect the Qualitative Data:         • Obtain AUP/ IRB permissions. Recruit samples via Informant.         • Qualitative sample analysis.         • Collect open-ended data with protocols.	Analyze the Qualitative Data:     Analyze the qualitative data using coding, procedures of theme development.				
Step 3	Mix the Two Sets of Results: •Identify content areas represented in both data sets and compare, contrast, and/or synthesize the results in a table. •Identify differences within one set of results based on dimensions within the other set and examine the differences within the dimensions. •Develop procedures to transform one type of result into the other type of data (e.g., turn themes into counts). Conduct further analyses to relate the transformed data to the other data (e.g., conduct analysis such as thematic counts).					
Step 4	Interpretation: <ul> <li>Summarize and interpret the results</li> <li>Discuss to what extent and in what ways results from these two types of data converge, diverge, related to each other, and/or produce a holistic understanding.</li> </ul>					

Figure 4 – Flowchart of the Procedures in Implementing the Mixed Methods Sequential Design

#### Philosophical Assumptions and Research Design for the Quantitative Portion

This quantitative research approach examined the knowledge sharing relationship among the variables (behavioral, cultural, social and technological aspects). These variables were measured using instruments so that the numerical data could be analyzed using statistical procedures (Creswell, 2014). The approach deals with research problems using mathematical and statistical techniques to identify facts and causal relationships, and adheres to the practices and norms of the natural scientific model (Fitzgerald & Howcroft, 1998). This approach is deductive in nature, as it focuses on developing theories, operationalizing concepts and subjecting them to empirical testing.

The focus of the quantitative research design was to identify causal relationships between the independent variables and the dependent variables. A non-experimental survey design with standardized questionnaires was used to measure thoughts, attitudes, feelings or behaviors of samples draw from a population. The choice and design of a survey instrument, with the focus on the research goals, is to access in a descriptive way the attributes of subjects, or to test theories in an explanatory way, using a structured questionnaire as part of the study plan (Gill & Johnson, 1997).

For this research study, because of the subjects' being in geographically-dispersed locations, and with a limited research budget, the researcher used an online survey questionnaire to collect data. The aim of the survey was to provide a general view of the employees' overall orientation in their behavioral, cultural, social and technological aspects. This survey was particularly used to understand the causal relationship between the variables: *organizational size* and *knowledge sharing*.

# Data Collection and Procedures for the QUAN Study - Surveying

# Survey Development

Stage One: Review Other Surveys and Design Survey Instrument

During the first stage of the survey development, the researcher conducted a literature review to learn from existing research about similar problems experienced by other researchers (Watkins & Gioia, 2015) using databases such as *Business Source Complete*, *Library & Information Science Abstract* and *ACM Digital Library*. Twenty knowledge management survey design articles across different disciplines were reviewed. Three knowledge management survey design articles were then selected from these twelve articles as a final list (see below) for in-depth reading. By referring to these three articles, a survey instrument was designed, with the goal of exploring four research aspects (behavior, culture, social and technology) for this dissertation.

- Andersen, A. (1996). The Knowledge Management Assessment Tool (KMAT). London: Arthur Andersen KMAT Study.
- Kulkarni, U., & St Louis, R. (2003). Organizational self-assessment of knowledge management maturity. *AMCIS 2003 Proceedings*, 332.
- Yuan, Y. H., Wu, M. H., & Lee, J. C. (2012). Examining the role of knowledge transfer effect as a mediator variable among impact factors in knowledge innovation. *International Journal of Business and Information*, 7(2), 205.

The researcher contacted the Association of University Presses in November 2017 for their approval to use their members as respondents. The Association preferred to review the survey questionnaires prior to their distribution via the Association. Therefore, working with a restricted and indirect interaction mode between the researcher and the population, the sampling method was then determined to be self-selection sampling, for the privacy of the research subjects, as well as for the convenience of data collection from a geographical-dispersed population. Although a pilot test would have improved the internal validity of the questionnaire, there was no guarantee of the

effect of the pilot study on the success of the full-scale survey (Van Teijlingen & Hundley, 2001). Another concern was to decide if the pilot study participants should be included in the main study, as they had already been exposed to an intervention and therefore might respond differently from those who had not previously experienced it. It was also impossible to exclude these pilot study participants, because it would have resulted in too small a sample in the main study (Van Teijlingen & Hundley, 2001). Because of the university presses' academic affiliation and geographical dispersion, it was difficult to find a similar population with such uniqueness, in order to conduct a pilot test. Therefore, the researcher chose not to do the pilot test. Instead, to establish the face validity of this survey instrument, a group of five librarians who had worked closely with the University of Tennessee's University Press on a regular basis were invited to comment on the questionnaire and to evaluate whether the questions effectively captured the topics under investigation.

# Stage Two: Determining the Sample Size using Statistical Prior Power Analysis

The second stage was to determine the required sample size (Krejcie & Morgan, 1970). Sampling is a strategy used in selecting elements from a population, with the goal of providing a practical mechanism to enable extrapolation from a sample to a population. The determining of a sample size is an important and difficult step in planning an empirical study (Dattalo, 2008). Increasing the sample size provides higher statistical power for the analysis. However, a large sample size may simply waste time and resources for minimal gain. Therefore, using a statistical power analysis can estimate an optimal sample size beforehand to ensure the analysis is meaningful. The AUP has 144 members across the globe and it would have been expensive and impractical to survey every employee in all member presses. Therefore, the statistical priori power analysis technique helped to estimate the optimal sample size needed for making the study worthwhile, particularly for a self-selected sample with no prior knowledge of the total number of volunteers.

Power analysis is the procedure that allows quantitative researchers to determine the sample size needed to enable statistical judgments that are reliable (Williams & Zimmerman, 1989). It is the probability of detecting a "true" effect when it exists, and to avoid catching a false null hypothesis, i.e. a Type II error. A priori analysis estimates the sample size based on acceptable levels of effect size and power. To conduct a power analysis, a common choice for the significance level in research is alpha = 0.05. The following sample size formula is used to arrive at a representative number of respondents when the population estimate is known (Godden, 2004):

 $n = \frac{Z^2 \times p (1-p)}{M^2}$ 

Where:

n	=	Sample size
Ζ	=	Z value (e.g. 1.96 for a 95% confidence level)
Р	=	population proportion (expressed as decimal) (assumed to be 0.5 (50%)
М	=	Margin of Error at 5% (0.05)

The confidence level measures the reliability of the data, i.e., the percent indicates how confident the researcher is that the results are correct. The confidence level corresponds to a Z-score. Typical choices are 90% or 95%. Here are the z-scores for the most common confidence levels (90% - Z Score = 1.645, 95% - Z Score = 1.96).

The confidence interval is about the margin of error that the researcher can tolerate. The population size is the total number of people for which the researcher can generalize the results. The member presses did not disclose their total numbers of employees on their websites, nor was this count reported to the Association of University Presses. Therefore, the calculation of the sample size estimation for this research project was based on the addition of the numbers of email

registrations on the two AUP listservs (the General listserv and the Directors listserv).

In other words, the population size was set to 1,347. That was the total number of names listed on the two AUP listservs – the "General" list had 1,200 email addresses and the "Directors" list had 147 email addresses. For the default significance level of this research, the standard deviation for how much variance the researcher expected in the responses was set to 0.05.

This survey used the University's Qualtrics tool as the online platform to collect survey responses. Besides using the formula to calculate the estimated sample size, the sample-size calculator on the Qualtrics webpage was also used to cross-check the ideal sample sizes with the different confidence levels, the same population size (1,347) and a 5% to 8% margin of error. Thus, an estimated sample size of 299 was required on a 95% confidence level with a 5% margin of error, or sample size of at least 98 was required on a 90% confidence level with an 8% margin of error.

	Confidence Level = 95%			Confidence Level = 90%				
	Margin of error			Margin of error				
Population Size	5%	6%	7%	8%	5%	6%	7%	8%
1347	299	223	172	136	226	165	126	98

After the survey was conducted, a total of 123 completed surveys were obtained. After the data clean-up process was completed, the actual usable sample size for this research was 107. Therefore, a 90% confidence level with 8% margin of error was set for this research.

### Survey Instrument

A questionnaire design was used for this study to assess the four aspects of knowledge sharing among university press employees: behavioral, cultural, social and technology. The selfadministered survey allowed for data collection from respondents with a focus on specific issues (Thayer-Hart, Dykema, Elver & Schaeffer, 2010) and addressed a vast array of objectives (Brick, 2011). The questions posed were predominantly close-ended questions, with a finite set of answers provided and with scales to measure responses. A five-point Likert scale was employed for questions measuring various aspects of knowledge sharing. The responses options were labeled as "Strongly Disagree, Disagree, Neither Agree or Disagree, Agree and Strongly Agree" for more clarity, rather than using numeric labels (Thayer-Hart et al., 2010, p. 10). Open-ended questions were used for several measures to accommodate responses that might not have been anticipated (Thayer-Hart et al., 2010).

Qualtrics (an online software subscribed to by the University) was used as the platform to deploy and manage the survey. This tool provides a password-protected individual account for data collection, and the data can only be accessed by the researcher. The survey questionnaire had the informed consent page as Question 1, and the rest of the instrument was divided into two sections, with 25 questions in total. The first section had five questions on the key respondents' professional backgrounds, in order to identify that each respondent was qualified to take the survey. This safeguard assured that the survey was based on a representative sample, to support the validity of the survey and to evaluate any possible threat to the survey's validity. The five questions in the first section were about:

- The geographic region of the respondent's organization in (North America, South America, Europe, Africa, Asia, Others)
- 2) The participant's gender type
- 3) The participant's age
- The number of years that the participant has worked in the organization, and the nature of his or her job

45

#### 5) The participant's experience as a supervisor of other employees

The second section of the questionnaire consisted of 20 close-ended questions. Some questions had fill-in options and some questions provided answer choices using a five-point Likert scale system. There were five questions in each of the four different categories: behavioral, cultural, social and technological. Some questions consisted of several sub-sections about knowledge seeking, knowledge sharing and retention. The questions in the survey instrument were limited to a moderate and appropriate length, to avoid creating fatigue in the respondents, while still generating sufficient measurements of the study constructs (Fife-Schaw, 2006). The full survey questions can be found in Appendix D.

# Respondents

This study was conducted with employees at all position levels, from the 144 Association of University Presses' member organizations, who were subscribed to at least one of the AUP's two moderated listservs - the General list and the Directors list - that are closed to outsiders. Selfselecting sampling was used, so their participation was voluntary. The respondents were all over 18 years old and included both genders. The AUP requested the right to review the survey questions before distributing the survey through the listservs on behalf of the researcher, as they would not provide the actual list of their subscribers to an outsider.

Therefore, the sampling frame was derived from the two email listservs monitored by the AUP's Director of Research and Communications, with the following criteria: (a) all employees (1,200) who had already voluntarily signed up for the General email listserv, and (b) the directors (147) who had already voluntarily signed up for the Director email listserv. The population (1,347) of the study was determined from the total number of names on the two listservs combined, as provided by the AUP.

An internet-based survey method was used for delivering the survey to collect quantitative data about this specific population, because it was the most effective and efficient way to provide all members of the defined population with an opportunity to respond. As the member presses are located in different regions across the globe, it was not feasible for the researcher to administer the survey through direct contract or via postal mail or telephone. The use of the Internet facilitated the participation of the individual employees of the presses, to access and complete the survey.

On behalf of the researcher, the AUP's Director of Research and Communications sent out the survey invitation email with the survey link on February 7, 2018 to her constituents on the two listservs. A reminder to encourage participation was sent out on these two listservs via the same director on February 14, 2018, two days before the survey was closed. After the survey was conducted, 123 completed responses were collected. The common characteristic across all of the respondents was that they were listserv subscribers employed by an AUP member organization, and were contacted using their university press email addresses. The collected dataset was then cleaned up and analyzed using the statistical software SPSS through a subscription provided by the University of Tennessee. Different types of statistical tests were performed. The findings were presented in tables to provide the snapshots of the research results. The details of the quantitative survey data analysis and findings are presented in Chapter Four.

#### *Compliance*

Prior to collecting the data for the study, the researcher completed the required online training and obtained a certificate from the Institutional Review Board (IRB) to ensure compliance with the institution's policies and procedures for conducting ethical research with human subjects. After the survey questions were designed, the researcher sent the questionnaire, the invitation to take the survey and a description of the data collection procedure to the

47

University of Tennessee's Institutional Review Board in December 2017 for their approval, to ensure that the ethical standards for respect, justice and participant benefits were upheld by the study. The IRB approved the study in January 2018, prior to the initiation of data collection on February 7.

A critical step in the study was to obtain the informed consent of every participant to certify that they understood and acknowledged the benefits and risks associated with the survey. As the survey was self-administered online, the informed consent (Appendix A) included a statement ensuring the confidentiality of the responses and was presented as Question 1 on the homepage of the survey. Participants taking the survey had to click on the button labeled "Yes, I agree" in order to continue to the demographics section questions associated with the survey. Participants were also assured that they could withdraw from the study at any point in the survey if they wished to do so.

This web survey mode of data collection provided a convenient, cost-effective and secure (regarding privacy) method for obtaining responses from participants (Couper, 2011). It also provided for an easy extraction of the data into data analysis software such as *Statistical Package for the Social Sciences* (SPSS) (Carbonaro et al., 2002). The key point is that research subjects self-selected to take part in the research and completed the survey voluntarily. They were not approached directly by the researcher.

# Philosophical Assumptions and Research Design for the Qualitative Portion

A qualitative research approach tries to explore and understand the meanings individuals ascribe to a social problem. Data are typically collected in the participant's setting, using methods that allow for emergent questions and insights. Qualitative data analysis involves an inductive process that generates themes that evolve from the particular to more general levels. The researcher interprets the meaning of the data (Creswell, 2013). Social constructivism or interpretivism underlies qualitative research approaches (Lincoln & Guba, 1985). The basic generation of meaning arises during the interaction with a human community. The value of this type of research approach is the inductive process with a focus on the specific situation or people involved (Maxwell, 2009).

The philosophical assumption in qualitative research is mainly related to interpretivism. The interpretivist/constructivist paradigm grew out of phenomenological philosophy, and has the intention of understanding "the world of human experience" (Cohen & Manion, 1994), suggesting that "reality is socially constructed" (Mertens, 2005). The interpretive/constructivist researcher tends to rely upon the "participants' views of the situation being studied" (Creswell, 2013). Constructivists do not generally begin with a theory, but instead they "generate or inductively develop a theory or pattern of meanings" (Creswell, 2013) throughout the research process. Interpretivists contend that only through the subjective interpretation of and intervention in reality can that reality be fully understood. The study of phenomena in their natural environment is key to the interpretivist philosophy. They acknowledge that there may be many interpretations of reality, but these interpretations are in themselves a part of the scientific knowledge they are pursuing (Smith, Flowers & Larkin, 2009).

There are five different research designs for qualitative research, which encompass a continuum from a narrow to a broad focus (Creswell, 2013): narrative, ethnography, phenomenology, case study and grounded theory. For this research study, phenomenology was chosen because phenomenological research is the study of the ways a person's world is formed in part by the person who lives in it (Fischer & Wertz, 2002). This approach is concerned with the lived experiences of people regarding a phenomenon of interest, such as the participants'

experiences with knowledge practices, which is being studied for this research project. The emphasis is to describe personal perspectives and interpretations. An interview in some form is the main method used in Phenomenology research (Moustakas, 1994; Giorgi, 2009). For this study, in particular, Flanagan's (1954) Critical Incident Technique (CIT) was used as the interview process. The application of the CIT for the interviews was based loosely on the time-line approach used in Dervin's 1983 sense-making research. In individual interview sessions, the research participants were encouraged to talk about their past incidents instead of answering direct questions.

### Data Collection Procedure for the QUAL Study - Interviewing

For the qualitative research portion of data collection, open-ended questions for identifying critical incidents in participants' knowledge practices were used in the interviews. A purposeful sampling method was used to recruit participants by sending an invitation email to the key executives of AUP member presses, via the AUP's Directors listserv. The approval from the University of Tennessee's Institutional Review Board was obtained in February 2018, for the subject recruitment process, the invitation email, the interview guide and the consent form. The invitation email covered the goal of this research project, who was responsible for this research, any possible risk or benefits to participants, and that participation was voluntarily and anonymous. By the end of recruiting for the qualitative portion of the study, six directors replied to the researcher in order to participate in the first round of interviews, and two more directors responded to the researcher's invitation for the second-round interview.

#### Interview Development/Instrument

McCracken's (1988) long-interview research method, which uses the researcher as the research instrument, was used in this project. In the process of data collection through interviews,

the researcher serves as the instrument via social interaction, for entering into the world of the interviewee about the phenomena in question. For this study, the phenomenon of interest was critical incidents in their past experiences related to knowledge practices in their work setting. The process of observation and interviewing using appropriate questions had an important impact on this research. During the interview process, the researcher talked as little as possible so as not to impose her own concepts or judgments on the participant's stories, but instead she gently guided the participant through the discussion (Morrison et al., 2002). In order to guide the participants through the interview guide (Appendix B) for her reference during the interview process. The guide included an outline of the main issues to be explored using open-ended questions. The use of this interview guide allowed the researcher to remain free in building a conversation with the participants, while ensuring consistency in the kind of information that was obtained from the selected group of participants by asking everyone the same questions (Patton, 1990).

# Interview Guide/Protocol Development

A semi-structured interview guide (Appendix B) was developed to ensure consistency in the approach by which all of the interviews were conducted. A prepared interview guide/protocol serves "...to ensure that the same general areas of information are collected from each interviewee; this provides more focus than the conversational approach, but still allows a degree of freedom and adaptability in getting information from the interviewee" (McNamara, 2009). The semi-structured guide offered a balance between the flexibility of an open-ended interview and the focus of a structured ethnographic survey (Stuckey, 2013) in gathering focused qualitative textual data.

The interview questions were designed after conducting a review of the knowledge management literature, such as articles from *The Journal of Knowledge Management*. Flanagan's

Critical Incident Technique was adopted for questions in all sections to prompt the recall of personal experiences in memorable past incidents.

A pilot study is a small-scale preliminary study conducted to pre-test a particular research instrument (Van Teijlingen & Hundley, 2001). To ensure face validity, a limited pilot study was conducted with several librarians who had closely worked with the University of Tennessee's University Press to assess the timing of the interview, and which questions worked and which did not.

The structure of the Interview Guide (Appendix B) included four sections: Section A covered general information about the participant's access to and use of information and knowledge resources; Section B dealt with insights from the participant's experience – lessons learned, best practices, and conclusions and recommendations for consideration by others; Section C allowed for additional comments and feedback; and Section D served as a closing stage of the interview. Section A was mainly for warming up the social interaction and asking about personal experiences and critical incidents in knowledge seeking and sharing. Section B focused on the participant's perception about knowledge sharing in their office environment. Section C and Section D contained probes on any issues that the researcher missed or that participants wanted to share.

#### The Use of the Critical Incident Technique (CIT) in Interviewing

Interviews were conducted using John Flanagan's (1954) Critical Incident Technique (CIT) for qualitative data collection, which was designed to gather the participant's most memorable experiences. The CIT harvests descriptions of events that are remembered by users, and is widely used as a research technique for the identification of organizational problems (Zach, 2005). The technique can be customized by the researcher as needed, to suit the aims of the investigation. A critical incident is defined as "any observable human activity that is sufficiently complete in itself

to permit inferences and predictions to be made about the persons performing the act" (Flanagan, 1954). The CIT has been used in a variety of studies, including information-seeking behavior (Zach, 2005) in library science and communication (Radford, 1999) as well as in organization studies, focusing on factors affecting the performance of managers and employees (Breunig & Christoffersen, 2015). The critical incident data would be more precise and more usable than opinion poll data (Church, 2018).

During this study's interviews, participants were asked to describe from their perspectives a successful past knowledge-sharing incident, an unsuccessful past knowledge-sharing incident, and the factors that made each incident either successful or unsuccessful. They were also asked to provide their definition of "successful" and "unsuccessful," so the researcher could understand how the participants made sense of positive or negative incidents. This approach is useful in understanding the related behavior critical to complex situations, by collecting data for fact-finding and reflecting on professional practices (Hettlage & Steinlin, 2006).

Flanagan's Critical Incident Technique allows interview data to be sorted into patterns and then summarized descriptively (Radford, 1999). Using CIT to focus on specific incidents, it deemphasizes the inclusion of general opinions about management and working procedures. There are a variety of practical uses for developing and interpreting the research results (FitzGerald, Seale, Kerins, & McElvaney, 2008) by sorting interview data into patterns or relationships for summarization and description (Radford, 1999). In its application, the CIT provided procedures "for collecting direct observations of human behavior in such a way as to facilitate their potential usefulness in solving practical problems" (Flanagan, 1954).

# **Determining Sample Size**

In qualitative research, sample sizes should not be so small that it is difficult to achieve saturation, nor so large that it is difficult to undertake a deep case-oriented analysis (Onwuegbuzie & Leech, 2007). To estimate the number of samples (conducted interviews, in this case) needed for a non-probabilistic, purposive sampling approach, the standard criterion is theoretical saturation. This is the point at which no new information or themes are observed in the data (Mason, 2010), and serves to identify an adequate sample size in qualitative inquiry (Guest, Bunce & Johnson, 2006). To determine the qualitative power analysis, a literature search for guidelines on appropriate sample sizes led to three scholars' determinations of appropriate, specific sample sizes for different qualitative research approaches (Onwuegbuzie & Leech, 2007).

For phenomenological studies, Morse (1995) indicated that at least six participants are needed. Creswell (1998) recommended conducting between five and twenty-five interviews for phenomenological research. Kuzel (1992) recommended performing six to eight interviews for a homogeneous sample, such as the directors participating in this research project. Therefore, as an audit trail, the researcher stopped the first round of interviews after conducting the sixth interview, to review all transcripts. From a preliminary analysis on all six interview transcriptions, the researcher found that the emergent themes had hit the saturation point. To confirm that saturation had been achieved, the researcher interviewed two more directors in a second round of interviews that also included an experimental element, which was giving the definitions of "implicit knowledge" and "tacit knowledge" to the interviewees, prior to conducting the interview. This slight change for the final two interviews was done to test if a related emergent theme did exist, in order to conclude the qualitative data collection stage.

### **Interviewing Procedure**

After obtaining the permission of the AUP to conduct the research with its members, and obtaining the approval notification from the Institutional Review Board in February 2018, the researcher prepared the interview procedure. On behalf of the researcher, the AUP posted the recruiting email on the listserv for Directors in early May 2018. Six directors replied off the list to the researcher to schedule their interviews. A second round of interviews was needed, when two more directors agreed to participate, upon receiving the researcher's individual invitations sent to several directors in different countries. All participants (five males and three females, all over 18 years old) were from the United States. The profiles of the interviewees, including their location, their organization size and their publishing focus, are provided in Chapter Four.

Interviews were done from May to June 2018 prior to and after the AUP's 2018 annual conference. The interviews were conducted directly by the researcher at the interviewee's choice of location and were administered either by phone or online using a password-protected University of Tennessee Zoom account online meeting platform. In total, 60% took place on the phone, 20% were online using Zoom, and 20% were in-person interviews. Each interview lasted about 30 minutes. All interviews were audio taped with the interviewee's consent. The style of the semi-structured interview (Arksey & Knight, 1999) was used, with a prepared interview guide that served as a general guideline, as described in detail above.

Prior to the interview sessions, the interviewees signed and returned by email the consent form (Appendix A, about participating in the interview and agreeing to be recorded) to the researcher, via the University of Tennessee's Vault system. The Vault system is a secure filetransfer service that allows users to transfer files via encrypted HTTP and keep them securely in an encrypted data storage. All signed consent forms were then stored on a USB drive named "H". The drive was then locked in the researcher's office cabinet that was only accessible by the researcher.

Each interview proceeded through a series of open-ended questions, with the interviewer probing the answers and testing ideas from previous interviews, as appropriate (Strauss & Corbin, 1998). Minor changes to the interview protocol were implemented, as some issues emerged during the project. The interviewer had the flexibility to add, drop or reword the question sequence as necessary. After all interviews were conducted and taped, the audio recordings were transcribed by a paid transcriber who signed a non-disclosure agreement. The average interview lasted 30 minutes. For online sessions, no video was included. All interviews were audio taped and all interviewees gave their explicit consent prior to being interviewed and to being audio recorded. The consent form explained the study context and the possible uses of the data. The interview data were anonymized in a way that neither the respondents themselves, nor other people they referred to in the interview, could be identified.

Each interviewee was made aware of the opportunity to obtain a copy of their own interview transcript, upon their requesting it from the researcher, to ensure the accuracy of the interview and to provide any additional points if desired. The electronic copies of the interview and the transcription were securely stored on the USB drive "H" in the researcher's locked office cabinet that only the researcher can access. The interview transcripts contained no information allowing them to be linked back to the interviewee. To protect the privacy of the responding participant, the organization name was not addressed by the researcher during the interview and any organization names mentioned by the interviewees in the transcription were redacted before the analysis process. To protect the interviewees' privacy and for filing purposes, instead of using the interviewee's name and job title, the interview session was referred to as "DISS interview 1" (for

the first interview) and the interviewee was addressed as "Mr. [Initial]" or "Ms. [Initial]," and so forth.

After the interviews were conducted and transcribed, the transcriptions were uploaded into QDA Miner, a qualitative data analysis software subscribed to by the University of Tennessee. In the data analysis stage, a categorization scheme was developed to summarize and describe the data in a useful manner, without sacrificing its comprehensiveness, specificity and validity (Flanagan, 1954). A simplified open coding scheme for identifying and classifying the critical incidents and explanations of category placements was used in this research project. The coding scheme contained two major theme categories: Content Themes and Rational Themes, with several subcategories (Radford, 1999). The details of the qualitative data analysis and findings are presented in Chapter Four.

#### **CHAPTER FOUR**

# DATA ANALYSIS AND RESULTS

This chapter presents the data analysis process and the findings from both the quantitative and the qualitative research portions of the study. The quantitative research used a self-selecting survey questionnaire to collect data from AUP member presses' employees about different aspects of their knowledge practices. The qualitative research used purposive sampling, for which the researcher served as the instrument to collect data by interviewing AUP member presses' directors to assess critical incidents in their knowledge practices.

### **Quantitative Research Results**

# The Rate of Return

The electronic survey was completed by 123 of 1,347 possible participants. The rate of return, i.e., the proportion of questionnaires that were completed and returned (Antonius, 2003), was calculated as 9.13% of the population number (1,347 was the total number of subscribers on the two AUP listservs combined). Sixteen participants completed the informed consent page and the demographics section, but did not proceed to the survey. After these blank attempts were removed using the Listwise deletion (complete-case analysis) technique, the actual sample number was 107.

After further investigation using the respondents' IP addresses to identify their affiliated organization's size, one survey's IP address indicated that it was submitted from a location where the university press was closed in 2002. This record was removed because changes in technologies that have occurred since 2002 are likely to have had impacts on knowledge practices, suggesting that this particular submission was a faulty response that did not fit the parameters of this research project. All participants who answered questions presented after the demographics section were
included in all analyses using the pairwise deletion (available-case analysis) technique, even if they chose to skip some questions in other sections. Therefore, the ultimate sample size for the quantitative analysis was 106 completed surveys, yielding an adjusted rate of return of 7.87%. Given the fact that 90% of AUP's members are from North America, and that all completed surveys were from North America, the return rate could be calculated at as high as 8.7%.

### **Data Cleaning Process**

After the survey was conducted, the data cleaning and analysis processes of editing and coding (Zikmund, Babin, Carr & Griffin, 2008) were followed. The survey's raw data was downloaded from Qualtrics and imported into the University of Tennessee's data analysis software subscription platform - *Statistical Package for the Social Sciences* (SPSS). SPSS is a popular software for editing and analyzing all sorts of data in the social sciences (Suresh, 2015). Data were stored in a matrix that resembles a spreadsheet file with rows and columns. Each row represented a particular respondent's score on each variable, and the data in each column was set up as a separate, defined variable for which there was a value for every respondent.

The initial data cleaning, i.e., error checking and verification (Zikmund, Babin, Carr & Griffin, 2008), found that several participants did not fully complete the survey, dropping out at various questions. In order to preserve a larger effective sample size, a mixing of listwise and pairwise deletion (Peugh & Enders, 2004) was used to handle the missing data issue. The listwise deletion eliminated an entire sampling unit from the analysis, even if only a single response was missing. In contrast, the pairwise deletion allowed the data that the respondent did provide to still be used in the statistical analysis (Zikmund, Babin, Carr & Griffin, 2008).

Therefore, the listwise deletion technique was applied in the first cleaning cycle, because those responses had finished only the Demographics section. Therefore, surveys with all blank responses after Question 6 were treated as incomplete. These responses were removed from the analysis.

If there were any completed responses on a particular survey for the measures beyond Question 6, the pairwise deletion technique was applied to those surveys after the first cleaning cycle was done. The procedure did not include a particular variable when it had a missing value, but it did include the response in the analysis of all other variables with non-missing values. For example, if a case contains three variables: VAR1, VAR2, and VAR3 and has a missing value for VAR1, this would not prevent some statistical procedures from using the same case to analyze variables VAR2 and VAR3 that do have values present. Hence, after this handling of any missing-data issues, the total number of the responses to the survey was 107. In accordance with the guidelines of the statistical prior power analysis, having 107 cleaned-up and usable questionnaires exceeded the minimum required specification of 98, for a 90% confidence interval with an 8% margin of error.

When the survey instrument was designed, the main concern was to protect the respondents' privacy and to maintain full confidentiality. Therefore, no questions in the survey asked the individuals to identify themselves or disclose their organization names. However, in the later dataanalysis stage, the researcher wished to assess if there were any significant differences by organization size in the survey results. In order to determine this, an intermediate method was used to provide a basis for categorizing the responses by different organization sizes. First, the researcher submitted an amendment to the IRB for their approval of the amended procedures. After it was approved, the researcher created Spreadsheet "A" with a list of all university press members from the AUP website. The organization size of each member was defined, based on the number of employees listed on the member presses' websites and the AUP's 2017 and 2018 member directories. The university presses were then classified into three categories: small (less than 25 employees), medium (25 to 49 employees) or large (50 or more employees).

The researcher then located each IP address in the metadata section of each survey response by using the IP Checker tool (https://ipinfo.info/html/ip\_checker.php). The IP address was in the metadata section that was attached to each survey response and was used to identify the host name's ownership information and geolocation data, which allowed for the identification of the organization. There was no way to identify who filled out the survey or the serial number of the computer device the respondent had used. This was the point in the analysis when the one aforementioned survey was found, from the university press location that closed in 2002, and it was removed from the dataset. In total, the final sample size for analysis was 106.

Second, the researcher created a Spreadsheet "B" and downloaded only the IP addresses from the survey dataset into Spreadsheet "B". A column titled "organization size" was created in Spreadsheet "B" and the researcher keyed in variables of "large," "medium" and "small" for each survey response by matching each IP address and the corresponding university's organization size (i.e., large, medium or small) from Spreadsheet "A".

Third, the researcher created a variable column for "organization size" in the survey dataset inside SPSS and keyed in each field from each survey response with the corresponding values from the size groups "large", "medium" and "small" by matching the IP address in Spreadsheet "B." After this "organization size" column was filled, the IP address column in the SPSS data set was removed before the process of survey data analysis was started. Therefore, no IP address was still attached to the survey response prior to the initiation of the final data cleaning process, so the survey results were kept discreet and unidentifiable. After the data cleaning process was completed, the data analysis stage began, and the findings could then be reported in an aggregated format.

This survey questionnaire contained both close-ended questions on the participants' behavior related to knowledge practices, and some semi-structured open-ended questions such as Question 9, Question 10, Question 24 and Question 26, to probe their opinions using write-in answer options to give the participants an opportunity to reflect more deeply than with the provided answer choices. These written-in phrases as responses were reviewed for any typing errors and were coded into categories for the quantitative analysis in SPSS.

### **Developing a Data Codebook**

The SPSS data file generated a codebook (Appendix G) in order to communicate the research results clearly, and to make the data understandable for proper interpretation (Zikmund, Babin, Carr & Griffin, 2008). This codebook included information about each of the variables in the dataset, such as what label name was used to represent each variable, how each variable was measured (e.g., nominal, ordinal or scale), how each variable was actually recorded in the raw data (i.e. numeric, string), each variable's unit of measurement, and what category each variable represented.

Beside the variables originally stored in the SPSS data file for the close-ended structure questions, the researcher also performed the code-building process for the open-ended semistructured questions (Question 9, Question 10, Question 24 and Question 26) with variables such as organization size, work experience, job nature and write-in. The purpose of coding these questions was to transfer the meanings from the written responses into numeric codes accurately, by grouping individual responses into a few general categories of answers (Zikmund, Babin, Carr & Griffin, 2008). Therefore, an additional coding scheme for the questions with write-in options in this research was built.

# **Data Analysis Using SPSS**

After cleaning up the data, the researcher ran the SPSS software for statistical analysis. SPSS (Statistical Package for the Social Sciences) is a data management and statistical analysis tool, subscribed to by the University of Tennessee. SPSS was chosen because of its versatile dataprocessing capability that allows many different types of analyses, data transformations, and forms of output. It has been widely used in social science research (Bala, 2016).

In the context of small-scale survey evaluation like this research project, SPSS was used for electronically storing the questionnaire data in a spreadsheet-like table, calculating the frequency distributions of multiple-choice question responses, generating descriptive statistical data for question responses, such as frequency counts for closed questions, creating graphical presentations of questionnaire data for reporting purposes, exploring relationships between the responses to different questions, and collating the write-in responses.

The use of SPSS was employed to interpret the data, in order to use the results to answer the research questions for this study, listed below:

- RQ1: What factors influence individuals' knowledge-seeking and knowledge-sharing in knowledge-intensive firms such as university presses?
- RQ2: In what ways do knowledge-intensive firms like university presses apply and retain their operational knowledge in their organizations?
- RQ3: What types of knowledge barriers do university presses encounter in their daily processes?

A frequency analysis was run to identify initial data errors, including spelling errors, outliers, and missing data. The median, mean, and mode were compared to test the normality of the distribution of the data. Finally, descriptive statistics were run to present the direct findings and to prepare for the more in-depth analyses.

The researcher used descriptive statistics to analyze the responses for the knowledge sharing factors in the behavioral, cultural, social and technology sections. In addition, statistical tests were used in order to assess the statistical significance of any differences in the response levels to various measures.

### Results

The quantitative portion of this study was a survey made available to all levels of employees of all member presses of the Association of University Presses (AUP). The Association protected the listserv subscribers' privacy and would not provide their subscribers' email addresses to an outsider. As the Association's listservs are moderated and are closed to outsiders, at the researcher's request, the Association distributed the survey invitation on her behalf. The Association reviewed the survey questions before forwarding the participation invitation email to their subscribers.

The potential respondents were derived from the two email listservs monitored by the Research and Communications Director of the AUP, with the criteria: 1) all individuals (1,200) who had voluntarily signed up to the General email listserv, and 2) the university press directors (147) who had voluntarily signed up to the Directors email listserv. Therefore, the population (N= 1,347) of the study was determined based on the total number of names on these two listservs. After the data cleaning process was done, the number of valid, completed questionnaires was reduced to an n of 106. The structure of the questionnaire divided the measures into five sections

– demographics, behavioral factors, cultural factors, social factors and technological factors. Some respondents chose not to answer some questions in various portions of the instrument. Descriptive statistics were obtained for all questions, to provide a snapshot of the data for gaining meaning and insights (Watkins & Gioia, 2015).

Statistical tests were applied to determine if there were significant relationships between categorical variables. Chi-square tests were conducted (see Appendix F) on variables such as organization size and age range. The findings were not statistically significant. Questions in the cultural factor section were mostly Likert-type scales used for scaling survey responses. Therefore, in this section, the Cronbach's alpha test was applied to assess these questions for internal consistency. A composite score was then used to express how closely related the set of similar questions were as a group.

# **Profiles of the Respondents**

The survey link and the invitation email were delivered to the specific population by the Association of University Presses via their two closed listservs. Because these listservs are restricted to only press employees, no screening question was needed to check the qualification of the respondents. Instead, Question 1 of the survey was the informed consent page requiring their agreement. Those who answered "No" were automatically terminated from the survey.

Question 2 to Question 6 were about the respondents' demographics. Question 2 asked about the geographical region in which the participants were located. All respondents were from North America (n = 106 or 100%), with eight participants (7.6%) located in Canada and 98 participants (92.4%) from the United States. About 88 participants (83%) identified as females, 17 participants (16%) were males, while one participant (1%) selected "prefer not to answer" for Question 3 about gender. The high percentage of female participants aligns with the findings from other social science studies, that women are more inclined than men to participate in surveys (Smith, 2008). Question 4 about the respondents' age categories found that the majority of respondents (42 people or 40%) were older than 50 years old. Twenty participants (19%) were between 40 and 49 years old, 29 participants (27%) were 31 to 39 years old, and 15 people (14%) were between 18 and 30 years old.

Question 5 was a write-in question that asked about how long respondents had worked in their current positions, and about the general nature of their jobs. Many respondents indicated that they filled multiple roles in their organizations. In coding these write-in responses, the researcher honed in on the key role mentioned in their statements, to determine the main job classification category for the purpose of data analysis. The write-in data were coded into SPSS to facilitate statistical analysis. The most frequently-mentioned job titles were from Acquisitions (28 participants or 26.4%), followed by Marketing and Sales (22 participants or 20.8%).

In terms of their number of years of work experience in their current positions, the largest proportion of respondents had held their job from between one to five years (38 people or 36%), while 21% (or 22 people) had been in their jobs from between five to less than ten years. Interestingly, a smaller proportion of respondents had been in their jobs from ten to under 15 years (15 people or 14%) versus those who had held their jobs for 15 or more years (30 people or 28%). University presses had generally stable workforces, with only one participant having less than one year's experience on the job.

Question 6 sought to identify how many respondents had supervisory responsibilities, by asking about the number of staff members each participant supervises. The majority of respondents (65 people or 61%) were supervisors of other employees, while 41people (39%) did not have supervising duties. About half of all respondents (52 supervisors or 49%) were responsible for

supervising fewer than ten staff members, while only 13 supervisors (12%) supervised between ten and 50 staff members. No respondents supervised more than 50 people (See Table 1\_a).

By using each submitted survey's metadata, in a process described in detail earlier, each respondent's affiliated organization size was identified. Three types of organization sizes were defined: small (fewer than 25 employees), medium (between 25 to 49 employees) and large (50 or more employees). The analysis reveals that, out of 106 respondents, the majority of the respondents worked in small organizations (55 people or 52%), followed by 33 (31%) working in medium-sized organizations, and 18 (17%) employed by large-sized organizations (See Table 1\_b).

Variable Name	Category	Frequency	%
Region	North America	106	100
	South America	0	0
	Africa	0	0
	Europe	0	0
	Asia	0	0
	Others	0	0
Gender	Male	17	16
	Female	88	83
	Prefer-not-to-specify	1	1
Age	Above 18 and younger than 30	15	14
	31 – 39	29	27
	40 - 49	20	19
	50 and above	42	40
Job Nature	Administration	12	11.3
	Editorial	9	8.5
	Acquisition	28	26.4
	Design & Production	18	17
	Marketing & Sales	22	20.8
	<b>Business Services</b>	9	8.5
	Technical Services	6	5.7
	Non-disclosed	2	1.9

Table 1\_a: Demographics (N = 106)

Variable Name	Category	Frequency	%
Years Working for the			
Press	Less than one year	1	1
	1 year to less than 5 years 5 years to less than 10	38	36
	years 10 years to less than 15	22	21
	years	15	14
	Over 15 years	30	28
# of Staff Supervised	Less than 10	52	49
	10 to 50	13	12
	51 to 100	0	0
	More than 100	0	0
	I do not supervise	41	39

*Table 1\_a: Demographics (N = 106)* (Cont'd)

Table 1\_b: Demographics (N = 106)

Variable Name	Category	Frequency	%
Organization Size	Large (>50)	18	17
	Medium (25 - 49)	33	31
	Small (<25)	55	52

# Research Question 1: Factors Influencing Individuals' Knowledge-Seeking and Knowledge-Sharing in University Presses

The research survey was structured with sections focusing on behavioral, cultural, social and technology factors. Research Question 1 is the foundation for this study. Therefore, factors influencing the knowledge behavior of individuals were examined through multiple questions across these four focus areas, because all of these areas are influential to some degree in shaping individuals' knowledge practices.

In general, the most influential factors on university presses' employees' knowledge-seeking and knowledge-sharing practices were identified as the existence of an open culture, accompanied by high individual self-motivation to share, and a healthy level of trust among coworkers. The following exposition provides the statistical findings from each survey question that is related to this research question.

In the behavioral factors area, Question 7 explored how the respondents acquired the skills and expertise needed to perform their jobs. The most frequently-occurring answer was through self-learning (38 people or 35.8%). About a third of respondents (35 people or 33%) indicated that they acquired skills and expertise from colleagues, either supervisors or coworkers in the organization. About a fifth of respondents cited prior experience, having already learned skills at their previous job (21 people or 19.8%). Formal sources of learning were selected by relatively few respondents (seven people or 6.6%). Only five people (4.7%) chose "elsewhere" (See Table 2).

Category	Frequency	%
From supervisors or coworkers in this organization	35	33
Through self-learning	38	35.8
Through formal training	7	6.6
At my last job	21	19.8
Elsewhere	5	4.7
Total (N)	106	100

Table 2: (Q7) How did you acquire the skills and expertise that you use in your job?

To investigate how frequently the university press staff interacted with other members of their organization, Question 8 addressed how often respondents need to ask for information from their coworkers in order to perform their daily jobs. There were 105 responses to this question. Almost half of the respondents stated that they have to ask their colleagues for information on a daily basis (49 people or 47%), and another 38% (40 people) do so occasionally. Eleven people (10%) seek information from their coworkers once a week, while one person said he/she sought out colleagues for information once a month. Only four people (4%) indicated that they "never" seek information for their job from coworkers (See Table 3). Therefore, the knowledge-sharing process among university press staff occurs daily.

Category	Frequency	%
Never	4	4%
Occasionally	40	38%
Once a week	11	10%
Once a month	1	1%
Daily	49	47%
Total (N)	105	100%

*Table 3: (Q8) How often do you need to ask for information from your coworkers to perform your daily job?* 

In the culture focus section, Question 12 used several statements to examine the participants' perceptions about their reasons for or motivations to share knowledge. Overall, all five statements in this measure showed high frequency levels for the top two positions on the scale, which were "strongly agree" and "agree." The two reasons that drew the strongest level of agreement were that knowledge sharing was perceived to be important (95%), and that respondents were happy to share knowledge to improve the organization's daily operations (97%). Respondents also agreed with the knowledge-sharing reasons that they were important in their organizations (87%), and that they had many connections in their organizations with which to share (80%). Respondents did agree with the idea that they wanted their supervisors to see them as good employees, but their reaction to this reason was less enthusiastic (56%) (see Table 4).

Categories	Strongly Disagree	Disagree	Neither Agree nor Disagree	Agree	Strongly Agree	Total (N)
I am an important part of my organization's network	4	2	6	44	39	95
%	4%	2%	6%	46%	41%	100%
I have many connections in my organization to share knowledge	3	5	11	46	30	95
%	3%	5%	12%	48%	32%	100%
I am happy to share my knowledge at work in order to improve my organization's daily operations	3	0	0	29	63	95
%	3%	0%	0%	31%	66%	100%
Knowledge sharing is important	3	0	2	22	68	95
%	3%	0%	2%	23%	72%	100%
I want my superior to think I am a good employee	2	8	32	29	24	95
%	2%	8%	34%	31%	25%	100%

Table 4: (Q12) In my organization, I share my knowledge with co-workers because

Question 13 explored the perceptions of the university press staff members, about whether their management encourages employees to share work knowledge and experiences, by providing informal activities and opportunities to communicate. Ninety-five people answered this question. A high percentage of respondents agreed that their management created circumstances conducive to knowledge sharing during informal activities. Twenty-five people (26.32%) strongly agreed, and 48 people (50.53%) agreed with this statement. However, not all respondents felt that their management facilitated knowledge sharing in informal ways; 12.6% disagreed or disagreed strongly with this statement. (See Table 5).

*Table 5: (Q13) My organization's management encourages informal activities and opportunities to communicate/share experiences and work knowledge.* 

Category	Frequency	%
Strongly agree	25	26.32
Agree	48	50.53
Neither agree nor disagree	10	10.53
Disagree	9	9.47
Strongly disagree	3	3.16
Total (N)	95	100

Question 14 asked the respondents about their perceptions of formal activities offered by their organizations for sharing knowledge and experience. The same number of respondents completed this question as for Question 13, but the percentage answering "strongly agree" dropped and the percentage selecting "agree" increased. However, almost twice as many respondents disagreed that their organizations provided formal knowledge-sharing activities (23.2%), as compared to the disagreement level for informal activities (12.6%) measured in Question 13. (For full results, see Table 6).

Table 6: (Q14) There are formal activities such as training sessions, forums and meetings in the organization to share knowledge and experience.

Category	Frequency	%
Strongly agree	16	16.84
Agree	54	56.84
Neither agree nor disagree	3	3.16
Disagree	15	15.79
Strongly disagree	7	7.37
Total (N)	95	100

Two-part statements were asked in Question 15 to examine the respondents' perceptions of the value of lessons learned. One statement of the pair asked about learning from failed projects, while the second statement assessed learning lessons from successful projects. Ninety-five respondents answered this question. The frequency counts and percentages for each statement under each response category were the same. The highest frequency counts (54%) were for "Agree," followed by over a quarter of respondents (27%) who expressed strong agreement. These results indicated that most university press staff valued the lessons learned from the projects in their organizations, regardless of whether the projects were successful or failures (See Table 7).

Categories	Strongly Disagree	Disagree	Neither Agree nor Disagree	Agree	Strongly Agree	Total (N)
Lesson learned from failed projects are considered valuable	6	3	9	51	26	95
%	6%	3%	9%	54%	27%	100%
Lesson learned from successful projects are considered valuable	6	3	9	51	26	95
%	6%	3%	9%	54%	27%	100%

Tabl	le 7:	(Q15)	In my	organi	zation
------	-------	-------	-------	--------	--------

Another two-part statement was used for Question 16, to explore the staff's perceptions of their coworkers' willingness to share. There were 95 responses to this question. The results for these statements showed interesting findings. A high number of respondents considered that their coworkers in their organizations were extremely likely or moderately likely to share willingly with each other about their work experience on a regular basis. However, for the paired statement about their organization's willingness to recognize publicly or reward activities associated with lessons learned, more responses shifted over to moderately likely and neither likely nor unlikely. There is an implication that staff were willing to share about their work experiences with others, but they perceived that relatively few university presses had implemented reward systems to encourage this kind of behavior (See Table 8).

Categories	Extreme Likely	Moderate Likely	Neither Likely nor Unlikely	Unlikely	Extreme Unlikely	Total (N)
Co-workers routinely share work experience with each other	41	43	7	1	3	95
%	43%	45%	7%	1%	3%	99%
Activities associated with lessons learned are recognized publicly and/or rewarded by the management	9	34	28	17	7	95
 %	9%	36%	29%	18%	7%	100%

Table 8: (Q16) There is a willingness in my organization that

Question 17 examined the social norm of the university presses. There were 95 responses to this question. About 62% of the respondents indicated that staff do engage in social activities after work hours, while only about 26% said their organizations do not socialize after work. (See Table 9).

Category	Frequency	%
Strongly agree	11	11.58
Agree	14	14.74
Do not know	11	11.58
Disagree	56	58.95
Strongly disagree	3	3.16
Total (N)	95	100

*Table 9: (Q17) Do you agree that staff, in your organization, do not have social activities after work hours* 

Question 18 looked at member's perceptions of whether staff share about work experience at casual gatherings in the office. Ninety-five respondents answered this question. More than 83% of respondents (15.79% strongly agree and 67.39% agree) stated that their co-workers shared work experience in the hallways or in casual gatherings (See Table 10).

Category	Frequency	%
Strongly agree	15	15.79
Agree	64	67.37
Do not know	10	10.53
Disagree	6	6.32
Strongly disagree	0	0
Total (N)	95	100

*Table 10: (Q18) Do you agree that staff, in your organization, are eager to share work experience in the hallways or in casual gatherings* 

Question 20 asked respondents about their own degree of motivation to share their knowledge and experience, and 93 responded to this question. Close to two-thirds of these respondents (62.4%) indicated they were very motivated to share, because their sharing would build a better company. Another 28% of the respondents indicated that they felt an average level of motivation, because they wanted to help their coworkers (See Table 11). This result indicated that a general culture of collegiality exists in the university press organizations.

Category	Frequency	%
Very unmotivated because I keep my job secret to protect my job	1	1.08
Unmotivated because I do not care what is going on in the company	0	0
Motivated on average because I want to help my co-workers	26	27.96
Motivated because I think my co-workers need me	8	8.6
Very motivated because I think sharing mine will build a better company	58	62.37
Total (N)	93	100

Table 11: (Q20) How motivated are you to share your knowledge and experience?

Question 21 was used to assess the level of trust among the staff. This question contained three statements and garnered 91 responses. Overall, a very high number of respondents either strongly agreed or agreed with these three statements. These results indicated that there was a strong level of trust among employees in the presses.

Sixty-four percent of the respondents agreed and 31% strongly agreed that they found their coworkers' information to be trustworthy. For the statement about whether their co-workers would help them out with their problems when asked, 46% of respondents agreed and 33% of respondents strongly agreed with that assertion. Forty-one percent of respondents agreed and 48% of respondents strongly agreed with the statement that their co-workers were both willing to help and would not deceive for their own profit.

However, the counts from the "agree" and "strongly agree" columns for the first two substatements dropped, while the counts for the neutral position increased. This shows that the respondents trusted that their coworkers would help them, but they had doubts that their coworkers could come to rescue them. When this answer aligned with a write-in response from Question 26, it indicated that only one or two staff members have the expertise to do the task. The increased counts for the "strongly agree" response for the last sub-statement, in comparison to the level for the previous two statements, indicated that most respondents believed in the sincerity of their coworkers (See Table 12). This measure generally reflected a good level of trust among coworkers.

Categories	Strongly Disagree	Disagree	Neither Agree nor Disagree	Agree	Strongly Agree
I feel trustworthy with my coworkers ' information	1	0	4	58	28
%	1%	0%	4%	64%	31%
If I encounter problems in finishing my job, I know my coworkers will help me out when I ask	7	0	12	42	30
%	8%	0%	13%	46%	33%
My co-workers are willing to help and do not deceive for their own profit.	0	3	7	37	44
%	0%	3%	8%	41%	48%

Table 12: (Q21) Do you agree or disagree with the following statements? (N = 91)

# Research Question 2: Ways that University Presses Apply and Retain their Operational Knowledge in their Organizations

Research Question 2 was posed to explore the practices, in terms of procedures or technological means, for knowledge curation among university press staff. Therefore, survey Question 9 from the behavioral focus section asked respondents where would they look for the knowledge that they needed to do their jobs. One hundred respondents answered this question. Forty-six percent of the respondents chose to write in their answers. For the rest of the respondents, 24% of them said they would ask their team members, 18% of them chose their organization's central information system, 9% of them would use their own computers or hard drive, and only 3% of them would use paper-based documents (See Table 13).

*Table 13\_a: (Q9) Where would you look for knowledge that you need to do your work?* 

Category	Frequency	%
In paper-based documents	3	3%
In our team/department's members' minds	24	24%
In our central information system	18	18%
On my personal or workstation computer/hard drive	9	9%
Write-in response	46	46%
Total (N)	100	100%

A coding scheme was developed in order to analyze the write-in responses. The following table shows the coding scheme used and the counts for each code. Among the 46 write-in narratives, the highest count was for "all of the above" (57% of the 46 write-in narratives). This was followed by both "online searching only" and the "AUP listserv, AUP friends/mentors, The Scholarly Kitchen;" each of these answers was written in by 13% of respondents.

Comparing those respondents who chose one item to those wrote their own narratives, knowledge seeking through coworkers and the AUP were the most popular choices.

Table 13 b: (Q9) Write-In Answers

Cod	Pagnonga Tyma	#	0/
e	Kesponse Type	Count	/0
1	Ask coworkers only	1	2%
2	Online searching only	6	13%
3	Ask coworkers and self-document search	3	7%
4	AUP listserv, AUP friends/mentees, The Scholarly Kitchen	6	13%
5	Coworkers or outside the organization such as universities	1	2%
6	The organization's shared drive	1	2%
7	Team member's mind, central information system and my hard drive	2	4%
8	All of the provided choices in the question	26	57%
Total	(N)	46	100
10101		.0	%

Question 11 was used to identify if there were any documentation procedures in place. The question asked if the respondents depended on documented procedures to perform their jobs. Ninety-six respondents answered this question. Results for this question clustered toward the middle of the scale, with 7.29% of respondents strongly agreeing, 36.46% of respondents saying they agree, 23.96% of respondents being neutral, 26.04% of respondents disagreeing and 6.25% of respondents strongly disagreeing (see Table 14). Only a few respondents expressed a strong level of either agreement or disagreement, and the proportion who agreed with the statement was marginally higher than those who disagreed. Therefore, the results indicate that knowledge curation is not perceived as particularly important among the presses' employees

<i>Table 14: (Q11)</i>	I depend on document	ed procedures to	do my work when	n I have questions in
fulfilling my job	tasks.			

Category	Frequency	%
Strongly agree	7	7.29
Agree	35	36.46
Neither agree nor disagree	23	23.96
Disagree	25	26.04
Strongly disagree	6	6.25
Total (N)	96	100

In the Technology focus section, 91 respondents answered survey Question 22, which asked about whether documenting work knowledge was a required part of their current work practices. Only about one-third of respondents said it was "a must" to document their work knowledge in their organizations, while 67.03% of respondents said their organizations did not require this documentation. (See Table 15).

*Table 15: (Q22) In your organization, documenting of work knowledge is a required part of your work practices* 

Category	Frequency	%
Yes, it is a must	30	32.97
No, it is not	61	67.03
Total (N)	91	100

Question 23 explored the level of availability of technology mechanisms and tools to share knowledge, and 91 respondents answered this question. The majority of respondents answered in the negative, with 58.24% saying they did not think these mechanisms and tools to share work knowledge were widely available in their organizations. Only 41.76% of them indicated that there are these kinds of aids for sharing work knowledge in their organizations (See Table 16).

*Table 16: (Q23) Mechanisms and tools to share work knowledge are widely available in my organization.* 

Category	Frequency	%
Yes, it is widely available	38	41.76
No, it is not	53	58.24
Total (N)	91	100

Question 24 asked about respondents' use of specific tools and methods for knowledge sharing or curation. There were 87 respondents who answered this question, which was used to examine if the press staff rely more on in-person interaction or human-machine interaction for knowledge transfer. The largest proportion of respondents (42.53%) chose "Team Meetings," but 32.18% of the respondents chose to write in their own answer (See Table 17\_a).

Table 17\_a: (Q24) Do you use any of the following tools or methods to share or store knowledge, experiences or best practices within your company?

Category	Frequency	%
Colleague(s) (mentor/buddy/mentee)	9	10.34
Team meetings	37	42.53
Collaborative platform such as Google Doc, OneDrive, Drop Box	13	14.94
Social Media such as Facebook, Twitter	0	0
Write-In	28	32.18
Total (N)	87	100

A coding scheme was developed in order to analyze the write-in narratives for Question 24. The following table shows the coding scheme used and the counts for each code. Among the twenty-eight write-in narratives, the most frequently-occurring response was given by 32% of the respondents, who indicated that they use "All of the provided choices in the question." Two respondents indicated they used none of the provided choices. Across the entire collection of write-in narratives, the word "meetings" was used repeatedly. These results reflect that university press employees rely heavily on in-person interactions in meetings to share knowledge and best practices.

Code	Response Type	# Count	%
1	Co-workers, Meetings and Collaborative Platform	7	25%
2	Database, meetings, documents	2	7%
3	Shared Drives/Folders	2	7%
4	Co-workers and Meetings only	1	4%
5	Meetings and colleague relationships within the press	1	4%
6	We have a departmental Wiki, but very few people are comfortable using it.	2	7%
7	Co-workers, Meetings and online project management	1	4%
8	I write up comprehensive instructions for all the tasks I do, both for my own reference and as a record for other employees.	1	4%
9	All of the provided choices in the question	9	32%
10	None of the provided choices in the question	2	7%
	Total (N)	28	100%

Table 17\_b: (Q24) Write-In Answers

## Research Question 3: Types of Knowledge Barriers that University Presses Encountered

Research Question 3 is an ancillary question branching out from Research Question 1 and 2, to identify what context barriers or technology barriers that work against sound knowledge practices may exist in university presses. Survey Question 10 asked the respondents about the issues they experienced when they were unable to share or receive knowledge from their team members. Ninety-four respondents answered this question. The most frequently-chosen response was "Time limitation" (47.87%), followed by 11.7% of respondents who said that there was no support to share knowledge. The fewest respondents chose "No motivation to share" (3.19%). Some (8.51%) noted that the right tools were unavailable. As with the earlier measures, 28.72% of respondents chose to write in their own response (See Table 18\_a).

Table 18\_a: (Q10) What are the prohibitions in instances that you were unable to share o receive knowledge and/or best practices from your team members in your job?

Category	Frequency	%
Time limitation	45	47.87
No support to share	11	11.7
No motivation to share	3	3.19
No right tools available	8	8.51
Write-In	27	28.72
Total (N)	94	100

A coding scheme was developed in order to analyze the write-in responses for survey Question 10. The following table shows the coding scheme used and the counts for each code. Among the twenty-seven write-in narratives, the most frequently-given response (30%) was that there were "no prohibitions" that prevented knowledge sharing or transfer. Other respondents who wrote in provided various issues that were prohibiting knowledge flows related to their tasks. Two respondents did not understand the question, and only one respondent mentioned an issue with the culture of knowledge transfer. Other interesting findings were that the next two most frequently-occurring responses were that "coworkers are unavailable" and that there is a "lack of this kind of training or knowledge." If the two counts for the related codes "lack of this kind of training or knowledge." If the two how to do my job as I am the only person [who] knows" were tallied together, it would be the second-highest ranked result for this measure.

Code	Response Type	# Count	%
1	No prohibitions	8	30%
2	Lack of this kind of training or knowledge	3	11%
3	Too busy	2	7%
4	Coworkers are unavailable	4	15%
5	Depends on situations	1	4%
6	Do not know	2	7%
7	Difficult to access remotely	1	4%
8	No culture of knowledge transfer across silos	1	4%
9	All of the above choices	1	4%
10	Very rare occurrence	1	4%
11	Difficult coworkers who hoard knowledge in order to stay in power	1	4%
12	They don't really know how to do my job as I am the only person knows	2	7%
Total (N)		27	100%

Table 18 b	b: (Q10)	Write-In Answers
------------	----------	------------------

Question 19 asked about the respondents' opinions on what the challenges are, in relation to sharing knowledge with coworkers in other departments. Sixty-eight respondents answered this question. The largest proportion of respondents (33.82%) considered the challenge to be that coworkers "did not perceive there is an urgent need" for the knowledge. More than a fifth of respondents (22.06%) noted that the challenge lies in the lack of an open-minded knowledge-sharing environment, and another 17.65% considered the challenge to be that their coworkers "did not perceive for the challenge lies in the lack of an open-minded knowledge-sharing environment, and another 17.65% considered the challenge to be that their coworkers "did not know there is a knowledge need" (See Table 19).

*Table 19: (Q19) From your view, what are the challenges of sharing knowledge with people from other sections of your company* 

Category	Frequency	%
Don't perceive there is an urgent need	23	33.82
Lack of open-minded sharing environment	15	22.06
Lack of trust of others' knowledge	8	11.76
No proper technological platform to share	10	14.71
Do not know there is a knowledge need	12	17.65
Total (N)	68	100

Question 25 queried the respondents about the biggest barrier for their storing the information they receive. Of the 81 respondents who answered this question, the majority (58.02%) chose to say that they were too busy or lacked the time, followed by the issue of dealing with a "poor information system/process" (27%). Only one respondent said, "I do not consider it is important" (See Table 20).

Category	Frequency	%
Too busy, lack of time	47	58.02
Poor technology tools	6	7.41
Organization policy	5	6.17
Poor information system/process	22	27
I do not consider it is important	1	1.23
Total (N)	81	100

*Table 20: (Q25) The biggest barrier for you to store information you received more efficiently and effectively is* 

The last question (Q26) in the survey was an open-ended question for the respondents to use to provide any feedback on any topics that the survey did not cover, related to their knowledge practices. Sixty-eight of the respondents answered this final survey question. A coding scheme was developed in order to analyze the narratives. The code "Miscellaneous" was used to code those narratives that were not related to the topic of this research study, such as "better management." Table 21 shows the coding scheme used and the frequency counts for each code. Among the 68 write-in narratives, the most frequently-mentioned theme was the need for better technology systems, tools and training (28%). The runner-up at 15% was having more knowledge sharing across departments. These results reflect a desire among university press staff for better technology implementation and more cross-departmental sharing of knowledge.

Code	Response Type	# Count	%
1	Better technology system, tools, training	19	28%
2	More structured mechanism on documentation or meetings	3	4%
3	More knowledge sharing across departments	10	15%
4	We are satisfied with the current situation	4	6%
5	Project meetings across the presses that affect the organizations on a micro-level	1	1%
6	Structured mentoring with open-minded management; upper management respect the knowledge from operational employees; Management buy-in and enthusiasm for knowledge sharing.	4	6%
7	A central hub that can be accessed by staff to avoid duplicated effort in documentation and keeping them in silos.	5	7%
8	Better communication channels between departments	4	6%
9	Better documentation for keeping leaving employees' knowledge inside the organization to facilitate new employees' training	5	7%
10	Too busy, want more time to share	2	3%
11	More trust between staff, better institutional knowledge sharing practices and recording, more incentives for knowledge- sharing and apprenticeship	2	3%
12	Miscellaneous (not directly related to knowledge sharing issues)	9	13%
Total (N)		68	100%

Table 21: (Q26) From a knowledge sharing point of view, what changes would you like to see in your company? Write-In Answers

### **Qualitative Research Results**

## About the Interviews

An invitation to participate in the in-depth interview portion of the study was sent out via the AUP's Director listsery, after the study's approval by the Institutional Review Board was obtained. This listsery reached a subscriber list of 147 AUP member presses' directors. After the AUP sent out the invitation via the Director listsery, six directors from the United States replied individually to the researcher, volunteering to participate in the first round of interviews. After those six interviews were conducted, transcribed and analyzed, patterns emerged indicating a saturation point had been reached. The researcher decided to conduct a second round of two more interviews, to confirm her findings before ending the qualitative data collection process. The researcher individually contacted five directors in different countries to see if they were interested in volunteering for an interview. Only two directors, both from the United States, agreed to participate in a second-round interview. In total, all qualitative research participants (five males and three females) were from the United States. Two participants were interviewed via the Zoom online meeting platform. Four participants were interviewed via phone and two others were interviewed on-site face to face. The rate of response for this portion of the study was 5.4%, with an N of 147, which is the number of directors subscribed to the AUP's Director listserv.

#### **Interviewees' Privacy and Identity Protection**

Each interviewee signed an informed consent form that stated the purpose and the details of participating in this study, prior to sitting for the interview. By signing the document, they agreed to participate and acknowledged that parts of the study and fragments of their interviews could be used in any research documents resulting from this study. Each of the interviewees was offered the chance to receive his/her interview transcript to review. Interviewees were addressed only by

the first letter of their names, such as Mr. S. or Ms. S., in the interview recordings. Each interview recording and the corresponding transcript was identified with a sequential number, such as DISS Interview\_1, as the key to organize the recordings and transcriptions in an orderly and confidential manner. The interviews were transcribed by a paid transcriber who signed a non-disclosure agreement. The researcher was the only person to have exclusive access to the consent forms, recordings and all digital and paper copies of the interviews.

Any names or organization names mentioned by the interviewees were redacted from the transcripts before they were analyzed. All recordings related to this research were destroyed as soon as the transcription of each recording was completed, as per the Institutional Review Board's rules.

### **Coding and Analysis using QDA Miner**

In social sciences research studies, the application of computer-assisted qualitative data analysis (CAQDAS) is popular (Atherton & Elsmore, 2007; Mangabeira, Lee, & Fielding, 2004). Qualitative data analysis software (QDAS) helps researchers with the labor-intensive process of qualitative data analysis (Silver & Lewins, 2014). For this study, the qualitative data analysis software, QDA Miner 5.0, provided by the University of Tennessee, was used for coding the textual data. Like other qualitative software for content analysis, it offers the ability to code, to retrieve text, and to conduct frequency and statistical analyses of text content, using its advanced features.

QDA Miner was designed with options that specifically address mixed methods analytics support (Silver & Lewins, 2014, p. 72). This application allows the creation of coding categories for use in devising and retrieving text-based coding structures when working with a small sample size. For this project, all of the interview sessions were audio-recorded using a digital recording

phone app and were then transcribed by the paid transcriber. The researcher listened to the recordings and read the transcriptions several times to gain more insight, and a better understanding of what the interviewees were trying to convey, both in broader terms and within the context of the study. This technique allowed the transcripts to be thoroughly evaluated to ensure that the information they contained was interpreted accurately (Patton, 2002). The transcriptions were then uploaded to the platform of the QDA Miner software for the data analysis process.

Content analysis was done through the use of the QDA Miner program. The qualitative aspect of this study, involving eight interviews, was manageable enough to permit a thematic analysis. To mark the coded segments, this program also provided an entire spectrum of colors to use for coding data and for adding reflective memos. The coding process of "putting tags, names or labels against pieces of data" (Huberman & Miles, 2002) organized the related segments of the data into categories. The researcher used colors that were distinct enough to reveal as explicitly as possible when different categories occurred in the same segment.



Figure 5 - An Example of a Color-Coded Segment
By reading the transcripts, the researcher identified words and phrases that related to relational/interpersonal factors, and also those that indicated content or technical factors. By comparing the interview transcripts, relationships or patterns of similarities were grouped together. Those identified groups were then organized into the overall outline of categories, in accordance with the best possible formulation to be chosen, while any redundant themes were eliminated (McCracken, 1988).

In this manner, the transcript data were reduced from a large amount into a well-organized and coherent outline that summarized the interviews. When labelling the categories, the exact words from the transcripts were used whenever it was possible. The researcher only used her own words when it was necessary to name and order categories in grouping similar ideas.

### **Developing a Theme Codebook**

The data analysis process involved creating a coding classification, as "developing some manageable classification or coding scheme" is essential to analysis (Patton, 2002, p. 463). The codes are "words or phrases that represent themes" (Zikmund, Babin, Carr & Griffin, 2008, p. 468). Two coding strategies (open coding and axial coding) were used in this study to interpret data and relate concepts to each other (Corbin & Strauss, 2008). The process involved taking the data apart, rearranging and putting it back together in a manner that represented the concepts and categories as they related to each other in meaningful ways (Corbin & Strauss, 2008). The objective of the analysis was to build a structural synthesis of core elements that inform the respondent's view in general and the study's topic in particular (McCracken, 1988).

The data analysis began the open coding with a brainstorming approach to analyze all of the potential code names. Reading line by line from the interview transcripts, the researcher created the preliminary codes by treating each utterance as an entity of its own (McCracken, 1988) and

extracting concepts after considering their possible meanings and examining the context carefully (Corbin & Strauss, 2008). Then the researcher assigned codes to the concepts and grouped the codes into categories. She then arranged these categories into a cohesive research narrative. Some of the codes were assigned by the researcher during the raw data analysis, while some others were provided by the participants as "emic terms" (Patton, 1990) or "in-vivo" code (Corbin & Strauss, 2008), as mentioned in scholarly literature.

After completing the open coding stage from reading the interview transcripts, the researcher observed an emerging pattern related to the knowledge management practices shared by the participants. The axial coding phase was then conducted to identify potential themes to be used to highlight segments from the text. The concepts were identified in the data and assigned codes that were further grouped into categories or themes according to shared properties (Corbin & Strauss, 2008). Then Dey's (1993) "splitting and splicing" technique was used to group data into categories consistent with the revealed patterns. As per Dey's technique, splitting is to refine categories by subcategorizing data and splicing is done to combine categories to form an integrated conceptualization. Therefore, the categories in this research project were expanded into levels by dividing them into various smaller categories or by combining them into broader categories, as the coding process went on to reveal emerging themes from the data. An example of the process is shown in the following diagram.



Figure 6 - The Development of Theme Code Levels

The findings of this research were presented with verbatim quotes from the participants, arising from the interview transcripts, with the researcher's interpretation. This was to balance the two objectives of reflecting the participant's voice, while achieving an appropriate level of description and analysis (Patton, 1990). This is important because the goal of qualitative research, rather than producing findings that are generalizable to a large population, is to understand a phenomenon within its context and to provide a description of the findings for informing anyone's future attempt to make similar judgements (Lincoln & Guba, 1985).

Interview data were analyzed on the basis of a systematic coding (breaking down) of data according to a code list (or code system) to identify relevant patterns (Saldaña, 2012). The coded segments were then grouped and synthesized into (more general) categories, which in turn linked to more general themes. The code system (the categories and themes) was developed gradually and collaboratively on the basis of the coding process to summarize and describe the data in a useful manner (Flanagan, 1954).

There are two main types of codes: (1) descriptive codes and (2) inferential (or pattern, or thematic) codes (Miles & Huberman, 1994). Early labels may be descriptive codes requiring little

or no inference beyond the piece of data itself. These descriptive codes are valuable in getting the analysis started and in enabling the researcher to get a "feel" for the data. First level coding mainly uses descriptive, low inference codes which are very useful in summarizing segments of data and which provide a basis for later higher-order coding (Huberman & Miles, 2002).

In this research, the descriptive coding covers the first section of the interview template in which interviewees were presented wth such prompts as: Tell me about your organization. What is your job? What is your publishing focus? For example, the following shows an excerpt from the coding of the transcripts.

CASE DISS interview\_1 My organization is considered a mid-size university press

The later-level coding focused on pattern coding. A pattern code is more inferential and a sort of "meta-code" (Huberman & Miles, 2002). Pattern codes pull together material into smaller, more meaningful units. There is usually a range of possibilities when it comes to applying codes to the data or finding them within the data. The patterns "collegiality" and "do anything for me if I asked" were drawn from the response data. For example, the following shows the excerpts from the coding of the transcripts. The excerpts show how differently interviewees responded to the same question.

CASE DISS interview_2	We are a very, I think, egalitarian and collegial organization
CASE DISS interview_7	I know that anybody here would do anything for me if I asked them to, and I think that they feel that way about one another, too. There is not a single position where somebody is really off by themselves.

This approach allows for the deduction of common themes and patterns from the data. The final coding scheme (see Appendix H) was created and used as a template to interpret across the eight interview transcripts.

# Results

# Interviewees' Profiles

# Table 22: A Summary of Interviewees' Profiles in the Qualitative Research

		Interviewed		Org.	Located	
Interview	Gender	via	Country	Size	in *	Main Publishing Focus
#1	F	Online Zoom	USA	Small	East South Central	Military history, film studies, Appalachian studies, African- American studies
#2	М	Online Zoom	USA	Small	East North Central	Humanities and literary studies, comic studies, Victorian studies, medieval studies, gender and sexuality studies, regional and linguistics
#3	F	Face to Face	USA	Large	South Atlantic	Humanities, life sciences, health policy, public health and regional titles
#4	М	Phone	USA	Small	Mid Atlantic	Humanities, social sciences with emphasis on the fields of African-American studies, American studies, communication and media studies, religion and urban studies
#5	М	Phone	USA	Large	Mid Atlantic	Humanities, social sciences, art criticism and history, visual studies, gender studies, gay and lesbian studies and American studies
#6	F	Phone	USA	Small	East North Central	Biographies for Young Readers, Cambridge Center of African Studies Series, The Civil War in the Great Interior, Series in Appalachian Studies
#7	М	Face to Face	USA	Small	East South Central	American studies, Appalachian studies, African-American studies, folklore, historical archaeology and literature
#8	М	Phone	USA	Medium	East North Central	Political science, performing arts, classics, American studies and Great Lakes region studies

# Organization Size Classification: Large (= > 50 staff), Medium (25 - 49 staff) & Small (< 25 staff)

\* Based on the Census Bureau-designated regions and divisions.

# **Consistent Themes Observed**

Several patterns were observed across the interview transcripts. They are illustrated as follows, with an excerpt of a code segment for each one:

Theme 1 – Unified Social Norm within the Presses

The social norm across the university presses is that of a close-knit society. Each press operates in a similar way in their publishing procedures. The only competition among them is to recruit notable authors. Most presses host internships or similar programs. As it was portrayed in the transcripts, the presses expect that the newly-hired employees should already have had a certain degree of experience and knowledge working in the field, either through an internship program or from previous employment. For example:

DISS Interview\_2 I think, in the university press world, where you don't hire somebody who doesn't have some experience in scholarly communications.

Theme 2 – Collegiality and Open Culture Across the Presses

A high level of collegial spirit at each press was found across the eight interviews. The interviewees indicated they perceived that their employees trust each other. The directors giving these interviews reflected that their staff are eager to share within their press internally or via interactions with other member presses. These presses were empowered by management to share internally and collaborate with other departments within the host university. In fact, three out of the eight interviewees mentioned the actual word "collegial" in their interviews. For example:

DISS Interview_1	One interesting thing about university presses, we're very collegial, and so a lot of times we'll ask each other for advice
DISS Interview_2	We are a very, I think, egalitarian and collegial organization
DISS Interview_5	I would say the university press community, in general, is amazingly collegial.

Theme 3 – Meetings and Conferences are the Main Way to Share Tacit Knowledge in the Presses

Using the keyword retrieval feature in QDA Miner software on the eight interview transcripts, the word "meetings" was counted thirty-four times and the word "conferences" appeared six times. These high counts provided evidence that attending meetings and conferences was a popular way for university presses to share tacit knowledge.

🙋 Keyw	ord Retrieval - 34	Hits			
Criterio	Search Hits				
CODE:				<i>1</i> 9 + 1	💬 🏦 🕬 🖓
Case #	Case	Variable	Sentence	Nb hits	Text
2	DISS interview_1	DOCUMENT	287	2	Interviewee: Well, one of the other things that's a less technically-oriented kind of thing but I do think it's knowledge
2	DISS interview_1	DOCUMENT	288	2	One of the MEETINGS we've started doing happens every two weeks, and all the acquiring editors are present, and also
3	DISS interview_3	DOCUMENT	352	2	One of them is that we go to professional MEETINGS.
3	DISS interview_3	DOCUMENT	354	2	The SSP meeting, for example, is one of those MEETINGS that we go to that is mostly science or mostly humanities, but
3	DISS interview_3	DOCUMENT	363	2	Understanding what library needs are is incredibly important, and so we're going to all the library MEETINGS, as well.
3	DISS interview_3	DOCUMENT	394	4	Do you guys have any MEETINGS or any mechanism in place to share knowledge, either MEETINGS, or lots of
3	DISS interview_3	DOCUMENT	473	2	Interviewee: We have MEETINGS.
3	DISS interview_3	DOCUMENT	479	2	Interviewee: We have MEETINGS.
3	DISS interview_3	DOCUMENT	485	2	Some of that is going to be kind of celebratory MEETINGS, where we'll celebrate the fact that MUSE just launched their
3	DISS interview_3	DOCUMENT	500	2	Interviewee: At my last job, what we would do is because it was smaller, there were only 50 people, and here we have
4	DISS interview_4	DOCUMENT	561	2	We will seek out what are some of the trends and hot topics, and we'll set up MEETINGS with scholars and discuss their
4	DISS interview_4	DOCUMENT	570	2	I mean a lot of our projects are discussed internally, so we have the editorial projects MEETINGS, where we will review a
4	DISS interview_4	DOCUMENT	601	2	We have these seasonal planning MEETINGS where we determine what books are in the queue, what's coming down the
4	DISS interview_4	DOCUMENT	602	2	We kind of have these seasonal planning MEETINGS so we know in advance what's coming down the pike.
4	DISS interview_4	DOCUMENT	607	4	We have these weekly MEETINGS, these planning MEETINGS so we know what's coming down the pike, and how they're
4	DISS interview_4	DOCUMENT	673	2	I think it's stepping back and taking the time to do it, or setting up these MEETINGS or opportunities to share the
4	DISS interview_4	DOCUMENT	674	2	Just recently, we were all operating in our own silos and evaluating proposals and manuscripts, and I thought, "Why not
5	DISS interview_5	DOCUMENT	789	2	Interviewee: There were lots of MEETINGS, but most of it is in writing.
6	DISS interview_6	DOCUMENT	948	2	We have weekly staff MEETINGS.
6	DISS interview_6	DOCUMENT	984	2	In terms of interpersonal communication, as I mentioned earlier, we rely on I think pretty regular MEETINGS.
6	DISS interview_6	DOCUMENT	987	4	Most people are heads of, you know, a lot of single-person departments, and so we moved to these staff MEETINGS
6	DISS interview_6	DOCUMENT	990	2	We have other marketing MEETINGS once a week, and so on.
6	DISS interview 6	DOCUMENT	1001	2	What I did to fill that sort of need was that every week at our staff MEETINGS, we have a quick go-around where
6	DISS interview_6	DOCUMENT	1018	2	For example, each season, when we launch the books for either the spring or the fall season, there is all kinds of
6	DISS interview_6	DOCUMENT	1040	2	I would say those weekly MEETINGS and going through all of the information, you know, sort of saved us, if there was a
7	DISS interview 8	DOCUMENT	1131	2	I usually go to the Society for Scholarly Publishing, and I will every couple of years go to international MEETINGS, such as
7	DISS interview _8	DOCUMENT	1150	2	University of XXX is a Google Suite user, and so quite a lot of people will use Google tools, such as Google chat, and we use
7	DISS interview _8	DOCUMENT	1151	2	We have some MEETINGS, such as the operations group, that bring together people from different departments around
7	DISS interview 8	DOCUMENT	1230	2	They will have MEETINGS set up with people in other departments to get a sense of what those people do.
7	DISS interview _8	DOCUMENT	1232	2	There is a passport program that's what we call it but it's essentially just to make sure that people who start have

Figure 7 - An Example of Content Analysis Count #1

🙋 Кеум	vord Retrieval - 6 H	its				
Criterio	n Search Hits					
CODE: 🔽 🖉 A 🖉 + 🖵 🏦 👺 🕅						
Case #	Case	Variable	Sentence	Nb hits	Text	
4	DISS interview_4	DOCUMENT	560	1	As a director, I also acquire, and we go to the key academic CONFERENCES throughout the year, and we will go and look	
4	DISS interview_4	DOCUMENT	562	1	Some of it is revised dissertations, or it could be their second or third book, so a lot of it is going to these key academic	
7	DISS interview _8	DOCUMENT	1126	1	Interviewee: I go to CONFERENCES.	
7	DISS interview _8	DOCUMENT	1127	1	I go to the Charleston CONFERENCE.	
8	DISS interview_7	DOCUMENT	1347	1	Interviewer: When you say to collaborate on this big book, so the author, did you have a CONFERENCE with them in a	
8	DISS interview_7	DOCUMENT	1357	1	Interviewer: When you say the interaction, besides CONFERENCE, what else?	

Figure 8 - An Example of Content Analysis Count #2

There was also evidence in the transcripts that indicated that university presses preferred the personal touch, such as drop-in meetings. For example:

DISS Interview\_5 People also get introduced to various parts of the press through something we call Meet the Press, which is every month, pretty much. One group does a presentation, and then all the new employees within the last year, and any longer-term employee who wants to can come and hear about what's going on in that part of the press.

It is interesting that a female director considered that proximity was a barrier to sharing knowledge because her staff were located in two different buildings, while a male director indicated that placing staff in an open-office layout was a positive measure to facilitate knowledge sharing. Although they saw "office proximity" in opposite ways, the underline thoughts were the same.

**Theme 4** – The Association of University Presses is important for member presses to seek and share operational knowledge.

The two listservs of the Association of University Presses and the AUP's annual

conference in San Francisco were considered to be the main venue for the presses to exchange

information and create knowledge. For example:

DISS Interview_1	A very active e-mail listserv with our professional association
DISS Interview_3	We're looking outside to the Association of University Presses, where we're determining what our peers are doing.
DISS Interview_4	We rely on our society, our association to kind of give us the tools and background to help us train new employees.
DISS Interview_7	I'm a strong advocate of the AUP, which is the Association of University Presses. It's an umbrella group provides advice about legal matters and about day-to-day operations, and keeps us updated on legislation that might be coming, or even things as minor as announcing new projects, and things like that.

Theme 5 – Knowledge Management Theory and Concepts are Unfamiliar Among Presses

It caught the attention of the researcher during the course of the interviews that the interviewees were unfamiliar with knowledge management concepts. Several directors asked about the meaning of "knowledge management" during the interviews. Some directors did not ask about the meaning of knowledge management, but instead answered the questions from marketing strategies or human management perspectives. A minor experiment was done in the last two interviews to check if there would be a difference in answers. The researcher provided the interviewees with the definitions of explicit knowledge and implicit knowledge prior to the recordings of these two interviews. However, there was no difference in the outcomes, compared to the previous six interviews. Interview #7 answered the questions from the context of their book publishing activities and Interview #8 answered the questions from an office communication perspective. For example:

DISS Interview_3	It's something that I need to learn more about. I've certainly heard about this and read about this discipline and what you're studying, but I think that there is probably a lot more that we need to know as an organization moving forward. It could end up being part of this culture assessment, the culture visioning, the culture strategic planning that we might move forward with.
DISS Interview_5	Tell me what you mean by sharing knowledge. Do you mean within the organization, between our organization and other organizations?
DISS Interview_6	When you are speaking about shared knowledge, Judi, are you sort of speaking about the issue of sort of internal communications and systems?
DISS Interview_7	We use SharePoint for our editorial board. When we gather reports, we share documents that way, and things like minutes for the meeting and all that kind of stuff
DISS Interview_8	We also have written documentation about various processes Our internal processes are captured in documents and those are Google documents

Theme 6 – Knowledge Curation is Mainly Documented in Conventional Ways, Such as Being

Stored in Internal Databases and Online Cloud Storage Media

There is no sophisticated knowledge management system mentioned in any of the eight

interviews. For example:

DISS Interview_1	We have a press-wide database that the press has been using for a number of years.
DISS Interview_2	Have a shared server
DISS Interview_7	We use a weekly wiki for staff meetings to post agendas and have people add to it
DISS Interview_7	We use SharePoint for our editorial board. When we gather reports, we share documents that way, and things like minutes for the meeting and all that kind of stuff
DISS Interview_8	We also have written documentation about various processes, and a version of the guidelines that we share with our authors, which is a little bit more detailed for internal staff. Our internal processes are captured in documents that we call standard operating procedures, SOPs, and those are Google documents that are always called SOP in their title so they're easy to find.

Intervie w	The Classified Organization Size in This Research	Theme #1	Them e #2	Them e #3	Them e #4	Them e #5	Theme #6	Remarks
# 1	Small	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$		
# 2	Small	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$		
# 3	Large	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$		
#4	Small	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$		
# 5	Large	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$		
# 6	Small	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$		$\checkmark$	
<		Satura	tion Poin	ts Emerg	ed		>	
#7	Small	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	Mini- experimen t applied
# 8	Medium	$\checkmark$						Mini- experimen t applied

 Table 23: Emerging Themes by Each Interview

Legend

Organization Size Classification by Number of Employees: Small size (< 25 staff), Medium size (25 to 49 staff), Large size (= > 50 staff)

Theme Description:

Theme #1 Trust is the social norm in this close-knit society

Theme #2 Full of collegiality in their organization culture

Theme #3 In-person communication is important

Theme #4 The Association of University Presses is important for member presses

Theme#5 Knowledge management theory and concepts are unfamiliar to presses

Theme#6 No sophisticated knowledge management system was mentioned

# **Identified Outliers**

As reflected in the interview transcripts, these eight university presses demonstrated a similarity in their practices in knowledge sharing and retention. Amid the qualitative data analyses, three incidents from three of the interviews were identified as exceptions to Theme 1 (a unified social norm). These university presses performed slightly differently, in such a way that the researcher considered them to be outliers. These university presses' activities were:

- DISS Interview\_1 In order to look at this reorganization, we brought in a trainer from our own university who was offered to us... she is kind of guiding us through what are the steps of our current procedures, and then we put all of those down and who is responsible for what and in what order.
- DISS Interview\_2 We were approached by a faculty member about publishing actual comics. ... We were already used to using comics in our scholarship ... I'm a comic lover myself ... so I think that might be an example where my personal experience was useful in ... the production department deal with particular challenges that are specific to publishing comics.

DISS Interview\_3 We also just finished a major organizational culture assessment, and so that was done with quantitative and qualitative work.

All eight interviewees, as a whole, described the same kind of operational activities in scholarly publishing among the university presses. There was obviously a unified social norm pattern that emerged across the university presses, except for the three interviewees who mentioned the three incidents that made them veer slightly away from the social norm. The university press of Interview\_1 brought in a trainer from their own university to offer guidance on their current procedures. The university press of Interview\_2 accepted their faculty member's idea of publishing outside of the press's comfort zone (comics) and the university press of Interview\_3 conducted a major organizational culture assessment. These incidents represented the initiative of university presses in knowledge sharing with outsiders to innovate their usual processes.

# Research Question 1: Factors Influencing Individuals' Knowledge-seeking and

## Knowledge-sharing in University Presses

This research tried to understand knowledge practices among university presses, with specific foci on behavioral, cultural, social and technological issues. Research in the scholarly literature indicated that having a sharing culture and the presence of trust are two key factors influencing knowledge seeking and sharing behavior. The findings from this research provide evidence from the qualitative data supporting this school of thought.

While conducting a content analysis on the transcripts, the word "meeting" was counted thirty-four times and the word "conference" was counted six times, across the eight interview transcripts. The following coded segments support the importance of attending meetings and conferences in their knowledge-sharing processes.

DISS_Interview #4	We go to the key academic conferences throughout the year
DISS_Interview #5	We also have brown-bag lunches on a fairly regular basis at which people present things they know and want other people to know, or give people a chance to know

In fact, university presses are also collegial in their culture. Several directors noted in their interviews that this is characteristic of their organizational culture. For example, they said:

DISS_Interview #1	One interesting thing about university presses, we're very collegial, and so a lot of times we'll ask each other for advice
DISS_Interview #2	We are a very, I think, egalitarian and collegial organization
DISS_Interview #5	I would say the university press community, in general, is amazingly collegial
DISS_Interview #7	I know that anybody here would do anything for me if I asked them to, and I think that they feel that way about one another, too. There is not a single position where somebody is really off by themselves.

Collegial organizational culture works hand in hand with a trustful social norm. With a collegial organizational culture, people are eager to share knowledge. With a trustful social norm, people trust their coworkers and their coworkers' knowledge. Therefore, "collegiality" and "trust in the relationship" as independent variables correlated with the degree of "knowledge sharing" observed as a dependent variable.

For explicit knowledge, university presses tend to use similar literature sources in their specific professional areas, such as Publishers Weekly and Scholarly Kitchen. The Association of University Presses (AUP) email listserv served as the major knowledge-sharing platform in a group setting for individuals to interact and mingle with their fellows in the field, to get news and updates about their profession. The important role of AUP as an information source in the university presses' working life is reflected in the following interview transcripts from the qualitative research.

DISS Interview_3	We're looking outside to the Association of University Presses, where we're determining what our peers are doing.
DISS Interview_4	We rely on our society, our association to kind of give us the tools and background to help us train new employees.
DISS Interview_7	I'm a strong advocate of the AUP, which is the Association of University Presses. It's an umbrella group provides advice about legal matters and about day-to-day operations, and keeps us updated on legislation that might be coming, or even things as minor as announcing new projects, and things like that.

### Research Question 2: Ways that University Presses Apply and Retain their Operational

# Knowledge in their Organizations

Capturing knowledge in organizations for future use is important in any organization's strategic plans. One of the research questions that guided this project sought to find out if the presses have any knowledge-retention strategies in place. The findings support the scholarly

literature's assertion that organizations consider knowledge to be crucial for long-term organizational success, but they do not formulate any knowledge-retention strategies (Liebowitz, 2008).

The findings from this study's qualitative research portion reflected that the concepts of knowledge management were unpopular across university presses. There were no formal structures or any sophisticated knowledge-retention platforms to organize or classify operational knowledge mentioned in qualitative data.

To apply and retain operational knowledge, university press members rely mainly on operational manual paper documents, listserv communications, email, their organization's shared server, employees' personal computer drives or cloud computing collaboration tools such as wikis, SharePoint and Google Suite.

DISS Interview_2	Have a shared server
DISS Interview_7	We use a weekly wiki for staff meetings to post agendas and have people add to it
DISS Interview_7	We use SharePoint for our editorial board. When we gather reports, we share documents that way, and things like minutes for the meeting and all that kind of stuff
DISS Interview_8	We also have written documentation about various processes Our internal processes are captured in documents and those are Google documents

### Research Question 3: Types of Knowledge Barriers that University Presses Encountered

Several types of knowledge barriers identified by other scholars are listed in the literature review of this dissertation, such as technological barriers (McLaughlin, Paton & Macbeth, 2008). Other examples are having a weak culture of sharing, such as occurs in some Asian organizations that only share among family members and close colleagues (Yao, Kam, & Chan, 2007), or employees who "keep secret information to avoid losing their job" (Hermann, 2011). The literature also identified personal barriers such as those having no motivation to share, or those who hoard information to make themselves more competitive and to have an advantage over others (Hermann, 2011).

For this research on university presses, no evidence emerged in the findings of any of the above-mentioned obstacles (except for the technological barriers) that hinder knowledge sharing between team members on an intra-project level. However, the presence of the weak sharing culture barrier cannot be determined, because these interviewees were from a fairly-open, western-centric culture. There is no way to draw conclusions about these organizations' sharing culture by ethnicity from this research.

For the interviews with directors, issues with proximity were considered to be an obstacle, as two directors mentioned about their office layout. One director indicated that not having her staff on the same floor hindered knowledge sharing and another director credited his organization's office layout where his staff were nearby as facilitating knowledge sharing. Regardless of seeing it through a positive or negative lens, office proximity was considered to be a type of barrier.

- DISS Interview\_3 Our marketing department is split between floors here, and so I've been told that that's one of the reasons why the marketing department is not collaborating and sharing knowledge as they should be.
- DISS Interview\_8 They communicate in several different ways. One of them is walking around and talking to people, because we're all in the same building, and it's easy to go and talk to someone.

# Coded Segments in Emerging Themes by Organization Size

# 1) Unified Trust Social Norm

# Emerging Themes by Organization Size

(Unified	in	Trust	Social	Norm)
----------	----	-------	--------	-------

University		
Organization Size	Press	Coded Segment Examples
Large Organization Size	DISS	the visioning for the culture that we want to do a better job of encouraging people to
(50 staff or more than 50 staff)	interview_3	mix together, and building some kind of structure so that people have more of an opportunity to mix.
	DISS interview_5	I think there is a general sense that information needs to be shared across the press
Medium Organization Size (25 staff or less than 49 staff)	DISS interview_8	there are a variety of different systems and ways in which people are sort of introduced to their jobs. We have actually recently also introduced a mentoring system, and that's new for us. That's providing a mentor who is in a completely different department just to be a kind of informal guide to our culture, for example. We're also thinking about that person's role in the context of supporting diversity. For example, helping that person to navigate our organizational culture
Small Organization Size:(less than 25 staff)	DISS interview_1	have a vigorous internship program that we run both in partnership with our campus where we're located, but also we're a consortium press.
	DISS interview_1	I rely on a network of colleagues
	DISS interview_1	anywhere between three and seven or eight interns per semester
	DISS interview_2	A lot of university presses have strong internship programs.
	DISS	I think, in the university press world where you hire somebody who doesn't have
	DISS interview_2	We are really not prepared to bring on somebody who is completely new to the scholarly communication world.
	DISS interview_2	I think that the internship pipeline is definitely something that if we are looking for a new person and they have that internship line on their CV, that is one of the most important things to us if they don't have any actual experience in university press publishing
	DISS interview_2	I think, in the university press world where you hire somebody who doesn't have some experience in scholarly communications
	DISS interview_2	they're motivated by a shared mission, something that we all are working towards. I think that is probably the most important thing for having an organization that is sharing information openly and is creating informed colleagues. Really, that culture, I think, is key to that, and, frankly, I think it's key to our success, not just the knowledge-sharing portion of it but achieving our mission of maximum dissemination of our content. I really do think the culture is the primary element to being able to do all of those things.
	DISS interview_4	our own skill sets and knowledge that we impart on our new employees, and we meet with them on a regular basis, they get immerged in university publishing by doing, and we will train them and show them what's necessary.

# 2) Collegiality and Open Culture

#### Emerging Themes by Organization Size

### (Collegiality and Open Culture)

Organization Size	University Press	Coded Segment Examples
Large Organization Size (50 staff	DISS interview_3	I have empowered them to decide what they and their staff and their peers believe is the best way to share this
or more than 50 staff)		knowledge.
	DISS interview_3	when I first came in, I started a staff enrichment team. In the past, the collaborative work that's been done was coordinated mostly by one person. It was my assistant. The staff enrichment team has people who come from all the different divisions, who come together to determine what kinds of activities are best for enrichine our lives
	DISS interview_5	I would say the university press community, in general, is amazingly collegial.
Medium Organization Size (25 staff or less than 49 staff)	DISS interview_8	They can serve on committees in the library, there is an amity, a friendship committee that sort of greets new people more informally and offers opportunity for networking, et cetera
	DISS interview _8	we've moved responsibility for educating the editors to a single person, because previously it was becoming very confusing. It wasn't clear who to approach, so we've designated one person to be responsible for helping them to understand the systems, for receiving their feedback for refining the system. That's a situation where it's a problem that we've had around organizational communication, and it's an example of how we're addressing that problem,
Small Organization Size:(less than 25 staff)	DISS interview_1	One interesting thing about university presses, we're very collegial, and so a lot of times we'll ask each other for advice
	DISS interview_1	promotes trust between the department heads in that they know marketing is going to give them frank feedback, but also work with them and want to have a constructive conversation rather than when marketing first learns about a book when it's about to be transmitted and put in the catalog
	DISS interview_1	one of the other things that's a less technically-oriented kind of thing but I do think it's knowledge sharing and certainly impacts the overall success of the organization, is we're trying to think about making the meetings we have the most useful.
	DISS interview_2	They involved by a shared mission, something that we all are working towards. I think that is probably the most important thing for having an organization that is sharing information openly and is creating informed colleagues. Really, that culture, I think, is key to that, and, frankly, I think it's key to our success, not just the knowledge-sharing portion of it but achieving our mission of maximum dissemination of our content. I really do think the culture is the primary element to being able to do all of those things.
	DISS interview_2	culture is incredibly important, and creating the right culture at the press so that people aren't being motivated by fear
	DISS interview_2	I do encourage, particularly my department heads, to, as I said, be open to experiments, be open to failure.
	DISS interview_2	Each particular process being documented, you can certainly give them that document as a guide, but there is a lot of sort of hands-on training and discussion that happens with a new employee.
	DISS interview_2	We are a very, I think, egalitarian and collegial organization
	DISS interview_2	I encourage that, and I think it definitely benefits everybody. I do everything that I can to break the silos down between the individual departments so that there is always communication, and I think that it does work rather successfully here.
	DISS interview_2	I very much like to empower my colleagues to be able to make decisions themselves.
	DISS interview_4	Most people are eager to share their knowledge because it's something that should be shared
	DISS interview_4	Just having an open debate and a dialogue is important.
	DISS interview_4	open to other people's ideas, or for the other ideas
	DISS interview_4	openness and collaboration and creativity
	DISS interview_4	Everything we do has to be part of a collaboration
	DISS interview_6	As a director, you need to cultivate a culture of being able to speak out in a constructive fashion.
	DISS interview_6	I personally would say that a real do is to create an atmosphere and a culture where people feel perfectly comfortable about speaking out, and create a style of that kind of confirmation that is not confrontational, that is not blaming, that is just matter-of-fact.
	DISS interview_6	The director should be really proactive in using them and encouraging everybody, and making sure that all of the communications tools are respected and used.
	DISS interview_6	I would say from about eight years ago, we purposely and deliberately set towards making all of our information available to everyone no matter where they were, if they were remote or working at home or whatever
	DISS interview_6	We have an excellent work culture at this press. You couldn't hope for a more communicative and positive sort of attitude.
	DISS interview_7	I know that anybody here would do anything for me if I asked them to, and I think that they feel that way about one another, too. There is not a single position where somebody is really off by themselves.
	DISS interview_7	I kind of want there to be people offering solutions and collaborating with each other. Sometimes, I don't even need to be a part of that until we come to some kind of decision about it. Again, I think we try to emphasize fluidity and a certain flatness so that everybody has a chance to contribute to that exchange, particularly about knowledge itself.

# 3) Sharing Tacit Knowledge via Meetings and Conferences

Emerging Themes by Organization Size (Sharing Tactic Knowledge via Meetings and Confer Organization Size Large Organization Size (50 
 University Press
 Coded Segment Examples

 DISS interview\_3
 Understanding what library needs are is incredibly important, and so we're going to all the library meetings, as well.
 The SSP meeting, for example, is one of those meetings that we go to that is mostly science or mostly humanities, but science publishers tend to be ahead of humanities publishers and we need to know what's going on. We connect with people who are working in the sciences, journals as well as bodds, to keep up to dat. DISS interview\_3 One of them is that we go to professional meetings. Those are very, very good, of course, for networking but it's also good for going to panels and DISS interview\_3 LIAS INTERVEN\_3 Concerns and a give pro-DISS interview 5 We also have brown-bag lunches on a fairly regular basis at which people present things they know and want other people to know, or give people a chance to know DISS interview 5 People also get introduced to various parts of the press through something we call Meet the Press, which is every month, pretty much. One group does a presentation, and then all the new employees within the last year, and any longer-term employee who wants to can come and hear about what going on an that part of the press. There were lots of meetings, but most of it is in writing. It might be sent electronically. For me, I'm printing it out. For the younger people, they're probably just random it on the server. DISS interview 5 so our last planning round, which happened throughout the year of 2015, we convened I think sits groups, each led by one of our directors, to talk about areas in which we might develop plann for next steps. Almost everyone at the press participated in some way or when? We, on parpose, and pressow how was most directly connected to that part of the press. Yeah, we put taggether at of dates for where we needed to move forward, and then show all cause to me with reports from each of the directors who were leading the group. DISS interview 5 we have a very comparised endowed precess so that they get acquanited with all the parts of the press. and all the systems of the press DISS interview\_5 Every second Friday of the month, that morning is designated for what we call Not Doing Business As Usual, which can be anything. People can set up a program, they can set up a discussion. Medium Organization Size (25 staff or less than 49 staff) DISS interview it's essentially just to make sure that people who start have meetings across the organization to learn 8 DISS interview \_8 If it is a senior person, they will meet with other departments. It depends on the expectations of how regularly people will be working with other departments DISS interview They communicate in several differ \_8 easy to go and talk to someone. ent ways. One of them is walking around and talking to people, because we're all in the same building, and it's DISS interview Society for Scholarly Publishing, 8 DISS interview the Coalition for Networked Information DISS inter rview Charleston conference. DISS interview\_1 One of the meetings we've started doing happens every two weeks, and all the acquiring editors are present, and also our marketing director and our managing editor, and sometimes our fulfillment director, where we talk about potential book projects. Each of the acquiring editors will bring importon Knowkeege starting has been very valuable. Small Organization Size:(less than 25 staff) DISS interview\_2 just walk out of your office DISS interview\_2 We've only got about 13 folks working full time here, and that means it's relatively simple to just walk out of your office and go find somebody and ask them. I DISS interview 2 Each department, we meet on a weekly basis DISS interview 2 I also meet on a weekly basis with the heads of those depar DISS interview 4 we go to the key academic conferences throughout the year DISS interview 4 meet with different departments at the press DISS interview\_4 we'll set up meetings with scholars and discuss their works, and see if there is potential for them to want to publish that into a book DISS interview\_4 editorial projects meetings DISS interview\_4 seasonal planning meetings where we determine what books are in the queue, what's coming down the line, what are the deadlines for the manuscript submissions DISS interview\_4 these weekly meetings, these planning meetings DISS interview. 4. Think for septing back and laking he ince to 0 is or setting up bees needings or opportunities to abart the basedupts and transcentry, see were and upportunities and the state of DISS interview 6 the director, make time to speak with remote people really daily, if possible

DISS interview\_6 What I did to fill that sort of need was that every week at our staff meetings, we have a quick go-around where everybody doesn't speak about what they accomplished but they speak about what they have to accomplish in the week ahead.

 DISS interview, 6
 marketing meetings once a week, and so on

 DISS interview, 6
 Nots propels are heads of, you know, a bt of single-person departments, and so we moved to these staff meetings head market free of top popels. That really didn't create a good atmosphere of teamwork and cananaderia. It felt sort of exclusive in a way that was sort of unnecessary.

 DISS interview, 6
 We have weekly staff meetings.

 DISS interview, 6
 We communicate hy going to each other's offices.

 DISS interview, 6
 We communicate hy going to each other's offices.

 DISS interview, 6
 We communicate where there are lots of heads to of quartments.

 DISS interview, 7
 With the pople that have worked here, all but one of them has been here for five years or longer, and so they're experienced, and they can be trusted to the interview.

- DISS interview 7 intentioned, some production is fairly flat versures being hierarchical, DISS interview 7 intentioned, some stress being hierarchical, DISS interview 7 probably half a dozen planning meetings before they even got started on the writing of this.
- DISS interview\_7 they had several meetings early on just to try to talk about

DISS interview\_7 We have weekly staff meetings,

DISS interview\_7 meet three times a year

DISS interview\_7 After the planning stages, there was lots of interaction between the editors and us to keep the project moving along, and so, yeah, it was a pretty

DISS interview\_7 there is a pretty brisk conversation going on about all matter of things

# 4) The Association of University Presses is the Main Knowledge-sharing Platform

### Emerging Themes by Organization Size (The Association of University Press is the main knowledge sharing platform)

	University	
Organization Size	Press	Coded Segment Examples
Large Organization Size (50 staff or more	DISS	we're looking outside to the Association of University Presses, where we're determining what our peers are doing.
than 50 staff)	interview_3	
	DISS	In fact, when we're going to the AAUP meeting next week, I asked everybody who went to understand that when they come back, they're
	interview_3	going to need to share this knowledge.
	DISS	We just had our annual meeting, and anyone who comes into the community is amazed at how open people are with information. I think
	interview_5	that's something we're proud of
Medium Organization Size (25 staff or less	DISS	All Praces annual masting
than 49 staff)	interview 8	A 0 116565 annuar intering
ului 47 starry	interview_0	
Small Organization Size:(less than 25 staff)	DISS	a mer a ser est
5	interview 1	a very active e-mail listserv with our professional association
	DISS	
	interview_4	we rely on our society, our association to kind of give us the tools and background to help us train new employees.
	DISS	I'm a strong advocate of the AUP, which is the Association of University Presses. It's an umbrella group that nearly all American
	interview 7	university presses belong to. That group represents us to a larger public, and also provides advice about legal matters and about day-to-day
	Interview_/	operations, and keeps us updated on legislation that might be coming, or even things as minor as announcing new projects, and things like
	DISS	a variety of listservs that the organization can belong to, including one for directors that is more about what directors are up to and what
	interview_7	we're thinking about sometimes strategically.
	DISS	a very active e-mail listsery with our professional association
	interview_1	
	DISS	We rely on our society, our association to kind of give us the tools and background to help us train new employees.
	interview_4	The state of the ATTD which is the Accessing of the investigation of the second s
	DISS	The a sholing advocate of the AUP, which is the Association of Oniversity Presses. It's an uniform a group that hearty an American
	interview_7	university presses before to 1 in a group represents us to a rarger public, and also provides advice about regar matters and about day-to-day
	DISS	operations, and aceps us updated on registration that might be coming, of even unlings as minute as announcing new projects, and unlings inc
	interview 7	a variety of inside vs that the organization can belong to, including one for threefors that is more about what uncertors are up to and what
	DISS	we're uninking about sonietinies suategreany.
	interview 1	a very active e-mail listserv with our professional association
	DISS	
	interview 4	We rely on our society, our association to kind of give us the tools and background to help us train new employees.
	-	I'm a strong advocate of the AUP, which is the Association of University Presses. It's an umbrella group that nearly all American
	DISS	university presses belong to. That group represents us to a larger public, and also provides advice about legal matters and about day-to-day
	interview_/	operations, and keeps us updated on legislation that might be coming, or even things as minor as announcing new projects, and things like
	DISS	a variety of listservs that the organization can belong to, including one for directors that is more about what directors are up to and what
	interview_7	we're thinking about sometimes strategically.
	DISS	a vary active a mail listeery with our professional association
	interview_1	a very active e-main risider v with our professional association
	DISS	We rely on our society, our association to kind of give us the tools and background to help us train new employees
	interview_4	we very on our source, our association to know or give as are noted and outsiground to not us taun new oliphoyees.
	DISS	I'm a strong advocate of the AUP, which is the Association of University Presses. It's an umbrella group that nearly all American
	interview 7	university presses belong to. That group represents us to a larger public, and also provides advice about legal matters and about day-to-day
		operations, and keeps us updated on legislation that might be coming, or even things as minor as announcing new projects, and things like
	DISS	a variety of listservs that the organization can belong to, including one for directors that is more about what directors are up to and what
	interview_7	we're thinking about sometimes strategically.

# 5) No Knowledge Management System in Place for Knowledge Retention

	(No Knowledge Ma	nagement System in place to archive data)
Organization Size	University Press	Coded Segment Examples
Large Organization Size (50 staff or more than 50 staff)	DISS interview_3	I think now that I'm talking to you and thinking about the discipline of knowledge sharing, making sure that knowledge is documented, I think we need to do a better job of it, okay?
	DISS interview_3	It's something that I need to learn more about. I've certainly heard about this and read about this discipline and what you're studying, but I think that there is probably a lot more that we need to know as an organization moving forward. It could end up being part of this culture assessment, the culture visioning, the culture strategic planning that we might move forward with. Anyway, so I really appreciate you being here and bringing this to top of mind 1 think we can improve
	DISS interview_5	I haven't talked about knowledge sharing with other organizations
	DISS interview_5	Tell me what you mean by sharing knowledge. Do you mean within the organization, between our organization and other organizations?
Medium Organization Size (25 staff or less than 49 staff)	DISS interview _8	We use a performance management system where people set goals at the start of the year, in July, the financial year. They set goals at the start of the year, they sign those goals with their supervisor, and then they get evaluated at the midyear and also at the end of the year. People who have shown evidence of reaching beyond their own department get a higher grade at the end of the year, and that is tied to some financial reward, as well.
Small Organization Size:(less than 25 taff)	DISS interview_1	we're a very specific kind of operation, there aren't necessarily other people at the university who understand exactly what it is that we do, so that sort of institutional memory and training is very important, because it's a specialized business. once you've involved marketing in that discussion, they may see needs very differently for
	DISS interview_1	that book, so bringing them in very early, and sometimes literally when the book is just an idea or a five-page proposal, rather than a finished book that's just sort of being handed to marketing to be worked on. I think that kind of early knowledge sharing promotes institutional buy-in, everyone understands even before the book is transmitted sort of what it is, who it's for, how we're going to handle it in house. We also don't, I think, have as good a record of different versions of things, so if, for
	DISS interview_2	example, a procedure should change, we don't necessarily know when that happened or who actually changed the documents. We do lose a little bit of information by doing it tha way, but I guess, ultimately, I decided that it was more important that we had everybody buy into the concept of having a centralized place for the information, rather than which specific tool we were using for that centralized information.
	DISS interview_4	I think what a university press does, and what their mission is, is to share knowledge and to make voices heard. I mean that's the goal of a university press.
	DISS interview_6	I will say the one thing I wasn't sure about, referring to my question in the beginning, "Did this relate to knowledge sharing within the company?" I wasn't sure if it also referred to some of these big notions of open access and the ways in which scholarly publishing functions in the world right now.
	DISS interview_6	When you are speaking about shared knowledge, Judi, are you sort of speaking about the issue of sort of internal communications and systems?
	DISS interview_1	we're a very specific kind of operation, there aren't necessarily other people at the university who understand exactly what it is that we do, so that sort of institutional memory and training is very important, because it's a specialized business.
	DISS interview_1	once you've involved marketing in that discussion, they may see needs very differently for that book, so bringing them in very early, and sometimes literally when the book is just an idea or a five-page proposal, rather than a finished book that's just sort of being handed to marketing to be worked on. I think that kind of early knowledge sharing promotes institutional buy-in, everyone understands even before the book is transmitted sort of what it is, who it's for, how we're going to handle it in house.
	DISS interview_2	We also don't, I think, have as good a record of different versions of things, so if, for example, a procedure should change, we don't necessarily know when that happened or who actually changed the documents. We do lose a little bit of information by doing it that way, but I guess, ultimately, I decided that it was more important that we had everybody buy into the concept of having a centralized place for the information, rather than which specific tool we were using for that centralized information.
	DISS interview_4	I think what a university press does, and what their mission is, is to share knowledge and to make voices heard. I mean that's the goal of a university press.
	DISS interview_6	I will say the one thing I wasn't sure about, referring to my question in the beginning, "Did this relate to knowledge sharing within the company?" I wasn't sure if it also referred to some of these big notions of open access and the ways in which scholarly publishing functions in the world right now.
	DISS interview_6	When you are speaking about shared knowledge, Judi, are you sort of speaking about the issue of sort of internal communications and systems?

# 6) Knowledge Curation is Limited

(Know	vladae Curstion is limit	Emerging Themes by Organization Size
Organization Size	University Press	Coded Segment Examples
Large Organization Size (50 staff or more than 50 staff)	DISS interview_3	what the staff enrichment team has decided is that each person will write a pithy paragraph about what they've learned, and that will be the next newsletter so that people can learn about that.
	DISS interview_3 DISS interview_3 DISS interview_5	doing a newsletter process manuals There are a lot of other things to do. It's not a matter of not knowing how to do it, or at least that's the message I am being given: "We can do it but it will distract from many other things, and how high a priority is this?"
Medium Organization Size (25 staff or less than 49 staff)	DISS interview _8	various systems of documentation
	DISS interview _8	We also have written documentation about various processes, and a version of the guidelines that we share with our authors, which is a little bit more detailed for internal staff. Our internal processes are captured in documents that we call standard operating procedures, SOPs, and those are Google documents that are always called SOP in their title so they're easy to find. They document quite a lot of information about particular processes, and then we also have an integrated tile management system. This is technology that connects staff in their workflow in producing a book, so they have logins and they have different roles, and it's where we gather the information about the book that's then transmitted out to book retailers, to jobbers, to people who need to know about the book
	DISS interview _8 DISS interview _8 DISS interview _8	we have an intranet within the publishing division, and we also have an intranet within the library. There is a newsletter every Friday that goes out about new initiatives There are also project management software that we use, so Asana, for example, and Slack. This is more used in the sort of more technology-heavy production side of the building. We use Trello. For different projects, we're actually using different project management software, and they communicate in that way.
Small Organization	DISS interview_1	One of the things that we are trying to do a better job of, again, is getting information and notes into the central database system
Size.(iess than 25 starr)	DISS interview_1	the technology we already have in terms of really having all activities centered around the title management database.
	DISS interview_1 DISS interview_1	we have a press-wide database that the press has been using for a number of years. I subscribe to a number of daily e-mail newsletters having to do both with the book industry and with hister education so those are probably the primary sources
	DISS interview_1	I'll wait for the digest versions of, "What are the articles I really should be reading?" and then keep up that way. In house, we get Publishers Weekly, The Chronicle of Higher Education
	DISS interview_1	I would also do for other retiring employees is having them document timelines, write notes about essential components of the job
	DISS interview_2	also have a shared server
	DISS interview_2	Each of the departments created a folder on that shared server where they would keep that sort of institutional information that they've been gathering and that was relevant to their particular department. Everybody has access to all of those documents, so it's still being shared, it's just not as a server being the start of
	DISS interview_2	on the Wiki,
	DISS interview_4	There is documentation that we share with them
	DISS interview_4	through e-mail
	DISS interview_6	But this sharing of communication, these sort of divergent communication styles have been, I would say, the cause of when there is sort of antipathy and frustration. It is because of the sort of older-
	DISS interview_6	they still have a habit of using many, many e-mails, back and forth, back and forth, and back and forth
	DISS interview_6 DISS interview_6	through e-mail There is a place on this database that I mentioned so that every single book, the information is updated weekly. When we look at it for our updates once a week, all of the production and editorial
	DISS interview_6	information is completely up to date. they're all up on the shared drive
	DISS interview_6 DISS interview_7	we use is a shared drive the technology piece is not a major focus, but we think of it as useful tools, and also things that we could archive and go back and look at stuff and be able to track how we got here. That's kind of how we think about the technology.
	DISS interview_7	Yeah, it's an incredibly complicated project, so we still have boxes and boxes of old copy as related to this somewhere
	DISS interview_7 DISS interview_7	For most of the jobs here, we don't actually have a manual of standard operating procedures. That's one of the ways, probably the primary way that I keep up with issues that are related to publishing. I read some publications, too. I read Publishers Weekly and things like Choice and Library Journal, and things like that that are really focused usually on publishing issues specifically, but also scholarly communications in general. I guess those are some of the primary ways that I stay updated on things
	DISS interview_7	we use SharePoint for our editorial board. When we gather reports, we share documents that way, and things like minutes for the meeting and all that kind of shuff
	DISS interview_7	We use a weekly wiki for staff meetings to post agendas and have people add to it, so we do

### Emerging Themes by Organization Size

### (Outliers' Activities)

Organization Size	University Press	Coded Segment Examples
Large Organization Size	DISS	I sensed that when I came in, I observed that when I came in. with fresh eyes, that's a
(50 staff or more than 50	interview_3	good thing, but I also wanted this consultant, who I had worked with on other projects
staff)		in the past, come in and see if I was correct.
	DISS	We'll be hiring a consultant to help us do our own self-assessment. That person will
	interview_3	also then be comparing us to competition, and so that will be big knowledge gains,
		where we're learning what other people are doing in our field, not just university
		presses, but for-profit presses and other people who we really do compete with in the
		marketplace.
	DISS	The librarians are purchasing the platform, purchasing the aggregation, but it's for
	interview_3	patrons, and most of those patrons are scholars in the humanities and social sciences.
		That's one of the big projects where we're not just reading about best practices in
		platform development, not just reading about programming, but actually want to know
		what our users want to know, and so we're doing research like you're doing:
		quantitative surveys, qualitative interviews, putting people in front of the screen. That's
	DIGG	a big, major knowledge acquisitions
	DISS	we will do a gap analysis and determine what kinds of activities and programs we need
	interview_3	to put in place to get the vision to where we want to be, and then, in a year, we in
	DISS	Incasure again We also just finished a major organizational culture assessment, and so that was done
	interview 3	we also just infisited a major organizational earlier assessment, and so that was done with quantitative and qualitative work so we did a survey across all 130 people who
	interview_5	work here and we got a response rate of 88 percent 111 responses to that
		work here, and we got a response rate of 55 percent, 111 responses to that.
Small Organization	DISS	In order to look at this reorganization, we brought in a trainer from our own university
Size:(less than 25 staff)	interview_1	who was offered to us. We're still a little bit in progress with this, but she is kind of
		guiding us through what are the steps of our current procedures, and then we put all of
		those down and who is responsible for what and in what order. Then, after studying
		that, the next phase is going to be to look at, "Okay, what is the streamlined workflow?
		We're still a little bit in the middle of that, but I would say that is a positive example
	DISS	For example, marketing is one area. I was the marketing director at the previous press
	interview 2	that I worked at. When I came here, marketing, we were only publishing at that point
		monographs, and the marketing on it was very straightforward. But, since then, we
		have started publishing more trade books, and nobody here had any experience with
		trying to market trade books to a national audience. While I was at my previous job,
		we did actually publish quite a few trade books, so I knew a little bit more than my
		colleagues here about how that's done.
	DISS	When I first came here, we were approached by a faculty member about publishing
	interview_2	actual comics. I think that our experience publishing comic studies helped to lead the
		way in how to actually publish graphic novels. We were already used to using comics
		in our scholarship, but I think, also, to be frank, I'm a comic lover myself, and
		particularly of graphic novels. I had a lot of them, I've read a lot of them, and so I think
		that might be an example where my personal experience was useful in helping both the
		marketing department and the production department deal with particular challenges
		that are specific to publishing comics.

### Philosophical Assumptions of the Mixed Method Approach for this Study

Beginning in the 1980s (Green et al., 1989), the mixing of qualitative and quantitative data in a single study has been popularly termed as mixed methods research (Johnson et al., 2007). In fact, mixed methods research is defined as "the type of research in which a researcher ... combines elements of qualitative and quantitative research approaches (e.g. the use of qualitative and quantitative viewpoints, data collection, analysis, and inference techniques) for the broad purpose of breadth and depth of understanding and corroboration" (Johnson et al., 2007, p. 123).

Mixed methods approaches came from a pragmatism viewpoint (Patton, 1990). Pragmatism does not commit to one system of philosophy and is open to using multiple methods, different assumptions and different data collection methods. The researchers focus on the research problems and use the appropriate approach to understand the particular problem, instead of focusing on methods (Rossman & Wilson, 1985).

The design of mixed methods research addresses a problem in depth through qualitative data collection and analysis on the quantitative results (Creswell, 2015). Sequential mixed methods design collects data in a predefined sequence and then merges the results of the qualitative and quantitative data analyses.

For this mixed methods research, the researcher began with a quantitative strand (a bottomup approach) and then a qualitative strand (a top-down approach) to compare to the quantitative results. This order provided multi-dimensional views from the employees' perspectives as well as from the management's perspectives. With the mixing of the two datasets, the similarities and differences in the data can be identified to generate a better understanding of the research topic.

### **Data Integration and Interpretation – QUAN + QUAL**

The process of integrating the data from the QUAN + QUAL sets occurs at the end of a mixed methods study, bringing these datasets together after both datasets have been analyzed separately. The researcher listed the findings from each component of the study and considered whether findings from each method agree (convergence), offer complementary information on the same issue (complementarity), or appear to contradict each other (O'Cathain, Murphy & Nicholl, 2010).

There are four main types of integration of quantitative and qualitative data in a mixed methods study (Bazeley, 2009; Fetters et al., 2013; Maxwell & Loomis, 2003; Creswell, 2015): merging the data, explaining the data, building the data, and embedding the data.

For this mixed method research, the approach of "merging the data" was selected for the data integration and interpretation process, to compare the findings on the topic from the employees' level to the directors' level. This type of data integration allows each data set to occur as a single-method study, with the goal of integrating the results after both are completed (Creswell & Plano-Clark, 2011). At the point of mixing, conclusions or inferences were drawn to reflect what is learned by combining, comparing and synthesizing the study findings (Creswell, 2015).

At the end of the data analysis process, the mixing was conducted in order to determine how well the qualitative and quantitative data were integrated. The integration process started using the themes identified in the qualitative data set as the yardstick to match with the findings from the quantitative data (see Table 24). Several instances of convergence were identified from the comparison of these two datasets. The findings in the quantitative research portion matched with and supported the themes identified in the qualitative research portion.

				Conceptual
#	Convergence	Evidences in QUAN	Evidences in QUAL	Frameworks
				Applied
1	Trust is the social norm in this close-knit society	Question 21 presented three statements on the respondents' perceptions about their coworkers. All three statements "I feel trustworthy with my coworkers' information," "If I encounter problems in finishing my job, I know my coworkers will help me out when I ask" and "My coworkers are willing to help and do not deceive for their own profit" received higher than 75% of the total counts added up in "Agree" and "Strongly Agree" categories.	DISS Interview_2 - I think, in the university press world where you don't hire somebody who doesn't have some experience in scholarly communications	Dervin's Sense- Making Metaphor
2	Full of Collegiality in the Organizational Culture	Question 20 asked respondents about their degree of motivation to share their knowledge and experience. 62.4% of 93 respondents indicated they were very motivated because their sharing would build a better company. 28% of the respondents indicated they felt an average level of motivation because they wanted to help their coworkers. This indicated a collegiality existed in the presses.	DISS Interview _1 - One interesting thing about university presses, we're very collegial, and so a lot of times we'll ask each other for advice. DISS Interview _2 – We are a very, I think, egalitarian and collegial organization. DISS Interview _5 - I would say the university press community, in general, is amazingly collegial.	Dervin's Sense- Making Metaphor Weick's Sensemaking Theory

Table 24: Mapping of the QUAN and QUAL Datasets to the Conceptual Frameworks

Table 24: *Mapping of the QUAN and QUAL Datasets to the Conceptual Frameworks* (Cont'd)

#	Convergence	Evidences in QUAN	Evidences in QUAN Evidences in QUAL	
3	In-person communication is important	When respondents were asked in Question 24 about formal activities for sharing knowledge and experience, 42.53% of respondents chose "Team Meeting" and 32.18% wrote in about different types of in-person meetings.	DISS Interview 3 - Our marketing department is split between floors here, and so I've been told that that's one of the reasons why the marketing department is not collaborating and sharing knowledge as they should be. The word "meeting" appeared 34 times and the word "conference" appeared 6 times across the eight interview transcripts	Weick's Sensemaking Theory
4	Knowledge management concept is unfamiliar	When respondents were asked in Question 26 what changes would they like to see in knowledge sharing inside their organizations, 28% of respondents would like to have better technology system, tools and training; and 7% of respondents wanted to have a central hub that can be accessed by staff to avoid duplicated effort in documentation.	DISS Interview_6 - I will say the one thing I wasn't sure about, referring to my question in the beginning, "Did this relate to knowledge sharing within the company?" I wasn't sure if it also referred to some of these big notions of open access and the ways in which scholarly publishing functions in the world right now.	Nonaka and Takeuchi's SECI Model

In general, four convergences were found to be in common across the QUAN and QUAL portions. The following paragraphs illustrate these convergences in detail.

## **Convergence #1 – Trust is the Social Norm in this Close-knit Society**

Trust was part of the social norm concerning the relationship among staff in the university presses. Staff were also empowered by management to share ideas. They trusted each other and believed coworkers were helping each other with a sincere, not selfish mindset. Evidence for this insight was found in both the QUAN and QUAL datasets. From the quantitative survey Question 21, 81 out of 91 participants believed that their coworkers were willing to help them and their coworkers would not deceive them for their own profit. Small to medium size university presses were found to have a tighter-knit society.

The findings from the QUAN portion also matched a theme found in the transcripts from the QUAL portion, for example:

DISS interview\_7 I know that anybody here would do anything for me if I asked them to, and I think that they feel that way about one another, too. There is not a single position where somebody is really off by themselves.

### **Convergence #2 – Full of Collegiality in their Organization Culture.**

There was evidence that the university presses are full of collegiality in their organizational culture. Organizational members were willingly to share and help each other. From the quantitative research, on Question 19 about how they are motivated to share knowledge and experience, 92 out of 93 participants indicated they were motivated to share their knowledge. Fifty-eight out of these 93 participants chose to say they were "very motivated [to share their knowledge] because I think sharing mine will build a better company." The findings from the QUAN study on all employees matched a theme found in the transcripts from QUAL, as several directors used the exact word "collegial" when they described their organizations in general.

DISS Interview_1	One interesting thing about university presses, we're very collegial, and so a lot of times we'll ask each other for advice
DISS Interview_2	We are a very, I think, egalitarian and collegial organization
DISS Interview_5	I would say the university press community, in general, is amazingly collegial.

### **Convergence #3 – In-Person Communication is Important**

All sorts of meetings such as staff meetings, project meetings, brown bag lunch presentations/discussions and drop-in meetings across the presses were the main venue for sharing. From the quantitative research, Question 24 asked about formal activities for sharing knowledge and experience, and 42.53% of respondents chose "Team Meetings" and 32.18% chose to use a write-in response about different types of in-person meetings. The findings in the QUAN section on all employees matched to a theme in the transcripts from QUAL, as the word "meetings" was mentioned thirty-four times across the eight interview transcripts. In fact, two directors indicated that the proximity of staff offices (i.e., opportunities for staff members to have direct encounters) would be a factor in the level of knowledge being shared in their organizations.

DISS Interview\_3 Our marketing department is split between floors here, and so I've been told that that's one of the reasons why the marketing department is not collaborating and sharing knowledge as they should be.

# **Convergence #4 – The Knowledge Management Concept is Unfamiliar.**

Evidence from the study indicated that the concept of knowledge management is unfamiliar to members of university press organizations. Internal knowledge-sharing took place via internal meetings within the organization. The main vehicle for external knowledge-sharing was the Association of University Presses' annual conference and their listservs. There was no evidence that any sophisticated knowledge management system had been implemented. This finding was reflected in the quantitative survey's open-ended Question 26, with the result that 28% of the 68 narratives called for better technology facilities and training. The second highest frequency level on this measure was that 15% of the 68 narratives sought more inter-departmental knowledge sharing. These findings from the QUAN portion aligned with a theme in the transcripts from the QUAL portion, as questions were being asked in the interviews about what knowledge management really was about. For example:

DISS Interview\_6 I will say the one thing I wasn't sure about, referring to my question in the beginning, "Did this relate to knowledge sharing within the company?" I wasn't sure if it also referred to some of these big notions of open access and the ways in which scholarly publishing functions in the world right now

### Validity and Reliability

# QUAN

For the quantitative approach, validity includes external validity and internal validity. Validity is about the research accuracy, the truthfulness of the data and the degree to which the data is representative of and generalizable to the general population. The demographics portion of the survey questionnaire was used to identify the representativeness of the data from the sample. The second part of the survey questionnaire was designed with a reference to several published survey articles on knowledge-sharing. To ensure face validity, a mini-version of a pilot study, among librarians who worked closely with the University of Tennessee's University Press, was conducted to pre-test the survey research instrument. A statistical power analysis was conducted to calculate the required sample size and to estimate the margin of error. Although the collected samples in the QUAN met the required sample size in the power analysis, the low response rate in this survey was a concern regarding the meaningfulness of the data. This concern was offset by the conclusion of the article "Does Response Rate Matter?" (Carley-Baxter et al., 2009). The findings of that research article indicated that "it would appear that the perception among social

science researchers that journals weight response rates heavily in the manuscript review process is unfounded". Instead of making judgments merely on the response rate, the article concluded that "most journal editors think about any manuscript's worth or merit based more on intangible or global concepts, such as design (be it sample or questionnaire design) than they do on measures of survey quality" (Carley-Baxter et al., 2009). Therefore, the survey design is more important in gauging the study's external validity.

Reliability was assessed through internal consistency (Singleton & Straits, 2010). Cronbach's alpha (Cronbach, 1951) was utilized to estimate the reliability (internal consistency) of the measurements and to get a sense of the validity and reliability of the particular scale that related to the sample (Grinnell & Unrau, 2010). Cronbach's alpha tests were conducted and produced over .70 on reliability on both the culture and trust scales. According to Hayes (2005), a Cronbach's alpha of .70 is "the magic number" for the scale's reliability (p. 128).

Relial	Reliability				
Scale	Scale: Culture				
Cas	Case Processing Summary				
			N	%	
Cases	Valid		95	89.6	
	Exclud	led <sup>a</sup>	11	10.4	
	Total		106	100.0	
a. Listwise deletion based on all variables in the procedure. Reliability Statistics					
Cronbach's Alpha Based on Standardized					
Alpi	ha		tems	N of Items	
	.718		.722	11	

Figure 9 - Reliability - Scale of Culture

Reliability				
Scale: Trust				
Case Processing Summary				
			N	%
Cases	Valid		91	85.8
	Excluded <sup>a</sup>		15	14.2
	Total		106	100.0
a. Listwise deletion based on all variables in the procedure.				
Cronba	Cro Alph		nbach's a Based on dardized	
Alpl	ha	1	tems	N of Items
	.752		.753	3

Figure 10 - Reliability - Scale of Trust

In addition, methodological triangulation for this mixed methods research was also used to verify the truth of the study's findings, with the combination of two data sources to investigate the same phenomenon, in order to counterbalance the deficiency of a single strategy and thereby increase the credibility and validity of the results.

#### QUAL

For the qualitative approach, validity is about the appropriateness of the method used and reliability is about consistency. Qualitative researchers focus on the causes of bias rather than on eliminating them. Therefore, as mentioned in the literature (Feilzer, 2010), the following section provides a reflexive section in the finding report about the role of the researcher in the research process, and the context of the research design and its methods. To ensure the reliability of qualitative data, constant data comparison was used to identify emerging behavior categories, and the triangulation method was used to compare patterns from the qualitative research with the findings of the quantitative research (Creswell, 2007). Quantitative and qualitative methods in this mixed methods research were combined with the goal of increasing the validity of the measures through triangulation. Several themes in the phenomenon were assessed to determine if convergence across the methods exists, to generate a deeper understanding (Edmonson &

McManus, 2007).

In the data analysis stage, the researcher used Grant McCraken's framework to determine the categories, relationships, and assumptions that inform the respondent's view of the topic (McCraken, 1988). The coding process identified words and phrases that related to relational/interpersonal factors as well as those indicating content factors. For example, when the participant spoke about a coworker who was "collegial," this was categorized as relational. When the participant spoke of users who "knew what they were doing," this was classified as content.

### **Reflexivity Statement of the Researcher**

An individual interview method was used for this qualitative research and the researcher acted as an instrument to collect data. Each interview was conducted according to the participant's preferences, either face to face or over an electronic medium, as that was most convenient and comfortable for the interviewee. The researcher was mindful that the participants were given the lead in "setting the pace" of the interviews. By adopting a "taking a back seat" style in using a less active setting for the interviews to take place, the participants were provided with a feeling that they were exercising a measure of control over the interview process.

Within the context of this research study of a scholarly publishing setting, the researcher took into consideration that the interaction with participants might be influenced by her own professional background, experiences and prior assumptions. Knowing about the researcher's professional background and affiliation with academic fields could have impacted participants' willingness to talk openly and truthfully about their experiences, or affected how this knowledge might have shaped what was said.

### **CHAPTER FIVE**

### **DISCUSSION AND CONCLUSIONS**

This chapter provides a discussion, lessons learned and recommendations as conclusions. The limitations of the study are also presented, along with suggested areas for future studies that may build on these findings.

### Discussion

Scholarly literature on knowledge sharing and knowledge transfer have been saying that an open culture and trust are the two basic factors that are most influential on individuals' knowledge seeking and sharing. The level of knowledge sharing is significantly impacted by the level of trust (Rutten, Blaas-Franken & Martin, 2016). The investigation and findings of this research support this assertion. With a collegial culture and a trustful social norm in university presses, this harmonious environment has cultivated a positive atmosphere for knowledge sharing and transfer among employees.

This study was designed to understand knowledge practices of university presses, with specific foci on behavioral, cultural, social and technological issues. An examination of the findings on all of these issues revealed that employees' personal knowledge-seeking behavior was shaped to a certain degree by the culture and social norm in which they were immersed. Personal interactions to share tacit knowledge in meetings and at conferences were the main channels for individuals to seek knowledge and for coworkers to share knowledge. By applying Dervin's Sense-Making Metaphor, the individuals working within the context of a collegial culture, with heavy levels of trust in other people as knowledge sources, are more likely to engage in knowledge sharing and transfer behavior to fill the tacit knowledge gaps of the individuals involved.

In fact, the university presses' unified social norm can be seen in their use of the popular literature (explicit knowledge sources) in their specific professional areas such as Publishers Weekly and Scholarly Kitchen. In the results from the survey and interviews conducted for this study, the Association of University Presses (AUP) email listservs were considered to be the major knowledge sharing platform for news and updates and for connecting with their fellows in the presses.

In reviewing the types of knowledge barriers previously identified by scholars, the university presses do not have personal barriers as the employees are motived and eager to share knowledge, facilitated by their open, western, organizational culture. However, the presses do face content barriers when employees do not understand some procedures, such as how to compile the financial statement that was mentioned in one director's interview. Technological barriers were also identified as major deterrents to facilitating knowledge infusion among departments.

In general, being too busy and the unavailability of coworkers and technologies were the major issues with knowledge sharing, that were recognized by university press employees. Issues with the geo-proximity of staff was the major concern on the directors' level. It is not surprising that the dearth of sufficient time is a barrier to engaging in knowledge curation, given that many employees are covering more than one position, with the downsizing trend of university presses. However, the fact that the current knowledge-sharing behaviors relied solely on people's interactions, and that the presses are operating without a proper explicit knowledge management system, these conditions may be the reasons the employees are perceiving that unavailable coworkers and the geo-proximity of staff are the knowledge barriers in seeking operational knowledge. The lack of a good technology system and training was identified as a barrier in the survey of all employees. This particular barrier was considered to be a more major obstacle in

small-size press organization sizes than in large-size organizations. There was a high frequency level for the write-in response recognizing the need for "better technology systems, tools, training" for knowledge management.

Another objective of this research study was to explore the current knowledge retention practices in university presses. Most certainly, there were knowledge-sharing activities occurring in the presses. However, although knowledge was being shared, it was not being captured for future uses, so the current approaches might not be a complete solution. The findings indicated there was not only no sophisticated knowledge retention mechanism or policies in place to organize or classify operational knowledge, but also that the concept of knowledge management or knowledge curation were unfamiliar in university presses. This matches with another scholarly article's conclusion that organizations generally do not formulate any knowledge retention strategies even though they believed that knowledge is crucial for success (Liebowitz, 2008). In the sense of knowledge management, this situation is alarming. Even though the university presses are a close-knit society, with internship programs, people retiring or switching jobs within the presses circle, and organizations requiring new hires to have prior experience in their profession, the presses might not realize that a knowledge bleed (Liebowitz, 2008) is happening until the moment an employee actually leaves or retires.

### **Lessons Learned**

Through the process of working on this dissertation, the researcher learned some valuable lessons along the way to share with the readers. The first lesson learned was about the gender question in the quantitative survey. This issue was not found in the earlier literature on survey questionnaire development. For the gender question, participants were asked to identify as male or female. However, societies have changed, and a third option is required. When the Association of University Presses reviewed the survey questionnaire before forwarding the survey out, they
requested offering a non-binary "prefer-not-to-specify" gender option, even though respondents already had the option not to answer the question. The reasoning from the AUP was that the publishing community was becoming much more sensitive to multiple vectors of diversity and the binary gender division had the potential for causing someone to be excluded. With this change request, the survey schedule was delayed to seek an approval from the IRB on the amendment.

Second, retired university press staff members remain listed as subscribers to the Association's listserv. The AUP has two members-only subscription listservs: a general discussion list with approximately 1,200 subscribers and a director listserv with 147 subscribers. Individual university press organizations are not required to report their staff sizes to the AUP, so the Association does not have that kind of information to use to manage their listserv registrations. Since they do not have individual membership records, their data on press employees was not comprehensive. The AUP relied on member presses to volunteer to send their employees' information to the Association for its annual directory update process. Therefore, the two listservs were populated through a purely self-selected subscription process among press members. Former press employees could choose to stay on the listserv even though they were no longer affiliated with the presses. This was the reason that the researcher had to remove one record from the survey sample after it was identified as being from a location where the only university press had closed down in 2002. The lesson learned from this issue is not to assume that everyone subscribed to an active listserv is currently working in the field. A screening question should be included in the survey, to ask if the participant currently works with a university press.

Third, the effects of decisions or assumptions related to questionnaire administration have implications for the research process. Protecting the participants' privacy was one of the golden rules that the researcher kept in mind when the survey instrument was designed. Therefore, questions on respondent demographics were kept to a minimum. However, it was in the data analysis stage that the researcher realized that having an understanding of the organization size associated with each completed survey would help to illuminate knowledge sharing in the operations flow. Although the use of IP addresses in an indirect way was successful in identifying the organization sizes that participants were affiliated with and solved the problem, it was a tedious task. It took the researcher a tremendous of time and effort to fulfill this task, not to mention the requirement of submitting an amendment to IRB for their approval of using IP addresses. It would have saved the researcher a lot of time if organization size was simply self-disclosed by each participant.

Fourth, the rate of return of the completed surveys in this study was below the researcher's expectation, even though the sample size obtained still met the estimate specified by the statistical prior power analysis. The low survey response rate confirmed the scholarly literature discussion that stated that response rates to online surveys always are much lower than those obtained using paper surveys (Cook et al., 2000) and a low response rate to online survey research is an increasingly common phenomenon (Porter & Whitcomb, 2005). Another source said that there is a lower level of interest in taking online surveys than in paper surveys administered in person, with or without incentives (Nulty, 2008). As the responses were only from North America, there was no data obtained to test the planned research question on culture issues, to investigate the impact of racial differences in sharing knowledge.

Fifth, a common phenomenon that was observed was that knowledge management concepts were not well known among the university presses' society. Knowledge sharing was mistakenly interpreted as being the same as the presses' role in disseminating knowledge in the scholarly communication sector. Some interviewees considered knowledge management to be the same as

Human Resources and some considered it to be akin to strategic management. In order to clarify the concepts, the researcher explained the concepts prior to conducting the final two interviews (Interview\_7 and Interview\_8), to test out if there were differences in the collected data as compared to Interviews #1through #6. To conclude this mini-experiment, no differences tied to the experimental treatment were found in the data collected, across the eight interviews.

Last, the interviewees did not feel comfortable with being a solo talker in the conversation when they were being interviewed online or in person. The reason might be the impact of a collegial culture, and its trait of being respectful to anyone in the conversation by not being a dominator. Therefore, if there are more resources available like time and funding, a stratifiedsample focus group interview setting would be a better choice for a collegial culture group than the individual interview setting, particularly for the investigation of the comparison of different ethnic groups' knowledge behavior.

### Limitations of the Study

As with any other research endeavor, there are limitations that fall beyond the researcher's control. The primary one in this case is that this research project tried to understand the knowledge practices in the scholarly publishing environment. Knowledge management is a research topic in business operations research. It is difficult for researchers in business-related disciplines to generalize their research findings across a diverse range of businesses, in order to be useful for a wider audience. Instead, most researchers select a particular niche of a small business sector to examine (Burgess, 2002). Therefore, this research project mainly focused on knowledge-intensive firms, particularly in the scholarly publishing sector of university presses. There are many other types of organizational environments, and their knowledge practices may not be the same as those

of university presses. The findings from this research provide a general understanding of knowledge practices or knowledge-sharing barriers that the non-commercial publishing sector may face, but the research may or may not be adaptable to other industry sectors.

In addition, there are matters and occurrences that arose in this research study that are out of the researcher's control, such as having limited access to only certain people in each organization among the university presses, for the distribution of the survey instrument and administering of the interviews. Some employees and managers chose not to participate or gave incomplete answers. The specific limitations related to the data collection/analysis for this research project are listed in full in the quantitative (QUAN) and qualitative (QUAL) sections.

For the QUAN portion: First, self-selecting non-probability sampling was used, based on participants' volunteering. Non-probability samples may not be very representative, even when the full population was exposed to recruitment. Second, the survey response rate was low, with participation by volunteering only. The survey was mainly filled out by female respondents, so there might be a gender bias, particularly in the technological issues. Survey fatigue has been reported as a well-documented phenomenon (Porter, Whitcomb, & Weitzer, 2004). Some participants in the study dropped out for unknown reasons after completing the demographics section. Although the sample size, after it was cleaned up, met the requirement of the power analysis, the margin of error rose from the standard 5% to 8%. There is a possibility of getting lower-quality responses from the participants toward the end of the survey.

Third, a research quorum existed. Participants were mainly those subscribed to the official listserv maintained by the Association of University Presses. The invitation to take the survey might not have the opportunity to reach those employees or directors who did not subscribe to

these two listservs. Moreover, no survey participants came from the member presses in Europe, Asia or Africa. The survey questionnaires were filled out by participants from North America, which meant that the responses on the culture factor were dominated by an open Western culture. Without enough data, it was impossible to address the potential effect of racial differences and organizational culture to determine the correlation between variables like "culture" and "knowledge-sharing." Because of this reason, the original research question about testing these variables has been removed.

Four, university press members who did not volunteer to participate might differ in some ways from those participants in both the quantitative and qualitative portions. Therefore, results from the participants might have been skewed toward not reflecting a holistic picture of the knowledge sharing practices and perspectives.

Here are the limitations of the study for the QUAL portion: First, the method employed in this research was the long interview with a critical incident approach, which meant that the participants had to remember stories that occurred among themselves and their coworkers. As the knowledge management concepts were unfamiliar to participants, many were puzzled at first by some questions. The participants were able to recall one or two particular incidents from their organizational setting, but they usually reverted to talking about the general matters they shared about in their routine publishing activities.

Second, the nature of this research project was a doctoral dissertation and so the coding and interpretation were conducted by one person. The researcher was the instrument in data collection. Personal bias might occur in interpretation, as different people might have varied points of views about coding and grouping.

Third, the results of the interview data might be skewed by the way in which they were collected. This study relied on the perceptions and experiences of the directors. Since the researcher was associated with the academic field under scrutiny, the participants' responses could have been influenced by social desirability. In other words, the participants might have felt pressured to answer in ways that did not truly reflect their opinions. If the interview data had been collected by a person with no connection to the academic field, perhaps the response would be different (Patton, 2002).

### **Areas for Future Research**

Some of the limitations described above may also be applicable as areas for future research. These include but are not limited to the following. First, a follow-up qualitative study involving non-participants from university presses outside North America should be conducted to investigate the reasons for their non-participation in this research. Second, because no data about ethnicity were obtained in this research, a future comparative quantitative study using a stratified sampling method should be used to examine the demographic differences, particularly for the ethnicityculture interaction issue, in knowledge-sharing practices.

This research project in the knowledge management field may be helpful in introducing a structured approach to study knowledge sharing within knowledge-intensive firms in scholarly communication. The findings should lead to the investigation of how possible positive results can be transferred to other commercial sectors in terms of the organizations' knowledge needs, specific knowledge assets, knowledge gap issues, barriers to knowledge flow and knowledge management.

Similar to Nunes's study (Nunes et al., 2006) stating that "Knowledge acquisition and embodiment was perceived as a crucial task" by all interviewees, all interview participants in this research project unanimously agreed that knowledge sharing is a source for innovation. Several interviewees indicated that this research project reminded them to think about knowledge management. However, they were unsure if their respective organizations were capturing and managing this core intangible asset as effectively as they should be. Curiously, they all agreed that knowledge should be better managed and stored within their own organizations, and that the curation of knowledge could probably result in greater innovation and profitability for their services. It would be a good follow-up research topic, to study how and what the directors have done to address the knowledge management mechanism after being interviewed.

#### Recommendations

Initially, I had so many ideas for a dissertation topic, but I found my passion for this research (and my Ph.D. topic) through my 2016 individual study project about knowledge management by major commercial scholarly publishers, that was a part of my program course work. I thought I knew about knowledge management as a topic at that point, but very soon I realized I had more questions than answers. Therefore, this research on the knowledge-sharing topic is just at the tip of the knowledge iceberg in the research field of knowledge management. Through the interviews and the survey for this research, it is hoped that they would be able to reveal the importance of knowledge-sharing that the university presses need at this moment for sustainability.

Prior to starting the interview, the researcher made small talk with each director. Later, it came to the researcher's attention that there are similarities in the situations facing university presses and academic libraries for survival. Both sectors are not-for-profit oriented, and they exist for disseminating scholarly information. There are sustainable business models that the academic libraries have used, that can also be applied to the university presses, so as to visualize the organization's value to the host university, to conduct outreach to faculty to cultivate collaboration

opportunities, and to enhance the publishing role in preserving and disseminating their own university's faculty's intellectual knowledge.

Per the discussion about the three research questions contained in the Discussion section, it underscored that there is an urgent need for university presses to formulate a knowledge retention strategy. It needs to identify what knowledge may be lost, what the consequences of loss are to their organization and what actions it should take to retain that knowledge (Doan et al., 2011).

The university presses rely heavily on the Association of University Presses as a medium to share information and knowledge. The researcher suggests creating local chapters or consortia within the Association. The local chapters can enhance the sharing of knowledge for innovative ideas on special topics in the local communities as well as to enhance the uniqueness of each university press in publishing with specific subject focuses to represent local scholars. The presses may consider adopting the strategies that commercial publishers are using and should cultivate more collaborative opportunities with their university faculty or their university libraries.

# LIST OF REFERENCES

Ackoff, R. L. (1989). From data to wisdom. Journal of Applied Systems Analysis, 16(1), 3-9

- Academic Books 'at a crossroads' (2017, June 13), *Research Inform*. Retrieved from http://researchinformation.info
- Ahmed, A. & Ahsan, A. (2014). An integrated organizational knowledge management framework (IOKMF) for knowledge creation and usage. *Journal of Strategy & Performance Management*, 2(1), 17-30.
- Alavi, M., & Leidner, D. E. (1999). Knowledge management systems: Issues, challenges, and benefits. *Communications of the AIS*, 1(2es), 1.
- Al-Suqri, M. N. (2015). *Information seeking behavior and technology adoption: Theories and trends*. Hershey: Information Science Reference.
- Alvesson, M. (1995). *Cultural perspectives on organizations*. Cambridge, Mass: Cambridge University Press.
- Alvesson, M. (2004). *Knowledge work and knowledge-intensive firms*. Oxford: Oxford University Press.
- Andersen, A. (1996). *The Knowledge Management Assessment Tool (KMAT)*. London: Arthur Andersen KMAT Study.
- Annas, J. (2003). Plato: A very short introduction. Oxford: Oxford University Press.

Antonius, R. (2003). Interpreting Quantitative Data with SPSS. London: Sage.

- Argote, L., & Ingram, P. (2000). Knowledge transfer: A basis for competitive advantage in firms. Organizational Behavior and Human Decision Processes, 82(1), 150-169.
- Arksey, H., & Knight, P. T. (1999). Interviewing for social scientists: An introductory resource with examples. London: Sage.

Assinform (2010). Assinform Report, 41th edition, Milan, 2010.

- Atherton, A., & Elsmore, P. (2007). Structuring qualitative enquiry in management and organization research: *Qualitative Research in Organizations and Management: An International Journal.* 2, 62-77.
- Badaracco, J. L. (1991). *The knowledge link: how firms compete through strategic alliances*.Boston, Mass, Harvard Business School.
- Bala, J. (2016). Contribution of SPSS in social sciences research. International Journal of Advanced Research in Computer Science, 7(6).
- Bazeley, P. (2009). Mixed methods data analysis. In S. Andrew & E. Halcomb (Ed.), *Mixed methods research for nursing and the health sciences* (pp. 84-118). Chichester, UK: Wiley-Blackwell.
- Bennett, N., Wise, C., Woods, P.A. & Harvey, J.A. (2003). *Distributed Leadership*. Nottingham: National College of School Leadership.
- Bhatt, G. D. (2001). Knowledge management in organizations: examining the interaction
  between technologies, techniques, and people. *Journal of knowledge management*, 5(1), 68-75.
- Breunig, K. J., & Christoffersen, L. (2015). Researching the Outliers: Bridging the Rigor-Relevance Divide in Management Research With CIT? *Leading issues in business research methods*, 2, 33-49.

Brick, M. (2011). The future of survey sampling. Public Opinion Quarterly, 75(5), 872-888.

Brown, J.S. & Duguid, P., (1998). Organizing Knowledge. California Management Review, 40(3).

- Brown, J. S., & Duguid, P. (1991). Organizational learning and communities-of-practice:
  Toward a unified view of working, learning, and innovation. *Organization science*, 2(1), 40-57.
- Bryman, A. (2012). Social research methods. Oxford: Oxford University Press.
- Burgess, R. G. (2002). In the field: An introduction to field research. London: Routledge.
- Carbonaro, M., Bainbridge, J., & Wolodko, B. (2002). Using Internet surveys to gather research data from teachers: Trials and tribulations. *Australian Journal of Educational Technology*, 18(3), 275-292.
- Carley-Baxter, L. R., Hill, C. A., Roe, D. J., Twiddy, S. E., Baxter, R. K., & Ruppenkamp, J.
  (2009). 'Does Response Rate Matter? Journal Editors Use of Survey Quality Measures in Manuscript Publication Decisions. *Survey Practice*, 2(7), 1-11.
- Carlile, P. R., & Rebentisch, E. S. (2003). Into the black box: The knowledge transformation cycle. *Management science*, 49(9), 1180-1195.
- Cheuk, B. (2007). Applying sense-making and narrative techniques to capture lessons learnt. Journal of Information & Knowledge Management, 6(03), 165-171.
- Church, A. (2018). Chapter 16: Critical Incident Technique. In Small, R. V., & In Mardis, M. A. (Ed.), *Research methods for librarians and educators: Practical applications in formal and informal learning environments* (177-185). Santa Barbara, CA: Libraries Unlimited.
- Choo, C. W. (1998). The knowing organization: How Organizations Use Information to Construct Meaning, Create Knowledge, and Make Decisions. New York: Oxford University Press.

- Christensen, P. H. (2003). Knowledge sharing time sensitiveness and push-pull strategies in a non-hype organization. *EGOS Colloquium*, Copenhagen Business School, Copenhagen.
- Christensen, P.H. (2007). Knowledge sharing: moving away from the obsession with best practices. *Journal of knowledge management*, 11(1), 36-47.
- Clark, D. (2004). *Understanding and performance*. Retrieved from http://www.nwlink.com/~donclark/performance/understanding.html
- Cohen, L., & Manion, L. (Eds.). (1994). Research methods in education. London: Routhledge.
- Cohen, W. M., & Levinthal, D. A. (1990). Absorptive capacity: A new perspective on learning and innovation. *Administrative science quarterly*, 128-152.
- Collins, H. M. (1993). The structure of knowledge. Social research, 60(1), 95-116.
- Collis, J., & Hussey, R. (2009). *Business research: A practical guide for undergraduate and postgraduate students*. Charlesworth Press, Wakefield: Palgrave Macmillan.
- Consoli, D. (2012). Literature analysis on determinant factors and the impact of ICT in SMEs. *Procedia-social and behavioral sciences*, 62, 93-97.
- Cook, C., F. Heath, and R.L. Thompson. (2000). A meta-analysis of response rates in web or internet-based surveys. *Educational and Psychological Measurement*, 60(6), 821–836.
- Corbin, J., & Strauss, A. (2008). *Basics of qualitative research: Techniques and procedures for developing grounded theory*. Thousand Oaks, CA: Sage.
- Couper, M. P. (2011). The future of modes of data collection. *Public Opinion Quarterly*, 75(5), 889-908.
- Creswell, J. W. (1998). *Qualitative Inquiry and Research Design: Choosing among Five Traditions*. Thousand Oaks: Sage

- Creswell, J. W. (2007). *Qualitative inquiry and research design: Choosing among five approaches*. Thousand Oaks, CA: Sage.
- Creswell, J. W. (2008). *Educational research: Planning, conducting, and evaluating quantitative and qualitative research*. Upper Saddle River, NJ: Pearson Education.
- Creswell, J. W. (2013). *Qualitative Inquiry & Research Design: Choosing among Five Approaches*. Thousand Oaks, CA: Sage.
- Creswell J. W. (2014). *Research design: Qualitative, quantitative and mixed methods approaches.* Thousand Oaks, CA: Sage.
- Creswell, J. W. (2015). *A concise introduction to mixed methods research*. Sage Publications: London.
- Creswell, J. W., & Plano-Clark, P. V.L. (2011). *Designing and Conducting Mixed Methods Research*. London: Sage.
- Creswell, J. W., & Plano Clark, V. L. (2007). *Designing and conducting mixed methods research*. Thousand Oaks, CA: Sage.
- Cronbach, L. J. (1951). Coefficient alpha and the internal structure of tests. *Psychometrika*, 16, 297-334.

Dalkir, K. (2011). Knowledge management in theory and practice. Cambridge, MA: MIT press.

- Dalrymple, P. W. (2001). A quarter of a century of user-centred study: the impact of Zweizig and Dervin on library and information science research. *Library and Information Science Research* 23(2), 155-165.
- Dattalo, P. (2008). *Determining sample size: Balancing power, precision, and practicality*. Oxford: Oxford University Press.

- Davenport, T. H. (1994). Saving IT's soul: Human-centered information management. Boston Mass: Harvard Business School Press, 72(2), 119-31.
- Davenport, T. H., & Prusak, L. (1998). *Working knowledge: How organizations manage what they know*. Boston Mass: Harvard Business School Press.
- Denscombe, M. (2007). *The good research guide: For small-scale social research projects*. Maidenhead: Open University Press.
- Dervin, B. (1983). An overview of sense-making research: Concepts, methods and results.Paper presented at the annual meeting of the International Communication Association.Dallas, TX.
- Dervin, B. (1992). From the mind's eye of the user: The sense-making qualitative-quantitative methodology. In J. D. Glazi & R. R. Powell (Ed.), *Qualitative research in information management* (pp. 61–84). Englewood, CO: Libraries Unlimited.
- Dervin, B. (2014). Chapter 4: Dervin's Sense-Making Theory. In M. N. Al-Suqri & A. S.
   AlAufi (Ed.), *Information seeking behavior and technology adoption: Theories and trends* (pp. 59–80). Hershey: Information Science Reference.
- Desouza, K. C., & Awazu, Y. (2006). Knowledge management at SMEs: five peculiarities. *Journal of knowledge management*, 10(1), 32-43.
- Dey, I. (1993). *Qualitative data analysis: A use-friendly guide for social scientists*. London: Routledge.
- Dignum, M. V., & Van Eijk, R. M. (2005). Towards a model to understand the influence of trust in knowledge sharing decisions. *In Workshop on Trust AAMAS'05*.
- Doan Q.M., Rosenthal-Sabroux C., Grundstein M. (2011), A Reference Model for Knowledge Retention within Small and Medium-sized Enterprises. *KMIS*, 306-311

- Dougherty, D. (1992). Interpretive barriers to successful product innovation in large firms. *Organization science*, 3(2), 179-202.
- Doyle, L., Brady, A. M., & Byrne, G. (2016). An overview of mixed methods researchrevisited. *Journal of research in nursing*, 21(8), 623-635.

Drucker, P. (2012). Managing in turbulent times. Hoboken: Taylor and Francis.

- Durst, S., & Runar Edvardsson, I. (2012). Knowledge management in SMEs: a literature review. *Journal of Knowledge Management*, 16(6), 879-903.
- Edmondson, A. C., & McManus, S. E. (2007). Methodological fit in management field research. *Academy of management review*, 32(4), 1246-1264.
- Economist Intelligence Unit & Cisco Systems. (2006). *Foresight 2020: Economic, industry and corporate trends*. London: Economist Intelligence Unit.
- Eligon, J. (2012, July 17). Plan to Close University of Missouri Press Stirs Anger. *The New York Times*. Retrieved from http://www.nytimes.com
- Elliott, R., Fischer, C. T., & Rennie, D. L. (1999). Evolving guidelines for publication of qualitative research studies in psychology and related fields. *British journal of clinical psychology*, 38(3), 215-229.
- Evangelista, P., Esposito, E., Lauro, V., & Raffa, M. (2010). The adoption of knowledge management systems in small firms. *Electronic Journal of Knowledge Management*, 8(1).
- Feilzer, M. Y. (2010). Doing mixed methods research pragmatically: Implications for the rediscovery of pragmatism as a research paradigm. *Journal of Mixed Methods Research*, 4(1), 6-16.

Fellows, R. & Liu, A. (1997). Research Methods for Construction. Oxford: Blackwell Science.

- Fetters, M. D., Curry, L. A., & Creswell, J. W. (2013). Achieving integration in mixed methods designs-principles and practices. *Health services research*, 48(6pt2), 2134-2156.
- Fife-Schaw, C. (2006). Levels of measurement. In G. Breakwell, S. M. Hammond, C. Fife-Schaw & J. A. Smith (Ed.), *Research methods in psychology* (pp. 50-63). Thousand Oaks, CA: Sage.
- Finkl, K. & Ploder, C. (2009). Knowledge Management Toolkit for SMEs. *International Journal of Knowledge Management*. 5(1), 46-60.
- Fischer, C.T., & Wertz, F.J. (2002). Empirical phenomenological analyses of being criminally victimized. In M. Huberman & M.B. Miles (Ed.), *The qualitative researcher's companion* (pp. 275-304). Thousand Oaks, CA: Sage.
- Fisher, K. E., Erdelez, S., & McKechnie, L. (Eds.). (2005). Theories of information behavior. Medford: Information Today.
- Fitzgerald, B., & Howcroft, D. (1998, December). Competing dichotomies in IS research and possible strategies for resolution. In *Proceedings of the international conference on Information systems* (pp. 155-164). Association for Information Systems.
- FitzGerald, K., Seale, N. S., Kerins, C. A., & McElvaney, R. (2008). The critical incident technique: a useful tool for conducting qualitative research. *Journal of Dental Education*, 72(3), 299-304.

Flanagan, J. C. (1954). The critical incident technique. *Psychological bulletin*, 51(4), 327.

- Foreman-Wernet, L. (2003). Rethinking communication: Introducing the sense-making methodology. *Sense-making methodology reader: Selected writings of Brenda Dervin*, 3-16.
- Fox, W. & Bayat, M.S. (2007). A Guide to Managing Research. Cape Town: JUTA and Co Ltd.

- Gastil, R. D. (1961). The determinants of human behavior. *American Anthropologist*, 63(6), 1281-1291.
- Giddens, A. (1979). Central problems in social theory: Action, structure, and contradiction in social analysis. Berkeley: University of California Press.
- Giju, G., Badea, L., Ruiz, V. & Peña, D. (2010). Knowledge Management the Key Resource in the Knowledge Economy. *Theoretical and Applied Economics*, 17(6): 27-36.
- Gill, J., & Johnson, P. (1997). Research methods for managers. London: P. Chapman.
- Giorgi, A. (2009). *The descriptive phenomenological method in psychology: A modified Husserlian approach*. Pittsburgh: Duquesne University Press.
- Girgin, U., Kurt, A. A., & Odabasi, F. (2011). Technology integration issues in a special education school in Turkey. *Cypriot Journal of Educational Sciences*, 6(1), 13-21.
- Gloet, M., & Berrell, M. (2003). The dual paradigm nature of knowledge management: implications for achieving quality outcomes in human resource management. *Journal of knowledge management*, 7(1), 78-89.
- Godden, B. (2004). *Sample Size Formulas*. Retrieved from http://williamgodden.com/samplesizeformula.pdf
- Grant, R. M. (1996). Toward a knowledge-based theory of the firm. *Strategic Management Journal*, 17(S2), 109-122.
- Greene, J. C., Caracelli, V. J., & Graham, W. F. (1989). Toward a conceptual framework for mixed method evaluation designs. Educational evaluation and policy analysis, 11(3), 255-274.
- Grinnell Jr, R. M., & Unrau, Y. A. (2010). Social work research and evaluation: Foundations of evidence-based practice. Oxford: Oxford University Press.

Grossman, M. (2007). The emerging academic discipline of knowledge management. *Journal of Information Systems Education*, 18(10), 31-38.

Guba, E. G. (Ed.). (1990). The paradigm dialog. London: Sage.

- Guest, G., Bunce, A., & Johnson, L. (2006). How many interviews are enough? An experiment with data saturation and variability. *Field methods*, *18*(1), 59-82.
- Guo, Z., & Sheffield, J. (2007). Critical heuristics: a contribution to addressing the vexed question of so-called knowledge management. *Systems Research and Behavioral Science*, 24(6), 613-626.
- Gupta, J. N., & Sharma, S. K. (Eds.). (2004). *Creating knowledge based organizations*. Hershey, PA: Idea Group.
- Hansen, M. T. (1999). The search-transfer problem: The role of weak ties in sharing knowledge across organization subunits. *Administrative science quarterly*, 44(1), 82-111.
- Harrison, R. L., & Reilly, T. M. (2011). Mixed methods designs in marketing research. *Qualitative Market Research: An International Journal*, 14(1), 7-26.
- Hayes, A. F. (2005). Statistical methods for communication science. New York, NY: Routledge
- Herrmann, N. (2011). Barriers for an efficient Management of Knowledge. Experiences from a Southern African Organization. *Open Journal of Knowledge Management*, 3, 1-15.
- Hettlage, R., & Steinlin, M. (2006). The critical incident technique in knowledge managementrelated contexts. *Zurich: IngeniousPeoplesKnowledge*. Retrieved from http://www.click4it.org/images/6/68/Critical IncidentTechnique in KM.pdf
- Hibbard, J. (1997, Oct 20). Knowledge Management: Knowing what we know. *Information Week Online*. Retrieved from http://www.informationweek.com

- Hislop, D. (2013). *Knowledge management in organizations: A critical introduction*. Oxford:Oxford University Press.
- Hitt, M.A., Ireland, R.D., & Hoskisson, R.E. (2014). Strategic management: Concepts: Competitiveness and globalization. South-Western Pub.
- Horibe, F. (2007). Are you making familiar mistakes with innovation? *Knowledge Management Review*, 9(6), 8-10.
- Huberman, M., & Miles, M. B. (2002). *The qualitative researcher's companion*. Thousand Oaks, CA: Sage.
- Hunt, S. D. (1994). On rethinking marketing: our discipline, our practice, our methods. *European Journal of Marketing*, 28(3), 13-25.
- Hussain, F., Lucas, C., & Ali, M. A. (2004). Managing knowledge effectively. Journal of Knowledge Management Practice. Retrieved from http://www.tlainc.com/articl66.htm
- Imenda, S. (2014, February 1). Is There a Conceptual Difference between Theoretical and Conceptual Frameworks? *Journal of Social Sciences*, *38*(2), 185-195.
- Jashapara, A. (2003). *Knowledge management and culture, knowledge management: An integrated approach*. London: Prentice-Hall.
- Jashapara, A. (2004). *Knowledge management: An integrated approach*. Harlow, England: Prentice Education Ltd.
- Jashapara, A. (2010). *Knowledge management: An integrated approach*. Harlow, England: Financial Times Prentice Hall.
- Johnson, R. B., Onwuegbuzie, A. J., & Turner, L. A. (2007). Toward a definition of mixed methods research. *Journal of mixed methods research*, 1(2), 112-133.

- Klug J., Stein W., Licht T. (2001), Knowledge Unplugged. *The McKinsey & Company Global Survey on Knowledge Management*. Palgrave, Houndmills: McKinsey & Company
- Krejcie, R. V., & Morgan, D. W. (1970). Determining sample size for research activities. *Educational and psychological measurement*, 30(3), 607-610.
- Kuhn, T.S. (1962). *The Structure of Scientific Revolutions*. Chicago: University of Chicago Press.
- Kulkarni, U., & St Louis, R. (2003). Organizational self-assessment of knowledge management maturity. AMCIS 2003 Proceedings, 332.
- Kuzel, A. J. (1992). Sampling in qualitative inquiry. In: B. F. Crabtree & W. L. Miller (Ed.), Doing qualitative research. Research Methods for Primary Care (pp. 31-44). Newbury Park, CA: Sage.
- Lang, J. R., Calantone, R. J., & Gudmundson, D. (1997). Small firm information seeking as a response to environmental threats and opportunities. *Journal of small business Management*, 35(1).
- Lesgold A. 1988. Problem solving. In Sternberg R. J. & Smith E.E. (Ed.), *The Psychology of Human Thought* (pp. 188-213). Cambridge, UK: Cambridge University Press.
- Liebowitz, J. (2008). *Knowledge retention: strategies and solutions*. Boca Raton, FL: CRC Press.
- Liew, A. (2013). DIKIW: Data, information, knowledge, intelligence, wisdom and their interrelationships. *Business Management Dynamics*, 2(10), 49-62.
- Lincoln, Y. S., & Guba, E. G. (1985). Naturalistic inquiry. Newbury Park: Sage.

Little, S., & Ray, T. (2005). Managing knowledge: an essential reader. London: Sage.

- Liyanage, C. L., Elhag, T., & Ballal, T. (2012). *Establishing A Connection Between Knowledge Transfer and Innovation Diffusion*. TLA (The Leadership Alliance).
- Liyanage, C., Elhag, T., Ballal, T., & Li, Q. (2009). Knowledge communication and translation – a knowledge transfer model. *Journal of Knowledge management*, 13(3), 118-131.
- McNamara, C. (2009). *General guidelines for conducting interviews*. Retrieved from http://managementhelp.org/evaluatn/intrview.htm
- Malhotra, Y. (2000). Knowledge management for e-business performance: advancing information strategy to "internet time". *Information Strategy: The Executive's Journal*, *16*(4), 5-16.
- Mangabeira, W. C., Lee, R. M., & Fielding, N. G. (2004). Computers and qualitative research: Adoption, use, and representation. *Social Science Computer Review*, *22*(2), 167-178.
- Mason, M. (October 7, 2010). Sample size and saturation in PhD studies using qualitative interviews. *Forum Qualitative Sozialforschung*, 11, 3.
- Maxwell, J. A., & Loomis, D. M. (2003). *Handbook of mixed methods in social & behavioral research*. London: Sage.
- Maxwell, J. A. (2009). Designing a qualitative study. In L. Bickman & D. J. Rog (Ed.), *The SAGE handbook of applied social research methods* (pp. 214-253). Thousand Oaks, CA: Sage.

McCracken, G. (1988). The long interview. Newbury Park, CA: Sage.

McEvily, S. K., & Chakravarthy, B. (2002). The persistence of knowledge-based advantage: an empirical test for product performance and technological knowledge. *Strategic management journal*, 23(4), 285-305.

- McLaughlin, S., Paton, R. A., & Macbeth, D. K. (2008). Barrier impact on organizational learning within complex organizations. *Journal of knowledge management*, 12(2), 107-123.
- Mertens, D.M. (2005). *Research methods in education and psychology: Integrating diversity with quantitative and qualitative approaches*. Thousand Oaks, CA: Sage.
- Miles, M. B., & Huberman, A. M. (1994). *Qualitative data analysis: An expanded sourcebook*. Thousand Oaks, CA: Sage.
- Miller, D., & Shamsie, J. (1996). The resource-based view of the firm in two environments: the Hollywood film studios from 1936 to 1965. *Academy of Management Journal*, 39: 519-543.
- Moore, G. C., & Benbasat, I. (1991). Development of an instrument to measure the perceptions of adopting an information technology innovation. *Information systems research*, 2(3), 192-222.
- Morrison, A. P., Wells, A., & Nothard, S. (2002). Cognitive and emotional predictors of predisposition to hallucinations in non-patients. *British Journal of Clinical Psychology*, 41, 259–270.
- Morse, J. M. (1995). The significance of saturation. *Qualitative Health Research*, 5, 147–149.
- Morse, J. M. (2003). Principles of mixed methods and multimethod research design. *Handbook* of mixed methods in social and behavioral research, 1, 189-208.

Moustakas, C. (1994). Phenomenological research methods. Thousand Oaks, CA: Sage.

Neuman, W. L. (2006). *Social research methods: qualitative and quantitative approaches*. Boston, Mass: Pearson.

- Newell, S., Sc- arbrough, H., Swan, J. & Hislop, D. (2000). Intranets and Knowledge
  Management: De-Centred Technologies and the Limits of Technological Discourse. In
  Prichard, C., Hull, R., Chumer, M., & Wilmott, H. (Ed.). *Managing Knowledge* (pp. 88-106). London: Macmillan.
- Nulty, D. D. (2008). The adequacy of response rates to online and paper surveys: what can be done? *Assessment & evaluation in higher education*, *33*(3), 301-314.
- Nonaka, I., & Yamanouchi, T. (1989). Managing innovation as a self-renewing process. Journal of Business Venturing, 4, 299-315.
- Nonaka, I. (1991). The Knowledge Creating company. *Harvard Business Review*, 69 (6), 96–104.
- Nonaka, I. (1994). A dynamic theory of organizational knowledge creation. *Organization Science*, 5(1), 14-37.
- Nonaka, I., & Takeuchi, H. (1995). *The knowledge-creating company: how Japanese companies create the dynamics of innovation*. New York: Oxford University Press.
- Nunes, MB., Annansingh, F., Eaglestone, B., & Wakefield, R. (2006). Knowledge Management Issues in Knowledge-intensive SMEs. *Journal of Documentation*, 62(1), 101-119.
- O'Cathain, A., Murphy, E., & Nicholl, J. (2010). Three techniques for integrating data in mixed methods studies. *BMJ*, 341.
- Onwuegbuzie, A. J., & Leech, N. L. (2007). A call for qualitative power analyses. *Quality & Quantity*, *41*(1), 105-121.
- Omerzel, D., & Antoncic, B. (2008). Critical entrepreneur knowledge dimensions for the SME performance. *Industrial Management & Data Systems*, *108*(9), 1182-1199.

Patton, M. Q. (1990). Qualitative evaluation and research methods. London: Sage.

Patton, M.Q. (2002). Qualitative research and evaluation methods. Thousand Oaks, CA: Sage.

- Paulin, D., & Suneson, K. (2012). Knowledge transfer, knowledge sharing and knowledge barriers-three blurry terms in KM. *The Electronic Journal of Knowledge Management*, 10(1), 81-91.
- Peugh, J.L., & Enders, C. K. (2004). Missing data in educational research: A review of reporting practices and suggestions for improvement. *Review of Educational Research*, 74, 525-556.
- Polanyi, M. (1962). Tacit knowing: Its bearing on some problems of philosophy. *Reviews of modern physics*, 34(4), 601.
- Polanyi, M. (1966). The tacit dimension. New York, NY: Doubleday.
- Porter, S. R., Whitcomb, M. E., & Weitzer, W. H. (2004). Multiple surveys of students and survey fatigue. *New Directions for Institutional Research*, 2004(121), 63-73.
- Porter, S. R., & Whitcomb, M. E. (2005). Nonresponse in student surveys: The role of demographic engagement and personality. *Research in Higher Education*, 46 (2), 127-152.
- Radford, M. L. (1999). *The reference encounter: Interpersonal communication in the academic library*. Chicago: ACRL, A Division of the American Library Association.
- Rigoni, B. & Adkins, A. (2015, Jan 28), As Baby Boomers Retire, It's Time to Replenish Talent. *Business Journal: Gallup*. Retrieved from http://www.gallup.com/businessjournal/181295/baby-boomers-retire-time-replenishtalent.aspx
- Rhodes, C., & Brown, A. D. (2005). Narrative, organizations and research. *International Journal of Management Reviews*, 7(3), 167-188.

Rogers, E. M. (1995). Diffusion of Innovations. New York: The Free Press.

- Rossman, G. B., & Wilson, B. L. (1985). Numbers and words: Combining quantitative and qualitative methods in a single large-scale evaluation study. *Evaluation Review*, 9(5), 627-643.
- Rowley, J. (2007). The wisdom hierarchy: representations of the DIKW hierarchy. *Journal of Information Science*, 33(2), 163-180.
- Rutten, W., Blaas-Franken, J., & Martin, H. (2016). The impact of (low) trust on knowledge sharing. *Journal of knowledge management*, *20*(2), 199-214.
- Ryu, S., Ho, S. H., & Han, I. (2003). Knowledge Sharing Behavior of Physicians in Hospitals. *Expert Systems with Applications*, 25, 113-122.

Saldaña, J. (2015). The coding manual for qualitative researchers. Thousand Oaks, CA: Sage.

- Saunders, M., Lewis, P., & Thornhill, A. (2009). Research Onion. Research methods for Business Students, 136-162.
- Savolainen, R. (1993). The sense-making theory: Reviewing the interests of a user-centered approach to information seeking and use. *Information processing & management*, 29(1), 13-28.
- Schonfeld, R. (2016). Book publishing: University presses adapt. Nature, 540(7631), 35-37.
- Schonlau, M., Fricker, R.D., & Elliott, M. N. (2002). *Conducting research surveys via e-mail and the web*. Santa Monica: CA: Rand.
- Schwartz, S. H. (2006). A theory of cultural value orientations: Explication and applications. *Comparative sociology*, 5(2), 137-182.
- Senge, P. M. (1997). The fifth discipline. *Measuring Business Excellence*, 1(3), 46-51.

Singleton, R., & Straits, B. C. (2010). Approaches to social research. New York:

Oxford University Press.

- Silver, C., & Lewins, A. (2014). Using software in qualitative research: A step-by-step guide. Los Angeles: Sage.
- Silverman, D., & Marvasti, A. B. (2008). *Doing qualitative research: A comprehensive guide*. London: Sage.
- Skyrme, D. J. (2000). *Innovation: Knowledge connections*. Retrieved from: https://www.skyrme.com/insights/12knoinn.htm
- Small, C. T., & Sage, A. P. (2005). Knowledge management and knowledge sharing: A review. Information Knowledge Systems Management, 5(3), 153-169.
- Smith, G. (2008). Does gender influence online survey participation? A record-linkage analysis of university faculty online survey response behavior. *ERIC Document Reproduction Service No. ED 501717.*
- Smith J. A., Flowers P., Larkin M. (2009). *Interpretative phenomenological analysis: Theory, method and research*. London: Sage.
- Sparrow, J. (2001). Knowledge management in small firms. *Knowledge and process management*, 8(1), 3-16.
- Straumsheim, C. (2016, Aug 1). Pressing Challenges. *Insider Higher Ed*. Retrieved from http://www.insidehighered.com

Strauss, A., & Corbin, J. (1998). Basics of qualitative research. Thousand Oaks, CA: Sage.

- Stuckey, H. L. (2013). Three types of interviews: Qualitative research methods in social health. *Journal of Social Health and Diabetes*, 1(2), 56.
- Suresh, D. (2015, January 1). Important of SPSS for Social Sciences Research. SSN Electronic Journal.

- Sveiby, K. E. (2007). Disabling the context for knowledge work: the role of managers' behaviours. *Management Decision*, 45(10), 1636-1655.
- Szulanski, G. (1996). Exploring internal stickiness: Impediments to the transfer of best practice within the firm. *Strategic management journal*, 17(S2), 27-43.
- Thayer-Hart, N., Dykema, J., Elver, K., Schaeffer, N. C., & Stevenson, J. (2010). *Survey fundamentals: A guide to designing and implementing surveys.* Madison, WI: University of Wisconsin.
- Thomas, J. B., Sussman, S. W., & Henderson, J. C. (2001). Understanding "strategic learning": Linking organizational learning, knowledge management, and sensemaking. *Organization science*, 12(3), 331-345.
- Valkokari, K., & Helander, N. (2007). Knowledge management in different types of strategic SME networks. *Management Research News*, 30(8), 597-608.
- Van Teijlingen, E. R., & Hundley, V. (2001). The importance of pilot studies. *Social Research Update*, (35).
- Walczak, S. (2008). Knowledge management and organizational learning: An international research perspective. *The Learning Organization*, 15(6), 486-494.
- Watkins, D. C., & Gioia, D. (2015). *Mixed methods research. Pocket guides to social work research methods series*. New York, NY: Oxford University Press.
- Weick, K. E. (1979). Cognitive processes in organizations. *Research in organizational behavior*, 1(1), 41-74.
- Weick, K. E. (1995). Sensemaking in organizations. Thousand Oaks, CA: Sage.
- Wiig, K. M. (1993). *Knowledge Management Foundations: thinking about-how people and organizations create, represent, and use knowledge*. Arlington, Texas: Schema.

- Wiig, K. M. (1997). Knowledge management: Where did it come from and where will it go? *Expert systems with applications*, 13(1), 1-14.
- Wiklund, J. (1999). The sustainability of the entrepreneurial orientation -- performance relationship. *Entrepreneurship Theory and Practice* (Fall), 37-48.
- Wiklund, J., & Shepherd, D. (2003). Knowledge-based resources, entrepreneurial orientation, and the performance of small and medium-sized businesses. *Strategic management journal*, 24(13), 1307-1314.
- Williams, R. H., & Zimmerman, D. W. (1989). Statistical power analysis and reliability of measurement. *The Journal of general psychology*, *116*(4), 359-369.
- Wong, K.W. (2005). Critical success factors for implementing knowledge management in small and medium enterprises. *Industrial Management & Data Systems*, 105(3), 261-279.
- Wong, K. Y., & Aspinwall, E. (2004). Characterizing knowledge management in the small business environment. *Journal of Knowledge management*, 8(3), 44-61.
- Yao, L. J., Kam, T. H. Y., & Chan, S. H. (2007). Knowledge sharing in Asian public administration sector: the case of Hong Kong. *Journal of Enterprise Information Management*, 20(1), 51-69.
- Yao, L., Othman, A., Abdalla, A. N., & Jing, W. (2011). A novel sensemaking model of effective knowledge management within SMEs. *African Journal of Business Management*, 5(11), 4423.
- Yuan, Y. H., Wu, M. H., & Lee, J. C. (2012). Examining the role of knowledge transfer effect as a mediator variable among impact factors in knowledge innovation. *International Journal of Business and Information*, 7(2), 205.

- Zach, L. (2005). When is "enough" enough? Modeling the information-seeking and stopping behavior of senior arts administrators. *Journal of the Association for Information Science and Technology*, 56(1), 23-35.
- Zander, U., & Kogut, B. (1995). Knowledge and the speed of the transfer and imitation of organizational capabilities: An empirical test. *Organization Science*, 6(1), 76-92.
- Zikmund, W. G., Babin, B. J., Carr, J. C., & Griffin, M. (2008). *Business research methods*. Mason (Ohio): South-Western Cengage Learning.

# APPENDICES

# Appendix A – Consent Forms (Quantitative & Qualitative Portions)

# QUAN

# Q (1) Informed Consent Statement

### Introduction

You are invited to complete this survey, approved by University of Tennessee Institutional Review Board and administrated by an Information Science doctoral student at University of Tennessee, because of your affiliation with the Association of University Presses (AUP).

The purpose of this survey is to collect online sample information about your knowledge practices such as how you seek knowledge in handling your daily work, your experience in sharing knowledge with your co-workers and how you retain that knowledge in your organization.

## **Benefits**

Your responses to this survey will be used in two ways:

- (a) Your responses will be helpful in identifying the important issues in Knowledge Management.
- (b) Results of this research will be useful in the field of Information Science studies, particularly to those at institutions seeking to better understand perceptions of knowledge management practices.

While participants who complete the survey will receive no immediate benefit, their involvement and feedback will help inform new initiatives for knowledge management studies.

### **Participation**

Approximately 1,200 members on the AUP listserv received an invitation to participate in this research by completing the online survey.

Your participation in this research will involve only the completion of the following selfadministrated online survey, which should take no more than 20 minutes to finish. Please complete it in a private setting. Participation in this research is entirely voluntary. Clicking on the "I agree to participate" button (at the bottom of this page) indicates you are over 18 years old and consent to participate. If you click on "I do not want to participate" button (at the bottom of this page), the system will end this survey.

### **Confidentiality**

The online survey is completely anonymous. Should you choose to participate, no one is able to identify your response. Aggregated survey data, after cleaned and anonymized, may be put in an

open repository.

The researcher of this study has taken great care to ensure your protection by: (a) excluding questions that could potentially identify the participant, and (b) providing the option to skip any item in the survey.

### **Compensation**

You may exit the online survey at any time without penalty. However, once you submit your completed survey, your online data may not be withdrawn as the survey is anonymous and there will be no way to locate your responses within the data set. No payment or other compensation will be given to participants for their involvement in this research.

### <u>Risks</u>

There are no foreseeable risks to you, other than those you encounter in everyday life, if you complete this survey, as it contains no items that ask about sensitive or personal information.

## **Contact Information**

If you would like to obtain more information about this study, please feel free to contact the researcher via email at judyli@utk.edu. If you would like more information about your rights as a research participant or have questions about university policies and procedures for research involving human subjects, please contact the Compliance Officer and IRB Administrator for the University of Tennessee Knoxville, telephone 865-974-7697.

Thank you for your time,

Judy Li (Doctoral Student at University of Tennessee)

Choose from

I have read the above information. I agree to participate in this study.

I do not want to participate

### QUAL

Qualitative Research - Informed Consent Statement

Informed Consent Statement Research Project Title: Knowledge Sharing Matters

#### INTRODUCTION

You are invited to take part in a research interview, conducted by Judy Li, a doctoral student, at the University of Tennessee. The purpose of the research is to understand how knowledge is being used and shared in your company.

#### INFORMATION ABOUT PARTICIPANTS' INVOLVEMENT IN THE STUDY

The interview will last about 30 minutes and it will be audio recorded. You don't need to answer any questions if you do not want to and you can stop the interview at any time. Information from this interview will remain confidential. To protect your privacy, you will be referred with a code name in the recording. The audio recording will be erased after the transcription and the research project are complete. You must be 18 years of age or older to participate.

#### **RISKS AND PROTECTIONS**

The risk presented is not greater than what a person might reasonably expect to encounter in everyday life. However, the risk of a breach of confidentiality is always possible. Steps have been taken to minimize this risk.

#### BENEFITS

Your participation in the research will help the researcher identify the use and importance of knowledge in organizations that leads to a contribution in the research of Knowledge Management.

#### CONFIDENTIALITY

The information gathered in this study will be kept confidential. The interview is recorded and accessed only by the researcher conducting the study and a hired transcriber agency. No reference will be made in written or oral reports that could link participants to the study.

#### COMPENSATION

IRB NUMBER: UTK IRB-17-04189-XP IRB APPROVAL DATE: 01/23/2018 IRB EXPIRATION DATE: 01/22/2019 There is no known compensation of any kind.

#### CONTACT INFORMATION

If you have questions at any time about the study or the procedures, you may contact the researcher, Judy Li, at judyli@utk.edu, and office number 1-865-974-0013 or the advisor Dr. Suzanne Allard at <u>sallard@utk.edu</u>. If you have questions about your rights as a participant, you may contact the University of Tennessee IRB Compliance Officer at utkirb@utk.edu or (865) 974-7697.

#### PARTICIPATION

Your participation in this interview is voluntary; you may decline to participate without penalty. If you decide to participate, you may withdraw from the interview at any time without penalty and without loss of benefits to which you are otherwise entitled. If you withdraw from the interview before it is completed, your recording will be destroyed. This research has been reviewed and approved according to the University of Tennessee IRB procedures for research involving human subjects.

#### FUTURE DATA USES

Your research information may be used for future research studies or shared with other researchers for use in future research studies without obtaining additional informed consent from you. If this happens, all your identifiable information will be removed before any future use or distribution to other researchers.

#### CONSENT

Please sign your choice below to indicate you have read the above information. You are 18 years of age or older. You agree to participate in this study at the location of your choice.

I have read the above information. I have received a copy of this form. I agree to participate in this interview and approve of using my choice of the location to conduct this interview.

Participant's Name (printed)

Participant's Signature

IRB NUMBER: UTK IRB-17-04189-XP IRB APPROVAL DATE: 01/23/2018 IRB EXPIRATION DATE: 01/22/2019

### Appendix B – Interview Guide (Qualitative Research)

Qualitative Research – The Interview Guide (as a guideline only)

### Introduction

Thank You for meeting with me today. I am currently researching knowledge practices on University Presses. Specifically, I am interested in exploring how University Presses share knowledge internally and retain knowledge when there is staff turnover. I am also interested in any past critical incidents on knowledge practices in your organizations and your feelings related to this topic.

### Informed Consent form

To make you feel comfortable, I would like to present you the consent form. I'll answer any questions you have about what this study involves. I consider you an expert on this topic and would appreciate your ideas, perceptions and opinions.

- This form explains how we are doing our research. The purpose of this form is to help you decide whether you want to be interviewed or not.
- The interview will last about 30 minutes and it will be audio recorded. You don't need to answer any questions if you do not want to and you can stop the interview at any time.
- We will be using the information from the interview to help us create a better understanding about knowledge practices.
- Information from this interview will remain confidential.

### Section A: General information on access to and use of information and knowledge, resources

### *Warm-up questions*

- A1. Tell me a little about your company and your job?
- A2. How do you keep yourself updated on issues related to performing your daily work?
- A3. Specifically, give me a past incident of how did you do when you were preparing for a project (i.e. whom you consulted with and the process you obtained information)?
- A4. When you couldn't find information in those resources, where did you go to find it? Who would you contact? OR Who did you contact? In that order? What is the role of (X in) when preparing the project? Do you get enough information for (x) planning purposes? Would you contact colleagues from other offices for information on (x)?
- A5. Were the resources and information adequate in the preparation of projects? What else might have helped you and the project?
- A6. Based on the experience you are sharing with us, do you feel like online means of communicating and sharing knowledge will be a priority for you in the future? How did you capture that sharing? What tools did you use?
*If Yes: In what specific ways do you see yourself using these? Do you see any limitations?* 

# Section B: Insights from experience – lessons learned, best practices, conclusions and recommendations for consideration by others

An important part of knowledge management is collecting experience in the form of lessons learned and best practices. The next set of questions relates to this aspect.

- B1. Based on your past experience you mentioned in A3, what "lessons learned" and "best practices identified" relating to "knowledge management in work processes" can you share with me?
- B2. In the sense of Intellectual Capital, what have been the challenges you and the project struggled with most?
- B3. What are the top pieces of advice you'd give to a colleague on a "DOS" and "DON'Ts" list in job knowledge sharing? Would you also explain why they are important?
- B4. At what point during projects do you think it would be most useful to capture lessons learned and good or best? Practices? Who do you think would be the best person to do this?

# **Open-ended** questions

- I know there are a lot of challenges in finding ways to perform job tasks and some challenges are rather personal. Is it OK if I ask you some of your personal feelings and experience in encountering issues? Could you also talk about those issues?
- How did you choose what solutions to apply?
- What principles guided your actions?
- Did you consult anyone in your organization when the incident happened?
- That is an interesting event. Could you tell me more? What would be your next step if the solution you used turns out to be a dead end?

# Section C: Comments and Feedback

- C1. Do you have any final comments or remarks having gone through this interview?
- C2. Are there any questions you think we should add, or changes you think will improve the interview from your end or ours?

### Probes

Remember to constantly probe for details using non-verbal active listing cues as well as words like "tell me more about that", "what did that mean to you", "how did you feel at that moment', "can you elaborate more" and "please go on".

### Section D: Closing

### Closing

Thank you for sharing your time and insights today. Is there any other information that you think I should know about? Do you have any questions about this interview, or what we talked about?

# Key message 1: You've been great. Thanks so much for sharing your insights and your experiences.

### Key message 2: Please don't hesitate to send us additional information or remarks by email, if anything comes to mind later on.

### Wrap-up

Do you have anything else you want to share with us at this time?

May we contact you in the future if we have other follow-up questions?

I appreciate you sharing your valuable time and your insights with me. I know I learned a lot from our conversation. I'll provide you with the transcript of our conversation if you would like. If any other thoughts come to mind about the topic we discussed, please feel free to contact me.

# Appendix C – Recruiting Email (For Qualitative Research)

Dear XXX,

## Ref: Project name - Knowledge Sharing Matters

Hope this email finds you well.

I am a doctoral student at the University of Tennessee currently undertaking a study on Knowledge Sharing in University Presses under the supervision of the Dissertation Committee. Output from the study is intended to benefit organizations like yours in the areas of quality improvements through the exploitation and optimization of knowledge management.

I would like to invite you to participate in an open-ended interview during the AUP 2018 annual meeting in San Francisco, relating to the above research based on your expertise and years of experience in the Presses. The purpose of the interview is to understand knowledge practices in your organization and to investigate knowledge barriers on the knowledge management issues. The interview will last for about 30 minutes. Please note that your responses will be treated as highly confidential and transcripts will not contain reference to any persons (including yourself) or organizations.

Should you be willing to participate, please email me your available dates and time during the week of June 17 - 21 in 2018 and the choice of your location. If you prefer an online Zoom interview (during or out of this period), please contact me about your available dates and times.

The summary of results will be available at the conclusion of the project. Should you wish to obtain a copy of this, please let me know. Thank you very much for your consideration of this invitation. Your participation is highly valued as it will contribute to the understanding of knowledge practices that leads to good strategic planning within the University Presses.

I look forward to hearing from you.

Your Sincerely,

Judy Li The Project Researcher University of Tennessee 865-974-0013 judyli@utk.edu

### Appendix D – Survey (Quantitative Research)

#### 9/16/2018

Qualtrics Survey Software

#### **Default Question Block**

#### Informed Consent Statement

#### Introduction

You are invited to complete this survey, approved by University of Tennessee Institutional Review Board and administrated by an Information Science doctoral student at University of Tennessee, because of your affiliation with the Association of University Presses (AUPresses).

The purpose of this survey is to collect online sample information about your knowledge practices such as how you seek knowledge in handling your daily work, your experience in sharing knowledge with your co-workers and how you retain that knowledge in your organization.

#### **Benefits**

Your responses to this survey will be used in two ways:

(a) Your responses will be helpful in identifying the important issues in Knowledge Management.

(b) Results of this research will be useful in the field of Information Science studies, particularly to those at institutions seeking to better understand perceptions of knowledge management practices.

While participants who complete the survey will receive no immediate benefit, their involvement and feedback will help inform new initiatives for knowledge management studies.

#### **Participation**

1,200 members on the AAUP-L listserv received an invitation to participate in this research by completing the online survey.

#### Qualtrics Survey Software

Your participation in this research will involve only the completion of the following selfadministrated online survey, which should take no more than 20 minutes to finish. Please complete it in a private setting. Participation in this research is entirely voluntary. Clicking on the "I agree to participate" button (at the bottom of this page) indicates you are over 18 years old and consent to participate. If you click on "I do not want to participate" button (at the bottom of this page), the system will end this survey.

#### **Confidentiality**

The online survey is completely anonymous. Should you choose to participate, no one is able to identify your response. Aggregated survey data, after cleaned and anonymized, may be put in an open repository.

The researcher of this study has taken great care to ensure your protection by: (a) excluding questions that could potentially identify the participant, and (b) providing the option to skip any item in the survey.

#### **Compensation**

You may exit the online survey at any time without penalty. However, once you submit your completed survey, your online data may not be withdrawn as the survey is anonymous and there will be no way to locate your responses within the data set. No payment or other compensation will be given to participants for their involvement in this research.

#### <u>Risks</u>

There are no foreseeable risks to you, other than those you encounter in everyday life, if you complete this survey, as it contains no items that ask about sensitive or personal information.

#### **Contact Information**

If you would like to obtain more information about this study, please feel free to contact the researcher via email at judyli@utk.edu. If you would like more information about your rights as a research participant or have questions about university policies and procedures for research involving human subjects, please contact the Compliance Officer and IRB Administrator for the University of Tennessee Knoxville, telephone 865-974-7697.

https://utk.col.qualtrics.com/ControlPanel/Ajax.php?action=GetSurveyPrintPreview

9/16/2018

9/16/2018

Qualtrics Survey Software

Thank you for your time,

Judy Li (Doctoral Student at University of Tennessee)

- O I have read the above information. I agree to participate in this study.
- O I do not want to participate

Demographics Section -

### Region of your organization

- O North America
- O South America
- O Africa
- O Europe
- O Asia
- O Others

#### Your Gender is

- O Male
- O Female
- O Prefer-not-to-specify

#### Your age is

- O Above 18 and younger than 30
- O 31 39
- 0 40 49
- O 50 and above

#### Qualtrics Survey Software

9/16/2018

How long have you been working in your organization? What are the three main responsibilities of your job position? (write-in)

How many staff do you supervise?

- O less than 10
- O 10 to 50
- O 51 to 100
- O more than 100
- O I do not supervise

Knowledge Practices Section -

How did you acquire the skills and expertise that you use in your job? (ref:B)

- O from supervisors or co-workers in this organization
- O through self-learning
- O through formal training
- O at my last job
- O elsewhere

How often do you need to ask information from your co-workers to perform your daily job? (ref: B)

- O Never
- O Occasionally
- O Once a week
- O Once a month
- O Daily

Where would you look for knowledge that you need to do your work? (ref: B)

O In paper-based documents

https://utk.col.qualtrics.com/ControlPanel/Ajax.php?action=GetSurveyPrintPreview

4/10

9/16/2018

Qualtrics Survey Software

- O In our team/department's members' mind
- O In our central information system
- O On my personal or workstation computer/hard drive
- O Other (write-in)

What are the prohibitions in instances that you were unable to share or receive knowledge and/or best practices from your team members? (ref: B)



- O No right tools available
- O Other (write-in)

I depend on documented procedures to do my work when I have questions in fulfilling my job tasks. (ref: B)

0	Strongly	agree
---	----------	-------

- O Agree
- O Neither agree nor disagree
- O Disagree
- O Strongly disagree

In my organization, I share my knowledge with co-workers because (ref: C)

	Strongly Disagree	Disagree	Neither Agree nor Disagree	Agree	Strongly Agree
l am an important part of my organization network	0	0	0	0	0
I have many connections in my organization to share knowledge	0	0	0	0	0

https://utk.col.qualtrics.com/ControlPanel/Ajax.php?action=GetSurveyPrintPreview

5/10

9/16/2018	Qualtrics Survey Software				
	Strongly Disagree	Disagree	Neither Agree nor Disagree	Agree	Strongly Agree
l am happy to share my knowledge at work in order to improve my organization daily operations	0	0	0	0	0
Knowledge sharing is important	0	0	0	0	0
l want my superior to think I am a good employee	0	0	0	0	0

My organization management encourages informal activities and opportunities to communicate/share experiences and work knowledge. (ref: C)



O Agree

- O Do not know
- O Disagree
- O Strongly Disagree

There are formal activities such as training sessions, forums and meetings in the organization to share knowledge and experience. (ref: C)

#### O Strongly Agree

- O Agree
- O Do not know
- O Disagree
- O Strongly Disagree

# In my organization, (ref: C)

	Strongly Agree	Agree	Neither agree nor disagree	Disagree	Strongly disagree
Lesson learned from failed projects are considered valuable.	0	0	0	0	0

6/10

9/16/2018	Qualtrics Survey Software				
	Strongly Agree	Agree	Neither agree nor disagree	Disagree	Strongly disagree
Lesson learned from successful projects are considered valuable.	0	0	0	0	0

There is a willingness in my organization that - (ref: C)

	Extremely likely	Moderately likely	Neither likely nor unlikely	Moderately unlikely	Extremely unlikely
Co-workers routinely share work experience with each other.	0	0	0	0	0
Activities associated with lessons learned are recognized publicly and/or rewarded by the management	0	0	0	0	0

Do you agree that staff, in your organization, do not have social activities after work hours. (ref:  $S) \end{tabular}$ 

- O Strongly Agree
- O Agree
- O Do not know
- O Disagree
- O Strongly Disagree

Do you agree that staff, in your organization, are eager to share work experience in the hallways or in casual gatherings. (ref: S)



- O Agree
- O Do not know
- O Disagree
- O Strongly Disagree

#### Qualtrics Survey Software

9/16/2018

From your view, what are the challenges in sharing knowledge with people from other sections of your company. (ref: S)

- O Don't perceive there is an urgent need
- O Lack of open-minded sharing environment
- O Lack of trust of others' knowledge
- O No proper technological platform to share
- O Do not know there is a knowledge needs

How motivated are you to share your knowledge and experience? (ref: S)

- O Very unmotivated because I keep my job secret to protect my job
- O Unmotivated because I do not care what is going on in the company
- O Motivated on average because I want to help my co-workers
- O Motivated because I think my co-workers need me
- O Very motivated because I think sharing mine will build a better company

Do you agree or disagree the following statements? (ref: S)

	Strongly Disagree	Disagree	Neither agree nor disagree	Agree	Strongly agree
I feel trustworthy with my co-workers' information	0	0	0	0	0
If I encounter problems in finishing my job, I know my co-workers will help me out when I ask	0	0	0	0	0
My co-workers are willing to help and do not deceive for their own profit	0	0	0	0	0

In your organization, documenting of work knowledge is a required part of your work practices. (ref: T)

O Yes, it is a must

O No, it is not

#### Qualtrics Survey Software

9/16/2018

Mechanisms and tools to share work knowledge are widely available in my organization. (ref: T)

O Yes, it is widely available

O No, it is not

Do you use any of the following tools or methods to share or store knowledge, experiences or best practices within your company? (ref: T)

0	Colleague(s)	(mentor/buddy/	/mentee
$\sim$	o onouguo(o)	(internet) buddig,	111011000

- O Team meetings
- O Collaborative platform such as Google Doc, OneDrive, Drop Box
- O Social Media such as Facebook, Twitter
- O Other such as (write-in)

The biggest barrier for you to store information you received more efficiently and effectively is (ref: T)

- O Too busy, lack of time
- O Poor technology tools
- O Organization policy
- O Poor information system/process
- O I do not consider it is important

From a knowledge sharing point of views, what changes would you like to see in your company? (write-in) (ref: T)

Powered by Qualtrics

# Appendix E – Variables (Quantitative Research)

Variable Name	Description	Data Type	Messurement	Survey Question
V1: Consent	Respondent's university press total staff number	Numeric	Scale	Jul vey Question
V1. Consent	Demographics	rumene	beare	1
V2: Region	Respondent's geographical location	Numeric	Scale	2
V3: Gender	Respondent's sex	Numeric	Scale	3
V4: Age	Respondent's age	Numeric	Scale	4
V-1. Ngo	Number of years Respondent with the university	rumene	Seale	
V5(a):LengthOrg	prace	String	Nominal	5(a)
V5(h):VearsExp	Respondent's age	Numeric	Ordinal	5(a)
V5(c): IobNature	Respondent's job nature	Numeric	Ordinal	5(0)
V6:Supervise	the number of staff Respondent suprervises	Numeric	Scale	5(0)
v 0.3uper vise	the number of start - Respondent suprervises	INUMERIC	Scale	0
	Knowledge Pactices - Behav	ior Section		
V7:SkillSource	Where to seek knowledge	Numeric	Scale	7
V8:InquiryFrequency	Frequency to seek knowledge	Numeric	Scale	8
V9(a):RefSource	Source of knowledge	Numeric	Scale	9
V9(b):RefSource_Txt	Write-In	String	Nominal	9
Q9(c)_Code	Coding for Question 9 Write-In	Numeric	Ordinal	9
V10 (a):KnowlegeProhibition	Knowledge Barriers	Numeric	Scale	10
V10(b):KnowProhText	Knowledge Barriers - Write-In	String	Nominal	10
V10(c):Q10_Code	Coding for Question 10 Write-In	Numeric	Ordinal	10
V11: UseDocumentation	Depend on documented procedures	Numeric	Scale	11
	Knowledge Pactices - Cultu	re Section		
V12(a):Culture1	I am an important part of my organization network	Numeric	Scale	12
V12(b):Culture2	share knowledge	Numeric	Scale	12
V12(c):Culture3	improve my organization daily operations	Numeric	Scale	12
V12(d):Culture4	Knowledge Sharing is important	Numeric	Scale	12
V12(u):Culture5	I want my superior to think I am a good employee	Numeric	Scale	12
v12(e):Cultures	My anomigation anonymous informal activities to	Numeric	Scale	12
V12:Culture6	share knowledge	Numeric	Scale	12
V13:Cultuleo	Earmal trainings and meetings available to share	Numeric	Scale	15
V14:Culture7	romai trainings and meetings available to share	Numaria	Saala	14
v14.Culture/	L assans L assend from failure projects are	Numeric	Scale	14
V15(a))Culture	considered veriable	Numaria	Saala	15
v15(a).Cultures	L assons L aarnad from failura projects are	Numeric	Scale	15
V15(b):Culture0	considered variable	Numeric	Scale	15
v15(0).Cultures	Co. workers routingly share work experience with	Numeric	Scale	15
V16(a))Culture10	co-workers routinery share work experience with	Numaria	Saala	15
v lo(a).Culture10	A stivities associated with lessons learned are	Numeric	Scale	15
V16(b):Cultural1	recognized publicly and/ or rewarded	Numeric	Scale	15
v10(0):Culture11	recognized publicity and/ of rewarded	ivumene	Scale	15
	Knowledge Pactices - Socia	al Section		
V17:Social1	No social activities after work in my organizations	Numeric	Scale	17
	Staff are eager to share work experience in casual			
V18:Social2	gatherings	Numeric	Scale	18
	Challenges in sharing kowledge with people from			
V19:SharingChallenge	other sections of the organization	Numeric	Scale	19
V20:Motivation	Motivated to share knowledge	Numeric	Scale	20
V21(a):Social3	I feel trustworthy with my co-workers' information	Numeric	Scale	21
	Co-workers will help me out if I encounter			
V21(b):Social4	problems at work	Numeric	Scale	21
	Co-workers are willing to help and do not deceive			
V21(c):Social5	for their own profit	Numeric	Scale	21
	Knowledge Pactices - Techno	logy Section		
	Documentation of work knowledge is a required			
V22:RequiredDocumentation	part of my work practices	Numeric	Scale	22
	The availability of knowledge curation mechanisms			
V23:TechToolsAvailability	and tools in my organization	Numeric	Scale	23
V24(a):TechTools	Methods or Tools to curate knowledge	Numeric	Scale	24
V24(b):TechTools_Txt	Write-In	String	Nominal	24
V24(c):Q24_code	Coding for Question 24 Write-In	Numeric	Ordinal	24
V25:StoringBarrier	Kowledge Curation Barrier	Numeric	Scale	25
V26:Changes_Txt	Write-In: Changes would like to happen	String	Nominal	26
Q26_code	Coding for Question 26 Write-In	Numeric	Ordinal	26
	Non-survey Variabl	es		
TrustAvg	Average of Sub-scale	Numeric	Scale	
CultureAvg	Average of Sub-scale	Numeric	Scale	

Variables

# Appendix F – Statistical Result Charts (Quantitative Research)

# Gender

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Male	17	16.0	16.0	16.0
	Female	88	83.0	83.0	99.1
	Prefer-not-to-specify	1	.9	.9	100.0
	Total	106	100.0	100.0	

		Gen	Gender	
		Male	Female	Total
How motivated are you to share your knowledge and experience? (ref: S)	Very unmotivated because I keep my job secret to protect my job	0	1	1
	Motivated on average because I want to help my co-workers	3	23	26
	Motivated because I think my co-workers need me	3	5	8
	Very motivated because I think sharing mine will build a better company	8	49	57
Total		14	78	92

		Gender		
		Male	Female	Total
The biggest barrier for your being able to store information you received more efficiently and ef	Too busy, lack of time	9	38	47
	Poor technology tools	1	5	6
	Organization policy	0	5	5
	Poor information system/process	1	21	22
Total		11	69	80

		Gen	der	
		Male	Female	Total
Mechanisms and tools to share work knowledge are widely available in my	Yes, it is widely available	5	33	38
available in my organization. (ref: T)	No, it is not	8	44	52
Total		13	77	90

		Gen	der			
		Male	Female	Total		
RefSourceFinal	In paper-based documents	1	2	3		
	In our team/department's members' mind In our central information system	4	21	25		
	In our central information system	3	16	19		
	On my personal or workstation computer/hard drive	1	8	9		
Total		9	47	56		

		Gen	der	
		Male	Female	Total
From your view, what are the challenges in sharing knowledge with people from other sections of your company. (ref: S)	Don't perceive there is an urgent need	2	21	23
	Lack of open-minded sharing environment	4	11	15
	Lack of trust of others' knowledge	0	8	8
	No proper technological platform to share	2	7	9
	Do not know there is a knowledge needs	4	28	32
Total		12	75	87

		Gen	der	
		Male	Female	Total
Knowledge Practices Section – How did you acquire the	from supervisors or co- workers in this organization	5	30	35
you use in your job? (ref:B	through self-learning	6	31	37
	through formal training	1	6	7
	at my last job	4	17	21
	elsewhere	1	4	5
Total		17	88	105

	Gender	N	Mean	Std. Deviation	Std. Error Mean
How often do you need to ask information from	Male	17	3.82	1.468	.356
your co-workers to perform your daily job	Female	87	3.45	1.469	.157
I depend on documented procedures to do my work when I have questions in fulfilling my job tasks. (ref: B)	Male	14	2.64	.929	.248
	Female	81	2.89	1.084	.120
Do you agree that staff, in your organization, do	Male	14	3.57	1.016	.272
not have social activities after work hours. (ref: S)	Female	80	3.23	1.147	.128
Do you agree that staff, in your organization, are eager to share work	Male	14	2.36	.842	.225
experience in the hallways or in casual gatherings. (ref: S)	Female	80	2.01	.684	.077
TrustAvg	Male	13	4.1282	.39764	.11029
	Female	77	4.2078	.65570	.07472
CultureAvg	Male	14	3.2468	.49464	.13220
	Female	80	3.1159	.50914	.05692

# OrgSize

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	small	55	51.9	51.9	51.9
	medium	33	31.1	31.1	83.0
	large	18	17.0	17.0	100.0
	Total	106	100.0	100.0	

# Oneway

	Descriptives											
				Std.		95% Confiden Me	ce Interval for an					
		N	N Mean Deviation		Deviation Std. Error		Upper Bound	Minimum	Maximum			
TrustAvg	small	49	4.2177	.69239	.09891	4.0188	4.4166	2.33	5.00			
	medium	25	4.2133	.54331	.10866	3.9891	4.4376	3.33	5.00			
	large	17	4.1569	.56664	.13743	3.8655	4.4482	3.33	5.00			
	Total	91	4.2051	.62612	.06563	4.0747	4.3355	2.33	5.00			
CultureAvg	small	51	2.9857	.51341	.07189	2.8413	3.1301	1.45	4.00			
	medium	27	3.2795	.45589	.08774	3.0991	3.4598	2.45	4.27			
-	large	17	3.3209	.45695	.11083	3.0859	3.5558	2.73	4.55			
	Total	95	3.1292	.50748	.05207	3.0258	3.2326	1.45	4.55			

# Test of Homogeneity of Variances

		Levene Statistic	df1	df2	Sig.
TrustAvg	Based on Mean	.599	2	88	.552
	Based on Median	.609	2	88	.546
	Based on Median and with adjusted df	.609	2	84.632	.546
	Based on trimmed mean	.537	2	88	.587
CultureAvg	Based on Mean	.135	2	92	.874
	Based on Median	.207	2	92	.813
	Based on Median and with adjusted df	.207	2	83.897	.813
	Based on trimmed mean	.168	2	92	.846

	ANOVA										
		Sum of Squares	df	Mean Square	F	Sig.					
TrustAvg	Between Groups	.049	2	.025	.061	.941					
	Within Groups	35.233	88	.400							
	Total	35.282	90								
CultureAvg	Between Groups	2.284	2	1.142	4.791	.010					
	Within Groups	21.924	92	.238							
	Total	24.208	94								

#### ANOVA

#### **Multiple Comparisons**

		Mean Difference (I-			95% Confidence Interval		
Dependent Variable	(I) OrgSize	(J) OrgSize	J)	Std. Error	Sig.	Lower Bound	Upper Bound
How often do you need	small	medium	.136	.330	1.000	67	.94
to ask information from your co-workers to		large	167	.406	1.000	-1.16	.82
perform your daily job	medium	small	136	.330	1.000	94	.67
		large	303	.437	1.000	-1.37	.76
	large	small	.167	.406	1.000	82	1.16
		medium	.303	.437	1.000	76	1.37
I depend on	small	medium	673*	.248	.024	-1.28	07
procedures to do my		large	152	.299	1.000	88	.58
work when I have questions in fulfilling my job tasks. (ref: B)	medium	small	.673*	.248	.024	.07	1.28
		large	.521	.331	.357	29	1.33
	large	small	.152	.299	1.000	58	.88
		medium	521	.331	.357	-1.33	.29
Do you agree that staff,	small	medium	.050	.269	1.000	61	.71
not have social activities		large	294	.316	1.000	-1.07	.48
after work hours. (ref: S)	medium	small	050	.269	1.000	71	.61
		large	344	.350	.983	-1.20	.51
	large	small	.294	.316	1.000	48	1.07
		medium	.344	.350	.983	51	1.20
Do you agree that staff,	small	medium	017	.171	1.000	43	.40
eager to share work		large	275	.201	.527	77	.22
experience in the	medium	small	.017	.171	1.000	40	.43
gatherings. (ref: S)		large	257	.222	.752	80	.29
	large	small	.275	.201	.527	22	.77
		medium	.257	.222	.752	29	.80

#### Descriptives

				Std		95% Confiden Me	ce Interval for an		
		Ν	Mean	Deviation	Std. Error	Lower Bound	Upper Bound	Minimum	Maximum
How often do you need	small	54	3.50	1.514	.206	3.09	3.91	1	5
to ask information from your co-workers to	medium	33	3.36	1.454	.253	2.85	3.88	2	5
perform your daily job	large	18	3.67	1.495	.352	2.92	4.41	1	5
	Total	105	3.49	1.481	.145	3.20	3.77	1	5
I depend on documented procedures to do my work when I have questions in fulfilling my job tasks. (ref: B)	small	53	2.66	1.091	.150	2.36	2.96	1	5
	medium	27	3.33	1.000	.192	2.94	3.73	1	5
	large	16	2.81	.981	.245	2.29	3.34	1	4
	Total	96	2.88	1.078	.110	2.66	3.09	1	5
Do you agree that staff,	small	51	3.24	1.176	.165	2.90	3.57	1	5
not have social activities	medium	27	3.19	1.145	.220	2.73	3.64	1	4
after work hours. (ref: S)	large	17	3.53	.943	.229	3.04	4.01	1	4
	Total	95	3.27	1.125	.115	3.04	3.50	1	5
Do you agree that staff,	small	51	2.02	.735	.103	1.81	2.23	1	4
in your organization, are eager to share work	medium	27	2.04	.649	.125	1.78	2.29	1	4
experience in the hallways or in casual	large	17	2.29	.772	.187	1.90	2.69	1	4
gatherings. (ref: S)	Total	95	2.07	.718	.074	1.93	2.22	1	4

#### Descriptives

				Std	95% Confidence Interval for Mean		ce Interval for an		
		Ν	Mean	Deviation	Std. Error	Lower Bound	Upper Bound	Minimum	Maximum
How often do you need to ask information from your co-workers to perform your daily job	Above 18 and younger than 30	15	2.80	1.265	.327	2.10	3.50	1	5
	31 - 39	28	3.25	1.481	.280	2.68	3.82	1	5
	40 - 49	20	3.95	1.356	.303	3.32	4.58	2	5
	50 and above	42	3.67	1.541	.238	3.19	4.15	1	5
	Total	105	3.49	1.481	.145	3.20	3.77	1	5
l depend on documented	Above 18 and younger than 30	14	2.93	1.385	.370	2.13	3.73	1	5
procedures to do my work when I have questions in fulfilling my	31 - 39	24	2.83	1.007	.206	2.41	3.26	1	4
	40 - 49	18	3.00	1.283	.302	2.36	3.64	1	5
Job (asks: (ref. b)	50 and above	40	2.83	.931	.147	2.53	3.12	1	5
	Total	96	2.88	1.078	.110	2.66	3.09	1	5
Do you agree that staff, in your organization, do	Above 18 and younger than 30	13	2.92	1.441	.400	2.05	3.79	1	5
after work hours. (ref: S)	31 - 39	23	3.09	1.276	.266	2.54	3.64	1	5
	40 - 49	18	3.39	1.092	.257	2.85	3.93	1	4
	50 and above	41	3.44	.923	.144	3.15	3.73	1	5
	Total	95	3.27	1.125	.115	3.04	3.50	1	5
Do you agree that staff, in your organization, are	Above 18 and younger than 30	13	2.08	.862	.239	1.56	2.60	1	4
eager to share work experience in the	31 - 39	23	2.22	.736	.153	1.90	2.54	1	4
hallways or in casual	40 - 49	18	2.17	.707	.167	1.82	2.52	1	4
gatherings: (ren. 5)	50 and above	41	1.95	.669	.104	1.74	2.16	1	4
	Total	95	2.07	.718	.074	1.93	2.22	1	4

# Job Nature

		Frequency	Percent	Valid Percent	Cumulative Percent
Valid	Non-Disclosed	2	1.9	1.9	1.9
	Administration	12	11.3	11.3	13.2
	Editorial	9	8.5	8.5	21.7
	Acquisition	28	26.4	26.4	48.1
	Design and Production	18	17.0	17.0	65.1
	Markeing and Sales	22	20.8	20.8	85.8
	Business Services	9	8.5	8.5	94.3
	Technical Services	6	5.7	5.7	100.0
	Total	106	100.0	100.0	

# Appendix G - SPSS Codebook (Quantitative Research)

Value	Label
OrgSize	1 small
	2 medium
	3 large
Consent	1 I have read the above information. I agree to participate in this study.
	2 I do not want to participate
Region	1 North America
	2 South America
	3 Africa
	4 Europe
	5 Asia
	6 Others
Gender	1 Male
	2 Female
	3a Prefer-not-to-specify
Age	1 Above 18 and younger than 30
	2 31 - 39
	3 40 - 49
	4 50 and above
YearsExp	1 less than one year
	2 one year and less than 5 years
	3 5 years and less than 10 years
	4 10 years and less than 15 years
	5 over 15 years
JobNature	0a Non-Disclosed
	1 Administration
	2 Editorial
	3 Acquisition
	4 Design and Production
	5 Publishing
	6 Markeing and Sales
	7 Distribution
	8 Business Services
	9 Technical Services
Supervise	1 less than 10
	2 10 to 50
	3 51 to 100
	4 more than 100
~	5 I do not supervise
SkillSource	1 from supervisors or co-workers in this organization
	2 through self-learning

Variable Values

# Appendix G - SPSS Codebook (Quantitative Research) (Continue\_1)

	3 through formal training
	4 at my last job
	5 elsewhere
InquiryFrequency	1 Never
	2 Occasionally
	3 Once a week
	4 Once a month
	5 Daily
RefSource	1 In paper-based documents
	2 In our team/department's members' mind
	3 In our central information system
	4 On my personal or workstation computer/hard drive
	5 Other (write-in)
KnowlegeProhibitio	
n	1 Time limitation
	2 No support to share
	3 No motivation to share
	4 No right tools available
	5 Other (write-in)
UseDocumentation	1 Strongly agree
	2 Agree
	3 Neither agree nor disagree
	4 Disagree
	5 Strongly disagree
Culture1	1 Strongly Disagree
	2 Disagree
	3 Neither Agree nor Disagree
	4 Agree
	5 Strongly Agree
Culture2	1 Strongly Disagree
	2 Disagree
	3 Neither Agree nor Disagree
	4 Agree
	5 Strongly Agree
Culture3	1 Strongly Disagree
	2 Disagree
	3 Neither Agree nor Disagree
	4 Agree
	5 Strongly Agree
Culture4	1 Strongly Disagree

# Appendix G - SPSS Codebook (Quantitative Research) (Continue\_2)

2	2 Disagree
3	8 Neither Agree nor Disagree
4	4 Agree
4	5 Strongly Agree
Culture5	Strongly Disagree
2	2 Disagree
2	Neither Agree nor Disagree
4	4 Agree
4	5 Strongly Agree
Culture6	Strongly Agree
2	2 Agree
	3 Do not know
2	1 Disagree
4	5 Strongly Disagree
Culture7	Strongly Agree
2	2 Agree
3	B Do not know
4	1 Disagree
4	5 Strongly Disagree
Culture8	Strongly Agree
2	2 Agree
3	Neither agree nor disagree
4	4 Disagree
4	5 Strongly disagree
Culture9	Strongly Agree
2	2 Agree
	Neither agree nor disagree
4	4 Disagree
4	5 Strongly disagree
Culture10	Extremely likely
2	2 Moderately likely
	3 Neither likely nor unlikely
4	Moderately unlikely
4	5 Extremely unlikely
Culture11	Extremely likely
2	2 Moderately likely
	Neither likely nor unlikely
2	Moderately unlikely
4	5 Extremely unlikely
Social1	Strongly Agree
2	2 Agree
3	B Do not know

# Appendix G - SPSS Codebook (Quantitative Research) (Continue\_3)

Social2	<ul> <li>4 Disagree</li> <li>5 Strongly Disagree</li> <li>1 Strongly Agree</li> <li>2 Agree</li> <li>3 Do not know</li> <li>4 Disagree</li> <li>5 Strongly Disagree</li> </ul>
SharingChallenge	1 Don't perceive there is an urgent need
Sharingenanenge	2 Lack of open-minded sharing environment
	3 Lack of trust of others' knowledge
	4 No proper technological platform to share
	5 Do not know there is a knowledge needs
Motivation	1 Very unmotivated because I keep my job secret to protect my job
	2 Unmotivated because I do not care what is going on in the company
	3 Motivated on average because I want to help my co-workers
	4 Motivated because I think my co-workers need me
	5 Very motivated because I think sharing mine will build a better company
Social3	1 Strongly Disagree
	2 Disagree
	3 Neither agree nor disagree
	4 Agree
	5 Strongly agree
Social4	1 Strongly Disagree
	2 Disagree
	3 Neither agree nor disagree
	4 Agree
	5 Strongly agree
Social5	1 Strongly Disagree
	2 Disagree
	3 Neither agree nor disagree
	4 Agree
	5 Strongly agree
RequiredDocument	
ation	1 Yes, it is a must
	2 No, it is not
TechToolsAvailabil	
ity	1 Yes, it is widely available
	2 No, it is not
TechTools	1 Colleague(s) (mentor/buddy/mentee)
	2 Team meetings
	3 Collaborative platform such as Google Doc, OneDrive, Drop Box

# Appendix G - SPSS Codebook (Quantitative Research) (Continue\_4)

	4	Social Media such as Facebook, Twitter
	5	Other such as (write-in)
Q24_code	1	No prohibitions
	2	Lack of Training or Knowledge
	3	Too busy
	4	Co-works are unavailable
	5	Depends on Situations
	6	Do not know
	7	Difficult to assess remotely
	8	No culture of knowledge transferred across silos
	9	All of the Above
	10	very rare occurence
	11	difficult coworkers who hoard knowledge in order to stay in power
	12	they don't really know how to do my job as I'm the only person who knows
StoringBarrier	1	Too busy, lack of time
	2	Poor technology tools
	3	Organization policy
	4	Poor information system/process
	5	I do not consider it is important
KnowProhFinalText	1	Time limitation
	2	No support to share
	3	No motivation to share
	4	No right tools availabile
	99.0	Missing or Other
KnowProhFinal	1	Time limitation
	2	No support to share
	3	No motivation to share
	4	No right tools availabile
	99	Missing or Other
RefSourceFinal	1	In paper-based documents
	2	In our team/department's members' mind
	3	In our central information system
	4	On my personal or workstation computer/hard drive

### Appendix H – QDA Miner Codebook (Qualitative Research)



Internal Meetings Retreats

Meetings in the form of Email Exchange

### Diagram 1 - Saunders et al.'s Research Onion







Diagram 3 - Ackoff's DIKW model







Diagram 5 - Nonaka and Takeuchi's SECI model



#### Appendix J – Summary of Interviews

The following provides a background summary of each interview. In general, each university press reported to their host university as a single entity, except two outliers were identified in terms of reporting structure. The university press of Interview #1 (a small organization size) maintains a consortium press status with the state-supported colleges in its local area and the university press of Interview #8 (a large organization size) is part of the publishing division of its University Library.

For Interview #1, the interview was conducted using an online meeting platform. The subject was a female director of a small size (as per the employee size category defined by the researcher for this project) university press in East South Central region of the United States. They published around 50 to 60 titles per year. This university press had 18 full-time employees and one half-time employee. The press also had an internship program that partners with the university. Besides the host university that the press is affiliated with, this press also had consortium press status with seven state-supported colleges, five private colleges in the state, and included the state's two major historical societies. At the moment, the interview was conducted, this university press had a couple of acquisition staff leaving and a new director had just come onboard.

For Interview #2, the interview was conducted using an online meeting platform. The subject was a male director of a small size (as per the employee size category defined by the researcher for this project) university press in East North Central region in the United States. The press publishes around 50 titles per year. The press expected that newly-hired staff would already have worked in the field or at least had some sort of experience in the scholarly publishing environment.

For Interview #3, the interview was conducted face to face in the subject's office. The subject was a female director of a large size (as per the employee size category defined by the researcher for this project) university press in South Atlantic North Central region in the United States. She had assumed this director position recently and so her answers were about her observations on her current job and the previous job in general. This press provides an aggregated journal database and publishes around 140 books per year in the areas of humanities, life sciences, health policy, public policy and regional titles.

For Interview #4, the interview was conducted by phone. The subject was a male director of a small size (as per the employee size category defined by the researcher for this project) university press in the Mid-Atlantic region of the United States. The press publishes between 70 and 80 books per year, mainly in the humanities and social sciences, philosophy, literary theory, history, cultural studies, critical race theory, gender studies, history, and urban studies. They have both print and digital books.

For Interview #5, the interview was conducted by phone. The subject was a male director of a large size (as per the employee size category defined by the researcher for this project) university press in the Mid-Atlantic region of the United States. The press publishes 120 books and 60 scholarly journals per year. They are one of the larger American university presses, with 125 staff members. This press' revenues come more from journals than from books.

For Interview #6, the interview was conducted by phone. The subject was a male director of a small size (as per the employee size category defined by the researcher for this project) university press in the East North Central area of the United States. They had nine fulltime staff, two part-time staff, and a number of students. The press publishes biographies for

young readers, the Cambridge Center of African Studies Series, The Civil War in the Great Interior, and a series in Appalachian Studies.

For Interview #7, the interview was conducted face to face at the interviewee's location. The subject was a male director of a small size (as per the employee size category defined by the researcher for this project) university press in the East South Central United States. The press has nine employees. In an average year, this press publishes around 30 to 35 new projects, and some re-prints and paperbacks.

For Interview #8, the interview was conducted by phone. The subject was a male director of a medium size (as per the employee size category defined by the researcher for this project) university press in the East North Central region of the United States. The press publishes 100 books a year. It has revenue of above \$3 million a year. The press' special subject areas are political science, performing arts, classical studies, the history and culture of the Midwest, English language teaching, and American studies, including subjects such as disability studies. The press is part of the publishing division of the university library. The director of this press also supervises the university's publishing division, which publishes works such as journals, white papers and technical reports, and manages the university's institutional repository.

#### VITA

Judy Li started her Ph.D. program in Information Sciences at the University of Tennessee in Fall 2014. Being the eldest child, and her father was the only breadwinner to feed eight mouths in the house, she had to work full time for the family after her high school graduation. She started her college life after her kid brother received his Bachelor's degree, and then supported the family. Therefore, she was a late bloomer in college life.

Judy supported herself through multiple college degrees throughout her adult life but had never thought of the possibility of getting a doctorate. Her previous academic work included a Bachelor's in Business Administration from Ottawa University in Kansas, USA; a Master's in Library and Information Science from the University of Western Ontario in London, Ontario, Canada, as well as a Master's in Business Administration, with a concentration in Entrepreneurship, from Nova Southeastern University in Florida, USA.

Besides having these credentials, Judy has had decades of reference and teaching field experience. She has been working with a variety of patrons (college students of various age groups both on and off campus, faculty, researchers and small business owners), as a librarian: in special libraries (with Hong Kong Telecommunication Limited in Hong Kong) in the telecommunication industry; for the Canadian Federal Government - Office of the Superintendent of Financial Institutes (Toronto Research Center) in the finance industry; for Broward County Library System (Business, Law, Government Special Section) in the USA. She also worked in academic libraries (including the Chinese University of Hong Kong, Nova Southeastern University and Mississippi State University) in the USA, Canada and Hong Kong (British Commonwealth). At the time of this writing, she works full time as a tenured faculty, Associate Professor and Business Librarian, with the University of Tennessee's Hodges Library. Judy has been active in her professional communities by serving on the Editorial Board of the *Journal of Library and Information Services in Distance Learning*, and by serving on the Advisory Board of the Business and Finance Division of the Special Library Association. She is also a longtime scholarly reviewer with *Choice* (the publishing branch of the Association of College and Research Libraries, a division of the American Library Association).

In the past, Judy obtained several solo-authorship grants such as a financial literacy program grant from the University of Tennessee Alliance of Women Philanthropists in 2014, a business research information needs research grant from University of Tennessee Library Faculty Research Incentive Program in 2012, and a financial literacy credit course teaching grant from Mississippi State University's Teaching and Learning Center in 2010.

As a scholar, she was honored in four consecutive years at the Annual Bibliography of Business and Finance Division's Authors Honoring Event at the Special Library Association, an international library professional association, annual conferences: 2010-2011, 2011-2012, 2012-2013 and 2013-2014.

In fact, among her referred publications and presentations, one of her scholarly peerreviewed articles received an international award from a publisher in Europe – the 2015 *Emerald Citation of Excellence* award for her solo-authorship article "Serving as an Educator: A Southern Case in Embedded Librarianship." Receiving this *Citation of Excellence* represents one of the highest accolades that an author can achieve, as each year Emerald recognizes only the 50 most cited articles among the approximately 15,000 articles published by the top 300 management

journals in the world.

